COLLECTIVE AGREEMENT AND BENEFIT PLANS

BETWEEN

JOHN DEERE WELLAND WORKS

OF

JOHN DEERE LIMITED

WELLAND ONTARIO

AND

THE NATIONAL AUTOMOBILE,
AEROSPACE, TRANSPORTATION AND
GENERAL WORKERS UNION
OF CANADA (CAW - CANADA)
and its local 275

EXPIRES - 30 SEPTEMBER 2010

08815(07)

1.0	NUA	DV					FE	DDI	JAR	v				B4 A	RC	ш				
JAI S 7	M 1 8	T 2 9	W 3 10	T 4 11	F 5 12	S 6 13	S 4	M 5	T 6	w 7	T 1 8	F 2 9	S 3 10	S 4	M 5	T 6	W	T 1 8	F 2 9	S 3
14 21 28	15 22 29	16 23 30	17 24 31	18 25	19 26	20 27	11 18 25	12 19 26	13 20 27	14 21 28	15 22	16 23	17 24	11 18 25	12 19 26	13 20 27	14 21 28	15 22 29	16 23 30	17 24 31
AP	RIL						MA	Υ						JU	NE					
S 1	M 2	T 3	W 4	T 5	F 6	S 7	s	M	T 1	W 2	T 3	F 4	S 5	S	M	T	W	T	F 1	S
8 15 22 29	9 16 23 30	10 17 24	11 18 25	12 19 26	13 20 27	14 21 28	6 13 20 27	7 14 21 28	8 15 22 29	9 16 23 30	10 17 24 31	11 18 25	12 19 26	3 10 17 24	4 11 18 25	5 12 19 26	6 13 20 27	7 14 21 28	8 15 22 29	9 16 23 30
JUI	LY						AU	GU	ST					SE	PTE	MB	ER			
S 1	M 2	T 3	W 4	T 5	F 6	S 7	S	M	T	W 1	T 2	F 3	S 4	S	M	T	W	T	F	S 1
8 15 22	9 16 23	10 17 24	11 18 25	12 19 26	13 20 27	14 21 28	5 12 19	6 13 20	7 14 21	8 15 22	9 16 23	10 17 24	11 18 25	2 9 16	3 10 17	4 11 18	5 12 19	6 13 20	7 14 21	8 15 22
29	30	31					26	27	28	29	30	31		23 30	24	25	26	27	28	29
00	то	BER	1				NC	VEI	ИВЕ	R				DE	CEI	ИВЕ	R			
S	M 1	T 2	W 3	T 4	F 5	S 6	s	M	7	W	T 1	F 2	S 3	s	М	Т	W	Т	F	S
7	8	9	10	11	12	13	4	5	6	7	8	9	10	2	3	4	5	6	7	8
14 21 28	15 22 29	16 23 30	17 24 31	18 25	19 26	20 27	11 18 25	12 19 26	13 20 27	14 21 28	15 22 29	16 23 30	17 24	9 16 23 30	10 17 24 31	11 18 25	12 19 26	13 20 27	14 21 28	15 22 29

JA S	NU.A	T	w	т	F	s	FE S	BRI M	JAR T	Y W	т	F	s	MA S	RC M	H T	w	т	F	 s
6 13 20 27	7 14 21 28	1 8 15 22 29	2 9 16 23 30	3 10 17 24 31	4 11 18 25	5 12 19 26	3 10 17 24	4 11 18 25	5 12 19 26	6 13 20 27	7 14 21 28	1 8 15 22 29	2 9 16 23	2 9 16 23 30	3 10 17 24 31	4 11 18 25	5 12 19 26	6 13 20 27	7 14 21 28	1 8 15 22 29
ΔΡ	RIL						MA	Y						JU	NF					
6 13 20 27	7 14 21 28	T 1 8 15 22 29	w 2 9 16 23 30	T 3 10 17 24	F 4 11 18 25	5 12 19 26	\$ 4 11 18 25	5 12 19 26	6 13 20 27	7 14 21 28	T 1 8 15 22 29	F 2 9 16 23 30	\$ 10 17 24 31	\$ 1 8 15 22 29	M 2 9 16 23 30	T 3 10 17 24	W 4 11 18 25	T 5 12 19 26	F 6 13 20 27	\$ 7 14 21 28
JU	LY						AU	IGU:	ST					SE	PTE	МВ	ER			_
JU \$	LY	Ţ	W	T 3	F 4	S 5	AU S	IGU:	ST T	w	т	F 1	S	SE S	PTE M	T	ER W	T 4	F -5	 S 6
\$ 6	M 7	18	2 9	3 10	4 11	5 12	S	M 4	T 5	6	7	1	2 9	S 7	M 1 8	T 2 9	W 3	4 11	· 5 12	6 13
S	M	1	2	3	4	5	S	M	T		-	1	2	S	M 1	T 2	W 3	4	- 5	6
6 13 20 27	7 14 21	1 8 15 22 29	2 9 16 23 30	3 10 17 24	4 11 18	5 12 19	3 10 17 24 31	M 4 11 18	5 12 19 26	6 13 20 27	7 14 21	1 8 15 22	2 9 16 23	7 14 21 28	M 1 8 15 22	7 2 9 16 23 30	W 3 10 17 24	4 11 18	·5 12 19	6 13 20
6 13 20 27	7 14 21 28	1 8 15 22 29	2 9 16 23 30	3 10 17 24	4 11 18	5 12 19	3 10 17 24 31	M 4 11 18 25	5 12 19 26	6 13 20 27	7 14 21	1 8 15 22	2 9 16 23	7 14 21 28	M 1 8 15 22 29	7 2 9 16 23 30	W 3 10 17 24	4 11 18	·5 12 19	6 13 20
\$ 6 13 20 27 OC \$ 5	7 14 21 28 TO M	1 8 15 22 29 BEF T	2 9 16 23 30 W 1 8	3 10 17 24 31 T 2 9	4 11 18 25 F 3 10	5 12 19 26 S 4 11	3 10 17 24 31 NC S	M 4 11 18 25 VEI M	5 12 19 26 WBE T	6 13 20 27 27 R W	7 14 21 28 T	1 8 15 22 29 F	2 9 16 23 30 S 1 8	7 14 21 28 DE S	M 1 8 15 22 29 CEI M 1 8	7 9 16 23 30 WBE T	W 3 10 17 24 R W 3 10	4 11 18 25 T 4 11	5 12 19 26 F 5 12	6 13 20 27 S 6 13
6 13 20 27 OC S	7 14 21 28 TO	1 8 15 22 29 BEF	2 9 16 23 30 W	3 10 17 24 31 T	4 11 18 25 F 3	5 12 19 26 S 4	3 10 17 24 31 NC S	M 4 11 18 25 VEI	5 12 19 26 MBE T	6 13 20 27 R W	7 14 21 28	1 8 15 22 29	2 9 16 23 30 S 1	7 14 21 28 DE S	M 1 8 15 22 29 CEI M 1	7 2 9 16 23 30 WBE T	W 3 10 17 24 R W 3	4 11 18 25 T 4	5 12 19 26 F 5	6 13 20 27 S 6

ARTICLE I- PURPOSE	10
1.01 ARTICLE II - RECOGNITION	
2.01ARTICLE III - MANAGEMENT RIGHTS	
3.01	
ARTICLE IV - NO DISCRIMINATION	13
4.01 No Discrimination	13
4.02 Intimidation	.13
ARTICLE V - UNION SECURITY	13
5.01 Deductions	.13
5.02 Remittance	
5.03 Notification of Change	.14
5.04 Company Indemnity	
5.05 Part-time Employees	
5.06 Method of Deduction	.15
ARTICLE VI - UNION REPRESENTATION	15
6.01 General Representation	.15
6.02 Committeepersons, Chairpersons, Time Stud	ly
6.03 Negotiating Committee	.17
6.04 Notification of Names	.17
6.05 Duties and Responsibilities	.17
6.06 Allowance for Time	.18
6.07 Committeeperson During Lay-off/Shutdown	.19
6.08 National Representative and President	
6.09 Information Lists	
6.10 Union Office	.20

ARTICLE VII – GENERAL20
7.01 Pay Day 20 7.02 Physical Examinations 20 7.03 Bargaining Unit Work 21 7.04 Non Traditional Work Assignments 21
7.05 Current Addresses, Etc22
ARTICLE VIII - DISCIPLINE22
8.01 Probationary Employee22
8.02 Seniority Employee22
8.03 Grievance on a Discharge22
8.04 Presence of a Committeeperson,23
8.05 Private Interview23
8.06 Record of Discipline23
ARTICLE IX - GRIEVANCE PROCEDURE, 24
9.01 Agreement24
9.02 Definition
9.03 Discussion with Supervisor24
9.04 Time Limit24
9.05 Steps25
9.05A Unresolved Payment Plan Grievances26
9.06 Arbitrator's Decision26
9.07 Arbitrator's Authority27
9.08 Costs of Arbitration Proceedings27
9.09 Time Limits and Procedural Requirements27
9.10 'Bound By' Requirements of Arbitrator27
9.11 Union Policy Grievance or Company Grievance27
ARTICLE X - NO STRIKES OR LOCKOUTS 28
10.01 No Strikes or Lockouts28

6

t

6

10.02 Discipline or Discharge	28
10.03 Definitions	28
ARTICLE XI - ORIENTATION	29
11.01 New Employees	29
ARTICLE XII - SENIORITY	
12.01 Acquiring Seniority	29
12.02 Reductions in the Workforce	
12.03 Temporary Lay-offs	31
12.04 Recalls and Restorations	32
12.05 Job Posting	32
12.06 Transfer from Bargaining Unit	35
12.07 Loss of Seniority and Employment	35
12.08 Notice of Lay-Off,	36
12.09 Seniority Exceptions and Transfer of Daily Tasks	′
12.10 Definition of 'Qualification' and 'Vacant Wo	
12.11 Accommodation for Incapacitated	
Employees	
ARTICLE XIII - INCOME SECURITY BENEFI	TS39
13.01 Eligibility and Duration of Income Security	
Benefits	
13.02 Limitations	
13.03 Computation	40
ARTICLE XIV - HOURS OF WORK AND OVERTIME	40
14.01 General Provisions	40
14.02 Definitions	
14.03 Paid and Unpaid Lunches and Number of	
employees on Continuous Shift	44

14.04 Shift Premiums	45
14.05 Rest Period	45
14.06 Personal Clean-Up Time	46
14.07 Allocation of Overtime	46
14.08 Overtime	46
14.09 Report-In Pay -Call-In Pay	48
ARTICLE XV – WAGES	49
15.01 Basic Wage Rates	49
15.02 Cost of Living Adjustment	
15.03 Job Classifications and New Job	
Classifications	
15.04 Establishment of Rates of Pay:	
15.05 Continuous Improvement Process/Pay	= 1
System	
15.07 Adjustment of Wage Rate – Temporary	
Assignment	
15.08 Daily Work Record	
15.09 Inventory Rate	
15.10 Computation of Average Straight-Time	
Earnings	63
15.11 Productivity Bonus	
ARTICLE XVI - RETIREMENT BONUS	65
16.01	65
ARTICLE XVII - LEAVE OF ABSENCE	66
17.01 General	66
17.02 Disability Leave	
17.03 Union Business Leave	
17.04 Public Office Leave	67

	17.05	Educational Leave	67
	17.06	Pregnancy and Parental Leave	67
		Reinstatement at the end of all Leaves	
A	RTICL	E XVIII - JURY AND WITNESS DUTY.	68
	18.01	Payment	68
	18.02	Eligibility	68
	18.03	First and Second Shift Employees	69
	18.04	Third Shift Employees	69
	18.05	Purpose of Hours Paid	69
		Jury Duty During Vacation	
		Definitions,	
Α	RTIC	LE XIX - BEREAVEMENT LEAVE	70
	19.01	Eligibility and Duration	70
	19.02	Conditions of Payment	70
	19.03	First Day of Bereavement	.70
	19.04	Delayed Burials	.70
	19.05	Memorial Services	.71
	19.06	Multiple Deaths	.71
	19.07	Cremation	.71
	19.08	Bereavement During Vacation	.71
		LE XX - EXCUSED PERSONAL	
Α	BSEN	ICE	71
	20.01	Eligibility and Payment	.71
	20.02	Computation	.72
	20.03	With Vacation and a Holiday	.72
		Personal Illness	
A	RTIC	LE XXI - HOLIDAYS	73
	21.01	Holidays Observed	.73

21.02 Computation	./3
21.03 Criteria for 'Not Scheduled'	.75
21.04 Hours Paid	.75
21.05 Exceptions	.75
21.06 Day of Observance	.76
21.07 Computation	
21.08 Weekly Indemnity or Workplace Safety and Insurance Benefits	.76
21.09 Holidays at Christmas Shutdown	.76
ARTICLE XXII - VACATION PLAN	76
22.01 Vacation Year and Weeks of Vacation	.76
22.02 Computation	
22.03 Christmas Shutdown	
22.04 Vacation Bonus	
22.05 Terminated Employees	.80
22.06 Deceased or Retired Employees	
22.07 Continuous Employment	.80
22.08 Vacation Shutdown	.81
22.09 Pay in Lieu of Vacation	.82
22.10 HolidaysWith Vacations	
22.11 Deductions	.83
22.12 Receipt of Vacation Pay	.83
ARTICLE XXIII - BULLETIN BOARDS	83
23.01	.83
ARTICLE XXIV - SAFETY AND HEALTH	
24.01 Safety Rules and Regulations	.84
24.02 Safety Committee	
24.03 Physical Agents or Toxic Materials	
24.04 Protective Equipment	

24.05 Reimbursement for Safety Shoes	86
24.06 Prescription Safety Glasses	86
ARTICLE XXV - NEW TECHNOLOGY	. 86
25.01	86
ARTICLE XXVI - APPRENTICES AND SKILLE	
TRADES	
26.01 Classifications	
26.02 Eligibility	
26.03 Seniority Employees	
26.04 Application and Standards	
26.05 Seniority	
26.06 Rates	
26.07 Credit	
26.08 Related Classroom Study	
26.09 Supervision of Apprentices	
26.10 Tools	
26.11 Information	
26.12 Term of Apprenticeship Programs	
26.13 Schedules of Work Processes/ Skill Sets	
26.14 Joint Apprenticeship Committee	
26.15 PROVISIONS FOR SKILLED TRADES	
ARTICLE XXVII - PLANT CLOSURE	
27.01	
ARTICLE XXVIII -WAIVER	
28.01	
ARTICLE XXIX - APPENDICES	101
ARTICLE XXX - DURATION	102
30.01 Effective Dates	.102

30.02 Amendments102
EXHIBIT " A RATE SCHEDULE104
EXHIBIT "C" SENIORITY CLASSIFICATIONS105
EXHIBIT "F" RED-CIRCLE RATES109
EXHIBIT "G" SPECIAL PROVISIONS FOR EMPLOYEES HIRED ON OR AFTER 16 MARCH
1999 110
Section 1. Basic Wage Rates and Total Wage Rates 110
Section 2. Productivity Bonus110
Section 3. Excused Personal Absence110
EXHIBIT "G-1" 111 BASIC WAGE AND PROGRESSION TABLE 111
LETTERS OF UNDERSTANDING113
RE: AUTOMATED PAINT TECHNICIANS AND CIPP113
RE: CONTINUOUS IMPROVEMENT AWARD - OFP PAID TIME OFF114
RE: CAW/DEERE CONTINUOUS IMPROVEMENT
PROCESS117 RE: CHAIRPERSON COMPENSATION122
RE: CIPP ORDER FULFILLMENT/DEMAND FLOW SPECIAL PAYMENT PROCEDURES124
RE: CIPP PLANNED ACTIVITIES AND NEW PRODUCT INTRODUCTION127
RE: CIPP STEERING COMMITTEE134
RE: CONTINUITY OF PAY136
RE: EMPLOYMENT INSURANCE PREMIUM REDUCTION138
RE: EMPLOYEE ASSISTANCE AND HARASSMENT139
RE: HEALTH PLAN CHANGES141

RE: HOURS OF WORK	.142
RE: LIGHT DUTY COMPENSATION	.144
RE: LUMP SUM PAYMENT	.146
RE: ORDER FULFILLMENT/DEERE PRODUCT	
SYSTEM	.148
RE: OVERTIME DISTRIBUTION	.149
RE: PAID EDUCATION LEAVE	.150
RE: PRESCRIPTION DRUG MAIL ORDER	.151
RE: RETURN FROM TOTAL DISABILITY	.152
RE: SKILLED TRADES AND CIPP	.153
RE: SOCIAL JUSTICE FUND	.154
RE: SOURCING	
RE: SPECIAL PAY PROVISIONS FOR	
PREVENTATIVE MAINTENANCE	.157
RE: SUMMER SHUTDOWN	
RE: SUMMER STUDENT PROGRAM	
RE-TRAINING	162

COLLECTIVE AGREEMENT

This agreement to be entered into as of the 1st day of October 2007 and is effective 1 October 2007.

BETWEEN:

JOHN DEERE WELLAND WORKS of JOHN DEERE LIMITED,

Welland, Ontario

(Hereinafter referred to as the "Company'?

AND -

THE NATIONAL AUTOMOBILE, AEROSPACE, TRANSPORTATION AND GENERAL WORKERS UNION OF CANADA (CAW-CANADA) and its Local 275

(Hereinafter referred to as the "Union")

ARTICLE I- PURPOSE

1.01

The general purpose of this Agreement is to establish and maintain collective bargaining relations between the Company and the Union; to continue the cooperation and spirit of goodwill between the Company and its employees; to provide machinery for the prompt disposition of grievances arising under this Agreement, and to set forth negotiated conditions of employment for all employees who are subject to the provisions of this Agreement. The Union recognizes that in order to provide a proper relationship between the parties the Company must be kept in a strong, competitive market position and be satisfactorily profitable consistent with

fair labour standards, and by achieving continuing and ongoing improvement.

ARTICLE II - RECOGNITION

2.01

The Company recognizes the Union **as** the sole collective bargaining agent with respect to wages, hours of work, benefits and working conditions for all employees of the Company in Welland, Ontario save and except supervisors, persons above the rank of supervisor, office, clerical and technical staff, sales staff, security guards, students employed during the co-operative training program with a university or community college, and students employed during the school vacation period.

For the purpose of clarity, the parties agree that the term "office, clerical and technical staff includes plant office, clerical and technical staff such as set out hereafter: Engineer, Senior Engineering Analyst, Standards Engineer, Analyst Programmer, Advanced Engineering Analyst, Field Test Coordinator, Materials Engineer, Traffic Clerk, Clerk-Typist, Record Clerk, Shipping Coordinator, Secretary, Purchasing Expediter, Maintenance Planner, Production Scheduler, Experimental Scheduler, Buyer, Coordinator Production Scheduling, Computer Programmer, Key Punch Operator, Computer Terminal Operator, Standard Data Applicator, Industrial Nurse, On-Line Computer Coordinator, Inventory Analyst, Engineering Analyst, and those "acting" in those positions.

ARTICLE III - MANAGEMENT RIGHTS

3.01

Except as, and to the extent specifically modified by this agreement, all rights and prerogatives of Management are retained by the Company and remain exclusively and without limitation within the rights of the Company and its

Management and may be exercised by Management as it, in its discretion, sees fit.

Without limiting the generality of the foregoing, the Company's rights shall include:

- (a) the right: to maintain order, discipline and efficiency; to make, alter and enforce, from time to time, reasonable rules and regulations, policies and practices, to be observed by its employees; to discipline and discharge employees in accordance with Article VIII of this Agreement.
- (b) the right: to select and hire the working force and employees; to transfer, assign, promote, demote, schedule and classify employees; to retire employees in accordance with the Company's pension plan; to plan, direct and control its operations; to select and retain employees for positions excluded from the bargaining unit; to transfer employees into or out of the bargaining unit; to operate and manage the enterprise in all respects in order to satisfy its commitments and objectives.
- (c) the right to determine: the location and extent of its operations and their commencement, expansion, curtailment or discontinuance; the direction of the working forces; the work to be done; the products to be manufactured, merchandised and sold; the standards of performance; whether to perform or contract for goods and services: the schedules of work: the methods. processes and means of performing work; job content and requirements; the qualifications of employees; the use of improved or changed methods and equipment; the number of employees needed by the Company at any time and how many shall work on any job or assignment; the number of hours to be worked; starting and guitting time; methods to be used to ensure security of the Company's property; and generally, the right to manage the enterprise and its business without interference are solely and exclusively the right of the Company.

ARTICLE IV - NO DISCRIMINATION

4.01 No Discrimination

- (a) Neither the Company nor its representatives shall discriminate, coerce or restrain any employees because of such employees' membership or lawful activity in the Union. The Company and the Union recognize and support the Company's policy against workplace harassment.
- (b) The Company, the Union and the employees agree that they will not discriminate against any employee because of his/her race, ancestry, marital status, family status, sex, creed, sexual orientation, disability, colour, place of origin, ethnic origin, citizenship, age, same-sex partnership status, or record of offences, (as those terms are defined in the Human Rights Code eg. an offence in respect of which a pardon has been granted), contrary to the provisions of the Human Rights Code.

4.02 Intimidation

The Company and the Union agree that there will be no intimidation or coercion exercised by either of them upon an employee because he/she is exercising any right conferred by the Ontario Labour Relations *Act*, 1995 and the Union agrees that there will be no Union activity other than that which is expressly permitted by this Agreement.

ARTICLE V - UNION SECURITY

5.01 Deductions

All present employees of the Company shall have deducted from their pay monthly Union dues as a condition of employment and shall have the option of joining or not joining the Union. All new employees shall, as a condition of employment, have deducted from their pay monthly Union dues and shall make application for membership in the Union. The Company also agrees to deduct from the pay of an employee who authorizes it in writing, the Union initiation

fee and to forward such initiation fee to the Union. Union Form A-230-95 shall be supplied by the Union to the Company and used for this purpose.

5.02 Remittance

All dues and initiation fees deducted must be remitted to the Local Union Financial Secretary within fifteen (15) working days of the deduction along with a list of names from whom such deductions have been made and the amount of each deduction.

5.03 Notification of Change

The Financial Secretary of the Local Union will notify the Company of any change in the amount of Union dues and/or initiation fees to be deducted in line with the constitutional requirement of the National Union.

5.04 Company Indemnity

The Union agrees to indemnify and save the Company harmless against any and all claims or other forms of liability that may arise out of or by reason of deductions made or payments made in accordance with this Article V.

5.05 Part-time Employees

All part-time employees who earn the equivalent of forty (40) hours pay or any hours which are equal to a normal full week's pay during a calendar month must have the regular Union dues of two hours', twenty minutes (2.33) pay deducted by the Company and then forwarded to the Local Union Financial Secretary. A part-time employee is defined as one who is normally employed for not more than twenty-four (24) hours per week. A part-time employee who completes the regular probationary period must pay initiation fees as established by the Local Union as a condition of further employment.

5.06 Method of Deduction

The Union shall notify the Company by letter of the method upon which the deduction of Union dues is to be calculated detailing both amounts to be included and amounts to be excluded. In addition, the Union will advise the Company by letter under what circumstances dues are payable, e.g., when the employee is receiving vacation pay, holiday pay, etc. and under what circumstances dues are not payable, e.g., when the employee is receiving Weekly Indemnity or Workplace Safety and Insurance Benefits. The foregoing shall conform to the Constitution of the National Union and the By-Laws of the Local Union.

ARTICLE VI - UNION REPRESENTATION

6.01 General Representation

The Company acknowledges the right of the Union to appoint or otherwise select Committeepersons in accordance with this Article VI, for the purpose of representing employees in the handling of grievances. In order for an employee to be eligible to be a Committeeperson the employee must have successfully completed his/her probationary period.

6.02 Committeepersons, Chairpersons, Time Study Person

The Company agrees to recognize Committeepersons and Zones based on the following table.

Number of Wage Employees	Number of Committeepersons & Number of Zones
0 to 300	4
301 to 450	5
451 to 600	6
601 to 750	7
751 to 900	8

A Committeeperson must be elected or appointed from the zone he/she represents. In addition the Company agrees to recognize a Chairperson. The Chairperson shall be elected plant-wide and will work on the first shift. The zones will be identified as zone 1 through zone 8. The departments, areas, or buildings designated in each zone will be determined by the Union for the term of the agreement unless changes are mutually agreed to by the parties. No employee shall be represented by more than one Committeeperson.

All Committeepersons will be assigned to the first shift. The Union agrees that the Committeepersons will align their shifts to provide one hour **of** overlap on the 3rd shift (6 am to 7 am) and one hour of overlap on the 2nd shift (3:30 pm to 4:30 pm). These Committeepersons will represent all employees working on the respective off shift. Any time spent by a Committeeperson on union business on second or third shift other than specified above will only be compensated if requested by the Company. Alternates will not be used on the opposite shift.

The Chairperson will provide a schedule **of** hours to the Manager, Human Resources or his/her designate by Wednesday of the preceding week.

If a Committeeperson is going to be out of the plant for one (1) day or more, an alternate Committeeperson from his/her zone may act in his/her absence, provided the Union has given the Company advanced written notice of the name of such alternate Committeeperson.

Time used by an alternate Committeeperson is deemed to be time used by the Committeepersonfor whom he/she is the alternate.

Union Time Study Representative:

To select a Union Time Study Representative or Alternate, who shall not be a Committeeperson, the Union shall submit the names of five (5) employees who shall meet the qualifications established by the Company. From this group the Company shall select three (3) whom it deems to be suitable and qualified. The Union shall then select one (1) of these three (3) to receive training at Company expense.

Should the Company promote the Union Time Study Representative or Alternate to a position outside the bargaining unit a replacement will be trained by the Company. If the Union time study person needs to be replaced for a reason other than being promoted to **a** position outside the bargaining unit, the training **costs** shall be borne by the Union.

The Alternate Union Time Study Representative will be used only during periods when the regular union Time Study Representative is out of the factory for durations longer than one day (i.e. vacation). The Chairperson will notify the Company when the Alternate will be acting as the Time Study Representative.

6.03 Negotiating Committee

The Company agrees to recognize the National Representative, President of the Local Union if not a Committeeperson, Chairperson and the number of committeepersons in effect on the first of the month prior to the contract expiration date as the Negotiating Committee for the purpose of negotiating the renewal of this Agreement.

6.04 Notification of Names

The Union shall notify the Company in writing of the names of the Committeepersons and Chairperson, the effective dates of their appointments and the names of any Committeepersons they are replacing. The Company shall not be required to recognize any Committeeperson or Chairperson not **so** named.

6.05 Duties and Responsibilities

Chairperson:

The Chairperson will have the right to devote full time to the discharge of his/her duties as Union Chairperson as specifically enumerated in this agreement. The Chairperson will report to the Manager of Human Resources. When it becomes necessary to visit a department in the factory, he/she will obtain permission from the Manager of Human Resources or his/her designated representative. He/she will

advise the Manager of Human Resources of the general nature of the matter he/she is dealing with, if requested. The Company agrees to not unreasonably withhold such permission and the Union agrees to keep the time spent to the minimum necessary.

Committeeperson:

The primary obligation of a Committeeperson is to perform his/her regular work for the Company; accordingly, no Committeeperson shall leave his/her work to administer this Agreement without requesting and obtaining the permission of his/her supervisor. When returning to his/her regular duties he/she will report to his/her supervisor and advise the supervisor of the general nature of the matter he/she was dealing with during his/her absence, if requested. The Company agrees to not unreasonably withhold such permission and the Union agrees to keep the time spent to the minimum necessary. The jurisdiction of a Committeeperson is limited to the department or group of departments he/she represents.

Union Time Study Representative:

The duties of the Union Time Study Representative shall consist of supporting CIPP plan development and maintenance, and in the investigation of disputes involving Article XV Wages, Section 15.05 Continuous Improvement Pay Plan.

6.06 Allowance for Time

The Company agrees that Committeepersons shall not suffer loss of pay up to a maximum of one (1) hour each per day, five days per week, for time spent in the processing of grievances during the Committeeperson's regular straight time working hours.

In addition, the Chairperson shall not suffer loss of pay up to a maximum of eight(8) hours per day, five days per week, for time spent in carrying out his/her duties herein.

The Union time study representative shall not suffer loss of pay up to a maximum of two hundred and fifty (250) hours per calendar year for time spent in carrying out his/her duties

herein. The Manager **of** Human Resources and the Chairperson may by mutual agreement provide pay in excess of the hours provided in this paragraph.

Payment provided for herein shall not apply to arbitration or time spent meeting with the Company for the purpose of negotiations.

6.07 Committeeperson During Lay-off/Shutdown

The Company agrees to retain the Plant Chairperson, Committeepersons and the Time Study Representative at work in their zone in the event of a lay-off or reduction provided they have the skill and ability to do the work required.

In the event of a shutdown, the Company agrees, upon Union request, to retain the Plant Chairperson and appropriate Committeepersons for the zones where a significant number of employees are scheduled to work. The Committeeperson will work in his/her zone only when the Committeeperson is qualified to perform the available work.

6.08 National Representative and President

A National Representative and the President of the Local Union may be present and participate in any meeting of the Plant Committee and the Company.

6.09 Information Lists

The Company will supply to the Chairperson the following information:

- (a) a seniority list, monthly, showing employees' classifications:
- (b) lay-offs and recalls subject to Article 12.08;
- annually, names and addresses of bargaining unit employees and of bargaining unit employees who retire after 21 July 1989;
- (d) on a monthly basis, provided there is a change:
 - employees on leave of absence; and transfers into and out of the bargaining unit,

- (ii) employees on Workplace Safety and Insurance Benefits, Weekly Indemnity, Long-Term Disability, and last day worked,
- (iii) notification of an employee who has lost seniority under Article 12.07 or is sent a Registered letter pursuant to Article 12.07;
- (e) managerial personnel who have jurisdiction over bargaining unit employees.

6.10 Union Office

The Company agrees to provide the Union with an office, basic desk and chairs, filing cabinet and telephone. The Union shall pay all telephone charges other than the basic monthly charge.

ARTICLE VII - GENERAL

7.01 Pay Day

- (a) Pay day will be Thursday of each week for first and second shift employees and Friday for third shift employees unless a holiday falls on a pay day, then pay day will be on the day before the holiday. If a temporary change is necessary, employees will be notified.
- (b) Employees will be paid for services rendered during the previous workweek. All pay statements will be distributed during regular work hours no later than the regular pay day.
- (c) Employees will have their payroll cheques deposited directly to the employee's designated bank or credit union account(s) on Thursday of each week. Each employee will provide the Human Resource Department with the required banking information.

7.02 Physical Examinations

(a) All applicants must take a physical examination at Company expense after being accepted for employment.

(b) The Company may upon reasonable grounds require a medical examination of any employee which will be made at the Company's expense. A report thereof will be furnished to the personal physician of the employee, upon written request of the employee and his/her physician. The Company will pay mileage for an employee to drive their personal vehicle to attend a Company required medical appointment. Mileage will be based on the predetermined distances and the mileage expense rate established by the Company.

7.03 Bargaining Unit Work

It is the intent of the parties that members of Management will not perform the functions of bargaining unit work to such a degree as causes the lay-off of bargaining unit members or an erosion of bargaining unit work or the loss of an overtime assignment.

This clause is being written in a broad, general manner to avoid nit-picking and conflict when members of Management perform such functions on the one hand, while preserving the integrity of the bargaining unit on the other hand.

7.04 Non Traditional Work Assignments

Employees covered by this Agreement may be assigned to certain functions which they traditionally have not performed or which have not exclusively been performed by members of the bargaining unit for not more than one (1) year unless by mutual agreement of the parties. Examples of such functions are:

- (a) Programming and maintenance of computers and robots:
- (b) Identifying training needs, developing training materials and conducting training;
- (c) Leading or participating in project teams or working individually on subjects covering a wide range of functional areas;

- (d) Special assignments given to employees to accommodate temporary medical restrictions; and
- (e) Evaluating, testing and/or analyzing experimental, proved design and/or production parts, assemblies etc. in order to develop or prove engineering theory and design.

The performance of such functions by bargaining unit members shall not cause such functions to be bargaining unit work.

Upon completion of a Non Traditional Work Assignment with a duration greater than one year, the employee will be reassigned to an assignment within his/her home department and classification as seniority permits.

7.05 Current Addresses, Etc.

It is the obligation of every employee to keep the Human Resources Department advised of his/her current address, telephone number or change in marital or dependent status.

ARTICLE VIII - DISCIPLINE

8.01 Probationary Employee

The discipline or discharge of an employee who has not completed his/her probationary period, hereinafter referred to as a "probationary employee", shall be at the sole discretion of the Company and shall not be grievable nor arbitrable.

8.02 Seniority Employee

The discipline of an employee who has completed his/her probationary period, hereinafter referred to as a "seniority employee", shall be for just cause and shall be subject *to* the grievance procedure.

8.03 Grievance on a Discharge

A claim by a seniority employee that he/she has been discharged without just cause, shall be treated as a grievance and shall commence at Step No. 2 of Article 9.05

provided a written grievance signed by the employee and his/her Committeeperson is presented to the Manager of Human Resources or his/her designate within three (3) working days after the discharge. The National Representative of the Union will be permitted to attend the meeting held pursuant thereto, with Management.

8.04 Presence of a Committeeperson

A Committeeperson will attend a meeting at which the Company is disciplining or discharging a seniority employee unless the employee states, in the presence of a Committeeperson, that he/she does not wish to have a Committeeperson present.

8.05 Private Interview

- (a) When a seniority employee is required to attend a disciplinary hearing, he/she will be given the opportunity, if he/she so desires, to have a private interview prior to the hearing of not more than fifteen (15) minutes with his/her committeeperson.
- (b) When a seniority employee is discharged from employment while at work he/she will be given the opportunity, if he/she so desires, to have a private interview of not more than thirty (30) minutes with the Plant Chairperson or his/her Committeeperson.

8.06 Record of Discipline

In imposing discipline for causes other than absenteeism, the Company will not take into account any infraction of Company rules of conduct which occurred more than three (3) years prior to the date upon which the offense for which the employee is being disciplined occurred. Notwithstanding the foregoing, if an employee has a clean slate, i.e. no discipline for a continuous period of eighteen (18) months, discipline prior to such eighteen (18) month period will not be taken into account in imposing discipline. In addition, if a verbal or written reprimand has been on file for one (1) year without any intervening disciplinary action, this

discipline will not be taken into account in imposing discipline.

ARTICLE IX - GRIEVANCE PROCEDURE

9.01 Agreement

The Company and the Union agree that the specifically designated grievance procedure shall be properly followed within the time limits provided.

Whenever the term "grievance procedure" is used in this Agreement, it shall be considered as including the arbitration procedure.

9.02 Definition

"Grievance" shall mean a complaint or claim concerning the discipline or discharge **of** a seniority employee, or a dispute with reference to the interpretation, application, administration or alleged violation **of** this Agreement.

9.03 Discussion with Supervisor

No employee shall have a grievance until he/she has discussed his/her complaint with his/her immediate supervisor. The immediate supervisor will respond to the employee within one (1) working day from such discussion. If the immediate supervisor does not settle the matter to the employee's satisfaction, the employee's grievance may be submitted to Step No. 1 of the grievance procedure.

9.04 Time Limit

The Company shall be under no obligation to consider or process any grievance unless such grievance has been presented to the Company in writing at Step 2 of the grievance procedure within ten (10) working days from the time the circumstances upon which the grievance is based occurred.

9.05 Steps

Step No. 1:

If the employee's complaint is not settled under Article 9.03, a meeting will be held between the griever who will be accompanied by his/her Committeeperson, and the griever's supervisor and a representative of the Human Resources Department within three (3) days of the response of the supervisor under Article 9.03. The supervisor will give his/her reply to the Committeeperson by the end of the second working day following such meeting and the giving of such reply will terminate Step No. 1.

Step No. 2:

If the grievance is not settled at Step No. 1, the grievance shall be reduced to writing, signed by the griever and the Committeeperson and submitted to the Manager of Human Resources within the ten (10) working days specified in Article 9.04. A meeting of three (3) members of the Committee and the Manager of Human Resources and other Management representatives shall take place on the fourth (4th) Thursday of the month unless mutually agreed otherwise. A National Representative of the Union may be present at the Step No. 2 meeting if requested by either party. A decision shall be rendered by the Manager of Human Resources, in writing, within five (5) working days of such meeting.

Step No. 3:

In the event the grievance is not settled at Step No. 2, the party having carriage of the grievance may request arbitration of the grievance by giving notice in writing to the other party within ten (10) days from the delivery of the decision at Step No. 2, but not thereafter. The notice to arbitrate shall set out the nature of the grievance, the section or sections of the Collective Agreement claimed violated and the remedy sought. If a request for arbitration is not so given within such ten (10) day period, the decision at Step No. 2 shall be final and binding upon both parties to this Agreement, and upon any employee involved. A sole arbitrator shall be selected from among the following panel

(except in the case of arbitration of a payment plan grievance):

Maureen Saltman Ross L. Kennedy Professor Gail Brent Randy Levinson William Kaplan

The party giving notice to arbitrate shall indicate in writing two (2) of the arbitrators who are <u>not</u> acceptable as arbitrator in the particular matter. The party receiving the notice to arbitrate shall indicate in writing two (2) of the remaining three (3) arbitrators who are <u>not</u> acceptable for the particular matter. The remaining arbitrator shall be the arbitrator in the current case. In the event of death or retirement from practice of one of the foregoing arbitrators, he/she will be replaced by Professor Wesley B. Rayner (except in the case of arbitration of an incentive standard).

9.05A Unresolved Payment Plan Grievances

A special arbitrator who is an industrial engineer shall be selected by mutual agreement between the parties to deal with an unresolved payment plan grievance. In addition to hearing evidence by the Union and the Company the industrial engineer arbitrator shall make such observation and study of the operation in question as is necessary to enable him/her to develop the true facts in the case and shall have access to any information pertaining to the dispute including standard data or plant data or study(ies) of the operation. Upon completion of his/her study the industrial engineer arbitrator shall render his/her decision in the matter.

9.06 Arbitrator's Decision

The arbitrator shall hear and determine the matter and shall issue a decision which shall be final and binding upon the parties and upon any employee or employees affected by it

9.07 Arbitrator's Authority

The arbitrator shall not be authorized to make any decision inconsistent with the provisions of this Agreement or to add to, alter, modify or amend any part of this Agreement, nor to adjudicate any matter not specifically assigned to him/her by the notice to arbitrate.

9.08 Costs of Arbitration Proceedings

Each party hereto shall bear its own costs of and incidental to any such arbitration proceeding. The fees and charges of the arbitrator shall be borne equally by the two parties hereto. Each party has the right to require the attendance of the griever at the arbitration hearing.

9.09 Time Limits and Procedural Requirements

The time limits and other procedural requirements set out in this Article IX are mandatory and not merely directory, therefore Section 48(16) of the *Ontario Labour Relations Act,1995* or as it may be amended from time to time shall not apply. The parties may agree to extend time limits provided they do so in writing.

9.10 'Bound By' Requirements of Arbitrator

It is agreed that any arbitrator appointed pursuant to Section 48 of the *Ontario Labour Relations Act,1995* or as it may be amended from time to time will be bound by the requirements of the grievance and arbitration procedure set out herein

9.11 Union Policy Grievance or Company Grievance

A Union policy grievance or a Company grievance may be submitted to the Company or the Union, as the case may be, in writing within five (5) days from the time the circumstances upon which the grievance is based occurred. A meeting between the Company and the Union shall be held within five (5) days of the presentation of the written grievance or at such other time as is mutually agreed

between the Union Representative and the Manager of Human Resources and shall take place within the framework of Step No. 2 of Article 9.05 hereof. The Company or the Union, as the case may be, shall give its written decision within five (5) days after such meeting has been held.

If the decision is unsatisfactory to the grieving party, the grievance may be submitted to arbitration within ten (10) days of the delivery of such written decision and the arbitration sections of this Agreement shall be followed.

The provisions of this Article 9.11 shall not be used by the Union to institute a grievance directly affecting an employee or employees which such employee or employees could themselves institute and the provisions of Article 9.05 hereof shall not be bypassed.

ARTICLE X - NO STRIKES OR LOCKOUTS

10.01 No Strikes or Lockouts

The Union undertakes and agrees that while this Agreement is in operation neither the Union nor any employee shall take part in or call or encourage any strike, picketing, sitdown, slowdown or any suspension of or stoppage of or interference with work or production which shall in any way affect the operations of the Company, and furthermore that there shall be no sympathy strikes nor secondary boycotts. The Company agrees that it will not engage in any lockout during the term of operation of this Agreement.

10.02 Disciplineor Discharge

Any employee who participates in any of the foregoing conduct shall be subject to discipline which may include discharge.

10.03 Definitions

The word "strike" and the word "lockout" shall be deemed to have the meaning given to these words under the Ontario *Labour Relations Act, 1995*.

ARTICLE XI - ORIENTATION

11.01 New Employees

The Union and the Company will cooperate and participate jointly in a new employee orientation program including a discussion of the importance of quality work, safety, general factory operations, and the employee's responsibilities and benefits under the provisions of this Agreement.

ARTICLE XII - SENIORITY

12.01 Acquiring Seniority

An employee shall not have any seniority and shall be considered as a probationary employee until he/she has completed three (3) continuous months of employment with the Company. During such probationary period he/she will have no seniority rights. The discharge of a probationary employee shall be in the sole discretion of the Company. Upon completion of the probationary period an employee's seniority will be backdated to the date of hire.

- A. The phrase "seniority classification", as used herein, refers to the seniority classification in which the employee has seniority rights, and it means the seniority classification to which he/she has restoration rights.
- B. An employee who, as a result of reduction is transferred to another department or classification shall have rights in this "secondary classification or department". Such employee will have seniority in two seniority classifications at the same time. Upon restoration to his "seniority classification" the employee will forfeit his/her rights to this "secondary seniority classification".
- C. An employee who, as a result of reduction is transferred to a secondary classification or department may elect to remain in that secondary classification or department. If he/she so elects his seniority will be permanently established in that seniority classification and

department as of that date. An employee making such an election will no longer be eligible for income security benefits, if eligible.

An employee can exercise this option by submitting a completed "Application for Permanent Transfer" to the department of Human Resources. Such application will only be accepted after the employee has been in the secondary classification for more than 60 days.

- **D.** In addition to classification rights, employees may hold job assignment rights.
 - An employee in their seniority classification will have the right to a job assignment. If as a result of a reduction, he/she is transferred to another job assignment within his/her department, he/she will have secondary job assignment rights.
 - While in a secondary classification, an employee will have the right to a job assignment. If as a result of a reduction, he/she is transferred to another job assignment within this department, he/she will have secondary job assignment rights.
 - Upon restoration to their seniority classification employees forfeit their rights to the secondary job assignment.

12.02 Reductions in the Workforce

Recognizing that seniority is inherent to any employment security program a workable system must include minimal disruption and loss of productivity.

To this end, when a reduction in the workforce is required, other than lay-offs provided for in Articles 12.03 and/or 12.09(a)(iii), the following procedure will take place:

Step 1:

The junior employee in the classification affected in the department will be reduced from his/her department and will exercise seniority to displace the most junior employee in his/her classification in the bargaining unit provided he/she is qualified to perform such junior employee's job.

Step 2:

Such junior employee displaced from his/her classification will exercise seniority to displace the most junior employee in his/her seniority unit provided he/she is qualified to perform such junior employee's job.

Step 3:

Such junior employee displaced from his/her seniority unit will exercise seniority to displace the most junior employee in the group, as indicated in Exhibit "C", provided he/she is qualified to perform such junior employee's job.

Step 4:

Such junior displaced employee will exercise seniority to displace the most junior employee in the bargaining unit, if there is one, provided he/she is qualified to perform such junior employee's job.

The junior displaced employee will be laid off.

In each of the foregoing steps, if a vacant work assignment exists in the employee's classification, seniority unit or bargaining unit, as the case may be, which the employee is qualified to perform, then such employee will be assigned directly to such vacant work assignment rather than exercise seniority to displace the junior employee.

In each of the foregoing steps, when there are multiple reductions, the company will endeavour to place employees in assignments and/or departments in which they have prior experience.

12.03 Temporary Lay-offs

Notwithstanding anything contained in Articles 12.02 and 12.04 hereof, in the event of a temporary lay-off of a known duration, of more than five (5) working days but not exceeding six (6) continuous weeks at any time, the following will apply to those employees who are affected within a department, area or focused factory, as the case may be.

(i) Employees with less than one (1) year's seniority in each affected group will be laid off first,

- (ii) Thereafter, if there are still more employees remaining in the affected group than are required, employees in such affected group shall be laid off in order of seniority with the most senior employees being laid off first, (upon being notified such senior employees may displace the most junior employees in the affected area, provided such senior employees are qualified to perform the work required)
- (iii) If, for some reason, the lay-off of an affected group exceeds six (6) weeks, the senior employees in such affected group will return in order of seniority with the most senior employees being returned first, and junior employees will be laid off in order with the most junior employees being laid off first, provided such senior employees are qualified to perform the work required.

12.04 Recalls and Restorations

When an increase in the workforce is required and employees are on lay-off, employees will be recalled to vacant work assignments in order of seniority subject to the following.

An employee will be recalled to work in his/her classification, if it is available, and if not then to other available work, provided that in either case he/she is qualified to perform the work required. The goal herein is to restore the employee to his/her classification and unit in line with his/her seniority as work becomes available which he/she is qualified to perform.

When an employee is recalled, restored or assigned to a job assignment that is caused by an employee's absence due to an accident or illness, upon the return of the absent employee, the most junior employee who filled the vacancy will be displaced.

12.05 Job Posting

(A) When it becomes necessary to add employees to any seniority classification within an area or focused factory, such vacancies, except those created by job bidding, will be subject to job bidding after the recall or restoration of all employees with recall or restoration rights to the seniority classification within the area or focused factory.

Any vacancy which in the best judgment of the Company is of a temporary nature, that is, of less than sixty (60) days' duration, **shall** not be listed as a vacancy and it shall not be posted for bidding.

- (B) A vacancy created through an employee accident or illness and based on medical opinion that it will likely exist for sixty (60)calendar days or longer will be posted as a "SpecialDuration Job Bid'.
 - (i) The successful bidder will retain recall rights to his/her previous seniority classification.
 - (ii) The successful bidder will return to his/her previous seniority classification when this assignment is completed providing his/her seniority will permit such return.
 - (iii) If the "Special Duration" status is removed from this job bid, the employee will lose recall rights to his/her previous seniority classification and retain recall rights to this classification.
 - (iv) Any vacancies created by a "Special Duration Job Bid will in turn be filled under the provisions of paragraphs(i), (ii) and (iii) of this clause (B).
- (C) The Company shall post the vacancy including the Pay level to be filled on bulletin boards in the plant. No vacancy shall be posted during the annual inventory and vacation shutdown period. The posting shall list the seniority classification and department or area in which the vacancy exists and the qualifications for the work assignment. The Company has the right to prescribe the qualifications necessary for each posted vacancy.
- (D) Any employee with seniority, excluding those employees in the seniority classification within the department in which the vacancy exists, who is qualified or believes himself to be qualified, may apply for the vacancy by submitting a completed "Application For Vacancy" to the

Department of Human Resources within the period specified. Any employee who is working in the department and within the same seniority classification in which the vacancy exists may request from the supervisor of the department, assignment to the vacancy. If more than one (1) employee requests assignment to the vacancy, the supervisor will assign the employee with the greatest seniority, provided he/she is qualified to perform the work, and provided the employee's request has been made before the expiration of the posting date referred to in this Article 12.05.

- (E) The senior job bid applicant will be selected on vacancies in Pay level 8 and 9 except where physical qualification and/or specific skills are necessary to perform the required work.
- (F) When the vacancy has been posted for three (3) working days (seventy-two (72) hours), the Company shall remove the posting from the bulletin board and accept no more applications for the vacancy. From the applications filed the Company shall determine those who are qualified, and from this group, if there is more than one (1), shall fill the vacancy under the following procedure:
 - (i) The senior qualified employee from any department in the plant on the basis of the applicant's plantwide seniority.
 - (ii) If it is determined by the Company that there are no qualified applicants, then the Company may fill the vacancy by transfer or hire.

The senior employee who submits an application form, if qualified, will be deemed to have accepted the position.

(G) No employee selected to fill a vacancy by this procedure shall be required to be placed in the vacancy until the Company has secured a replacement for the vacancy caused by the applicant's transfer and the Company may fill the listed vacancy temporarily by any method until the replacement has been secured.

- (H) Any employee who fills a vacancy by this bidding procedure and who fails to perform the job satisfactorily shall be returned to the seniority classification he/she was in at the time he/she was transferred as a result of the operation of this bidding procedure, provided his/her seniority will permit such return.
- (I) Any vacancies created or caused by this procedure may be filled by the Company by recall or restoration before transfer or hire. In filling such vacancy the senior employee who placed his/her name on the vacant work assignment list, if qualified, will be deemed to have accepted the position.
- (J) The Company will endeavour to transfer the successful job bidder to the vacancy within fifteen (15) working days of the Human Resources Department notifying the successful bidder of the assignment. In special circumstances where it is not possible to transfer the successful bidder in accordance with this paragraph, the Chairperson and the Manager of Human Resources (or their designated representative) will jointly review the situation

12.06 Transfer from Bargaining Unit

Any employee who has heretofore been, or may be promoted or transferred any time in the future to any salary position and/or to any job not included and/or covered by this Agreement, and who later returns to the bargaining unit of employees covered by this Agreement shall have his/her seniority classification and plant-wide seniority accumulated and re-established.

12.07 Loss of Seniority and Employment

A seniority employee shall lose his/her seniority standing and employment and his/her name shall be removed from the seniority list for any of the following reasons:

(a) if the employee quits his/her employment with the Company;

- (b) if the employee is discharged for just cause and such employee is not reinstated;
- (c) if an employee who is recalled to work while on lay-off fails to notify the Company within three (3)working days that he/she accepts the recall, or fails to return to work or furnish a satisfactory reason for not doing so within five (5)working days after telephoned or notice of recall has been sent by registered mail, delivery or courier, to the last address that the Human Resources Department has in its files for the employee. It is the responsibility of an employee to ensure that the Company's employment files contain his/her correct name, address, telephone number and social insurance number. If the employee fails to do so the Company will be relieved of any obligation to the employee not fulfilled because of incorrect address or telephone number:
- (d) if the employee overstays a permitted leave of absence or vacation without securing extension of such leave of absence or vacation from the Manager of Human Resources, unless the employee can establish that he/she has a verifiable bona fide reason and that he/she used his/her best efforts to return to work on time;
- (e) if the employee is absent from his/her duties for three (3) consecutive days, unless the employee can establish that he/she has a verifiable bona fide reason and that he/she used his/her best efforts to be at work;
- (9 any absence from active employment except for a formal leave of absence, a plant incurred injury, Weekly Indemnity, Long-Term Disability Benefits, Total and Permanent Disability Retirement or military service for a period of time equal to the employee's seniority prior to such absence or for a period of five (5) consecutive years, whichever is the lesser. However, in no instance will an employee's seniority and employment be broken by an absence of less than two (2) years.

12.08 Notice of Lay-Off

In cases of an anticipated and planned lay-off which is expected to last more than five (5)working days and when

notice is not required by the Regulations of the Employment Standards Act of Ontario the Company will endeavour to give employees affected by such lay-off five (5) days' notice thereof. The Committee Chairperson shall be provided with a list of employees to be so laid off or to be recalled from layoff. Within five (5) days of being provided with such list the Chairperson shall inform the Manager of Human Resources of any seniority problems in such list that he/she is aware of. In addition, the Company will post up a list of the employees to be laid off. The Company shall not be responsible for any improper lay-off or recall from lay-off if the person affected does not notify the Manager & Human Resources within ten (10) working days of such list being posted. The notice of layoff provided for herein need not be given to an employee who is ready to return to work after being on Workplace Safety and Insurance Benefits, Sickness or Accident leave, or any other leave of absence.

12.09 Seniority Exceptions and Transfer of Daily Tasks

The seniority provisions provided for herein shall not apply in the following circumstances:

- (a) (i) against apprentices or trainees;
 - (ii) during the annual inventory and/or vacation shutdown period;
 - (iii) in the case of any reductions or lay-off of five (5) working days or less;
 - (iv) in the case of restorations or recalls for a period of ten (10) working days or less.
- (b) When a major portion of a daily task is permanently transferred to another department within the factory, the senior employee(s) who normally performs such task(s) shall be given first opportunity to transfer with the work to the new department providing the work remains within the same classification. When a specific machine is permanently transferred to another department then the senior employee on that machine shall be given the first opportunity to transfer with the machine providing the work remains within the same classification.

12.10 Definition of 'Qualification' and 'Vacant Work Assignment'

- (a) The terms "qualified" and "qualifications" as used in this Agreement shall mean that the employee possesses the necessary experience, skill and ability to satisfactorily perform with normal supervision the work for which he/she is being considered.
- (b) The term "vacant work assignment" as used in this Agreement means a work assignment which is not required to be filled by job bidding, restoration, recall, or a work assignment which is created by the lay-off or reduction of an employee with less than one (1) year of seniority at the time of his/her lay-off or reduction.

12.11 Accommodation for Incapacitated Employees

An employee who has become permanently incapacitated for his/her regular work will be transferred by the Company to any other work he/she can perform in accordance with the following procedure:

- (a) The incapacitated employee will be placed in available work if such work exists, provided he/she is qualified to perform such work,
- (b) If there is no available work, and if there are no jobs held by probationary employees which the incapacitated employee is qualified to perform, then the plantwide seniority list will be checked with the employee with the least seniority upward until a job is found which the incapacitated employee is qualified to perform. It is understood that no employee will be displaced who has greater seniority than that of the incapacitated employee. The employee displaced will be assigned in accordance with the provisions of Article 12.02 of this Article XII. The Company will review the circumstances with the Union Chairperson before exercising this provision.
- (c) The rate of pay for any incapacitated employee transferred under this provision will be the rate of the job to which he/she is transferred as set out in Exhibit A or Exhibit G-1.

(d) If and when the incapacitated employee is, once again, on the basis of medical evidence, qualified to perform his/her former job, he/she shall be returned to his/her former classification provided his/her seniority will so permit.

ARTICLE XIII - INCOME SECURITY BENEFITS

13.01 Eligibility and Duration of Income Security Benefits

- (a) Eligibility for Income Security Benefits will arise:
 - (i) when an employee with one (1) year or more of seniority is subject to reduction from his/her seniority classification but is reassigned by the Company to a vacant lower rated work assignment.
 - (ii) when a laid-off employee with one (1) year or more of seniority at the time of his/her lay-off is recalled by the Company to a lower rated work assignment in another seniority classification before the exhaustion of his/her period of eligibility for Income Security Benefits.
 - (iii) when an employee with one (1) year or more of seniority is restored to his/her home classification to replace an absent employee, the period of eligibility will be restarted after the employee has been restored for more than 10 working days.

(b) Duration of Benefits

The period of eligibility for Income Security Benefits for an employee with one (1) or more years of seniority will be the one hundred and four (104) weeks immediately following his/her lay-off or reduction from his/her seniority classification **as** provided in the Seniority Article.

13.02 Limitations

An employee will not become eligible for Income Security Benefits until he/she has acquired one (1) year of seniority.

13.03 Computation

- (A) The computation rate for an employee shall be established as the employee's average wage rate (excluding any premium or bonus of any kind) for all hours paid for as hours worked on assignments during the last four (4) pay periods he/she worked prior to his/her lay-off or reduction from his/her seniority classification. The Income Security Benefit shall be established as the difference, if any, between the use of his/her computation rate and the use of the wage rate(s) for the work performed on the lower rated work assignment(s)) to which he/she is assigned or to which he/she is recalled during his/her eligibility period.
- (B) When a CIPP employee who otherwise qualifies, who is recalled or assigned to a NON-CIPP work assignment, the computation rate as determined in Article 13.03(A) will be converted to a NON-CIPP computation rate.
- (C) The Income Security Benefit shall be paid on a pay period basis.
- (D) Income Security Benefit payments will be considered as a part of wages for the purpose of determining wage related benefits.

ARTICLE XIV - HOURS OF WORK AND OVERTIME

14.01 General Provisions

The provisions of this Article XIV are not to be interpreted as a guarantee of, or limitation upon, the hours of work to be done per day or per week or otherwise, nor as a guarantee of working schedules but shall serve to assist the parties in the computation of regular pay and overtime pay.

Any major change in established shift hours shall be discussed with the Union Committee as far in advance of any such change as reasonably possible.

14.02 Definitions

- (A) Definitions
 - (1) "Workweek": For the purposes of straight-time and overtime pay for hours over forty (40) in a week, the workweek shall be a seven (7) day period Sunday midnight to Sunday midnight, with two (2) consecutive off-duty days. Except for employees on seven (7) day continuous operations the off-duty days will be Saturday and Sunday. For employees on seven (7) day continuous operation the employee's work schedule will be such that he/she will have two (2) consecutive off-duty days though the two (2) off-duty days might fall in two (2) different workweeks.
 - (2) "Workday", "holiday" and "off-duty day": The period of twenty-four (24) consecutive hours from the time the employee normally or would normally begin his/her shift or the time the employee is required to report for work, whichever is earlier. In no case will specific hours worked be considered in more than one (1) day for purposes of determining overtime pay. The holidays celebrated are defined in Article XXI.
 - (3) Saturday: Saturday shall be interpreted as the hours falling between midnight Friday and midnight Saturday except that third shift employees whose fifth day starts on Friday night shall be considered as working on Friday for the entire shift.
 - (4) Sunday: Sunday shall be interpreted as the hours falling between midnight Saturday and midnight Sunday except that third shift employees whose sixth day starts on Saturday shall be considered as working on Saturday for the entire shift.

- (5) Overtime Pay: The payment required of the Company when an employee is worked under Article XIV.
- (6) Premium Pay: The Payment required of the Company when an employee is worked under Article XIV.

(B) Sunday Work Premium Pay

An employee shall be paid time and one-quarter (1-1/4) for all hours worked on Sunday when Sunday is one of the employee's workdays for which straight-time would be paid were it not for this clause, such as Powerhouse employees.

(C) Holiday Work Premium Pay

An employee shall be paid double (2) time for all hours worked on a holiday(s) as defined. This payment shall be in addition to the holiday pay provided for in Article XXI.

- (D) Off-Duty, Saturday and Sunday Overtime Pay
 - (1) An employee shall be paid time and one-half (1-112) for all hours worked during the first off-duty day and double (2) time for all hours worked during the second off-duty day.
 - (2) Time and one-half (1-112) shall be paid for hours worked on Saturday except for employees working on seven (7)day continuous operations. Double (2) time shall be paid for hours worked on Sunday except for employees working on seven (7) day continuous operations.
- (E) Daily and Weekly Overtime Pay
 - (1) An employee shall be paid time and one-half (1-112) for all hours worked in excess of eight (8) in one (1) day and hours worked in excess of forty (40) in the workweek.
 - (2) This clause (E) shall not apply to shift changes including changes in rotating shifts and/or continuous operations.

(3) If an employee formally requests and the supervisor at his/her discretion gives prior approval, an employee's shift may be changed during the workweek and there will be no overtime penalty paid by the Company.

(F) Overtime Compounding

No employee shall receive overtime pay twice for the same hours. No employee shall receive overtime pay and holiday premium pay for the same hours. No employee shall receive overtime pay and Sunday premium pay for the same hours.

(G) Shift Hours

- (1) Regular shift hours shall be as follows:
 - (i) First Shift 7:00 a.m. until 3:30 p.m. with a thirty (30) minute unpaid lunch period.
 - (ii) Second Shift 3:30 p.m. until 12 midnight with a thirty (30) minute unpaid lunch period.
 - (iii) Third Shift 10:30 p.m. until 7:00 a.m. with a thirty (30) minute unpaid lunch period.
- (2) When employees are working continuous operations the regular shift hours shall be as follows:
 - (i) First Shift 7:00 a.m. until 3:00 p.m.
 - (ii) Second Shift 3:00 p.m. until 11:00 p.m.
 - (iii) Third Shift 11:00 p.m. until 7:00 a.m.
- (3) Third Shift Starting Times:
 - (i) It is understood that the third shift will start at 10:30 p.m. and/or 11:00 p.m. Sunday night, and it is agreed there will be no overtime pay for such hours worked on Sunday.
 - (ii) A holiday for third shift employees will begin at the start of the shift the day prior to the calendar holiday. The third shift may begin work following a holiday as early as 10:30 p.m. the calendar day of the holiday, and it is agreed there will be no holiday premium pay.

- (iii) For third shift employees on non-continuous operations who start their week on Sunday night, that and each succeeding shift will be considered to have begun on the calendar day following.
- (4) When the Company designates a variation in shift hours from the above the Company will give notice to the employees affected thereby at least twentyfour (24) hours prior to the change but in no case shall such a notice be required if an employee is directed to report to work prior to his/her shift or to continue beyond the end of his/her shift.

14.03 Paid and Unpaid Lunches and Number of employees on Continuous Shift

- (a) An employee scheduled to work eight (8) hours within eight and one-half (8-1/2) consecutive hours shall be allowed a thirty (30) minute unpaid lunch period.
- (b) An employee scheduled to work eight (8) hours within eight (8) consecutive hours on a continuous shift operation shall be allowed a twenty (20) minute lunch period and will be paid at the employee's regular hourly rate for an hourly paid worker or the occupational rate if an incentive worker.

Determination of Number of Employees on Continuous Shifts:

- (1) When three (3) shifts are worked, the least number of employees within the same classification and department on any one (1) shift will be used to determine the minimum number of employees within the same classification and department who will be scheduled to work on a continuous shift basis on the other two (2) shifts.
- (2) The number of employees determined to be on a continuous shift under the provisions of this Article will be based on the number of employees on continuous shift on Monday of each week. Any change in the number of employees on continuous shift because of either an increase or decrease in the number of such

employees shall only be made on Monday of each week, except that if an entire shift within the same classification and department is added or eliminated, then such change would be effective on the date the change is made

(3) First and third shift employees who are on a continuous shift will remain on a continuous shift when working their sixth (6th) working day. Employees who work second shift Saturday will remain on a continuous shift provided both the preceding third shift and first shift Saturday worked.

14.04 Shift Premiums

(a) Employees scheduled to work the second shift will receive a shift premium of 48.6 cents per hour for each hour worked on such shift.

Employees scheduled to work the third shift will receive a shift premium of 72.2 cents per hour for each hour worked on such shift.

The foregoing shift premiums will be paid only to employees who are actually regularly scheduled to work such shifts and the 48.6 cents premium shall not be paid to employees on the first shift whose work continues into the second shift nor will the 72.2 cents premium be paid to employees on the second shift whose work continues into the third shift, unless such employee works two (2) full consecutive shifts.

(b) Employees scheduled to work irregular shifts, eg. at times other than listed in Article, 14.02(G), will have their shift premium determined by the shift in which the majority d their hours fall.

14.05 Rest Period

The Company will allow a ten (10) minute rest period during each half (1/2) shift. Hourly-paid employees will continue to receive their regular hourly rate during such rest period.

14.06 Personal Clean-Up Time

All employees may discontinue productive work ten (10) minutes before the end of the shift to perform necessary clean-up, etc. and may leave their work area five (5) minutes before the end of the shift for personal wash-up, etc.

14.07 Allocation of Overtime

When overtime is required to be performed by employees of the same classification in a department, area or focused factory, the Company will endeavour as far as practicable to equitably allocate the overtime among qualified employees who presently and normally perform the major or dominant portion of the work required. A list shall be posted weekly showing the hours of overtime worked or offered but not worked by each employee. New employees and employees entering a new classification, department or focused factory will be credited with the average overtime hours of the new group. New lists shall be started at the commencement of each calendar year. Employees who are absent from work for any reason shall be considered not available for the allocation of such overtime and shall not be entitled to be called in. When, upon the request of the employee, the Company excuses an employee from overtime work allocated to him, the employee will be considered to have worked for the purposes of this Article XIV. In the event the Company overlooks an entitled employee in making an overtime assignment such employee shall not be entitled to money payment for such missed Overtime assignment but be entitled to preference in a future overtime assignment or assignments in order to enable him/her to make up the lost overtime opportunity.

14.08 Overtime

- (a) Daily Overtime:
 - (i) Except as provided in (ii) below, whenever possible an employee will be given twenty-four (24) hours prior notice of an overtime assignment.
 - (ii) In a situation created by the absence of an employee on the succeeding shift, an employee

may be asked to work one (1) hour of daily overtime without prior notification.

(b) Saturday and First Off-Duty Overtime:

When Saturday and first off-duty day overtime is considered necessary, the employees required to work will be given notice on the calendar Wednesday preceding the Saturday and first off-duty day overtime. No employee will be required to work overtime for more than three (3) consecutive Saturdays or first off-duty days. For no more than five (5) occasions during a year (June to 31 May), an employee required to work overtime under this paragraph will be excused from a Saturday or first off-duty day overtime assignment provided he/she notifies the Company two (2) working days prior to the overtime day. Should the number of employees requiring time off under this provision on a given Saturday or off-duty day seriously interfere with production or maintenance requirements, the number of employees excused from overtime may be limited.

(c) Saturday Preceding or Following a Holiday:

The Company will not require an employee to work on a Saturday which immediately follows a Friday holiday or precedes a Monday holiday with the exception of employees who work in the Power House, Maintenance, Receiving or Repair Parts Shipping work areas.

(d) Sundays, Holidays and Second Off-Duty Days:

The Company will not require employees to work Sundays, holidays or second off-duty days as defined in Articles XIV and XXI, unless they are part of the employee's regularly scheduled workweek.

(e) Exceptions:

The provisions of Articles 14.08(a), (b), (c) and (d) hereof will not apply when overtime work is required to protect, maintain or repair the physical plant and/or equipment.

The foregoing does not change the Company's current practice of considering, wherever possible, personal

plans that an individual employee may have that conflict with his/her ability to work overtime.

14.09 Report-In Pay -Call-In Pay

- (A) Any employee who has worked on the previous work day and who reports for work at hisher regular time on hisher regular shift unless he/she has been told in advance not to report, or unless notice to not report is broadcast over the designated radio station or stations, and whose regular work is not available, or whose regular work becomes unavailable prior to his/her completion of four (4) hours of work from the start of hisher regular shift, shall be offered other work on the following basis:
 - (1) Employees shall receive not less than four (4) hours' work which shall be paid for at the rate of the job assigned, but not less than their regular wage rate.
 - (2) If the employee's job is not in operation because of an occurrence beyond the Company's control, such as fire, flood, or other weather conditions, explosion or power failure, the above provisions shall not apply.
- (B) Any employee who, having left the plant after his/her regular shift, is recalled to perform additional work shall be given the minimum of four (4) hours' work on the following basis:
 - (1) Employees shall receive not less than four (4) hours' work which shall be paid for at the rate of the job assigned, but not less than their regular wage rate.
 - (2) An employee so recalled will be paid the appropriate shift premium, if any, determined by hisher starting time for the call-in periodfor the time he/she works.
- (C) In either A or B above, such part of the four (4) hours worked shall be paid at the appropriate overtime rate, if overtime, and the unworked time shall be paid at straight time.

(D) The straight-time provided under this section for unworked time, if any, shall not cancel out any payment for unworked time that may be due under Article XXI referring to Holiday premium pay.

ARTICLE XV - WAGES

15.01 Basic Wage Rates

The schedule of basic rates listed in Exhibit "A" and "G-1", Rate Schedule, shall be effective during the term of this Agreement and subject to the terms of this Article. All of the cents per hour cost of living adjustments as determined by the application of Article 15.02 shall be applied to the appropriate basic wage rates and the resulting total wage rates shall then be adjusted to the nearest one-half (½) cent.

15.02 Cost of Living Adjustment

- (a) A cost of living allowance as set forth in this section shall be effective on the effective date of this Agreement.
 - (1) The cost-of-living allowance effective as of the effective date of this Agreement shall be forty-eight thousandths percent (0.048%) and shall be applied as follows:
 - (a) The cost-of-living allowance percentage shall be applied to the basic rates as shown in Exhibit "A" and Exhibit "G-1" attached hereto and the resulting cents per hour shall be applied to the appropriate total rates.
 - (2) The cost-of-living allowance provided for in Section 15.02(a)(1) above shall be adjusted up or down during the term of this Agreement in accordance with succeeding provisions of this Section on the basis of changes in the official Canada Consumer Price Index (1992=100) published by the Statistics Canada for Canada and hereafter referred to as the Consumer Price Index.

- (a) The cost-of-living allowance effective on the effective date of this Agreement is based on the average of the published Consumer Price Index for May, June and July 2007.
- (b) Thereafter, during the term of this Agreement, adjustments in the cost-of-living allowance shall be as follows:

Effective Date of Adjustment	Based upon the average of the Consumer Price Index for:
First pay period beginning on or after 1 December 2007 and at three calendar-month intervals thereafter to 1 June 2010	August, September and October 2007 and at three calendar-month intervalsthereafter to February, March and April 2010

(c) The amount of cost-of-living allowance which will be effective for any three-calendar-month period as provided in Paragraph 15.02(a)(2) above, shall be made in accordance with the following table.

CONSUMER PRI	CE INDEX		
(1992=100)			
AVERAGE OF THE PUBLISHED		COST-OF-	
INDEX FOR THE	_	LIVING	
CALENDAR MONTHS		ALLOWANCE	
	132.89 or less	0.000%	
132.90	132.99	0.048%	
133.00	133.09	0.096%	
133.10	133.19	0.144%	
133.20	133.29	0.192%	
133.30	133.39	0.240%	
133.40	133.49	0.288%	
133.50	133.59	0.336%	
133.60	133.69	0.384%	
133.70	133.79	0.432%	
133.80	133.89	0.480%	
133.90	133.99	0.528%	
134.00	134.09	0.576%	
134.10	134.19	0.624%	
134.20	134.29	0.672%	
134.30	134.39	0.720%	
134.40	134.49	0.768%	
134.50	134.59	0.816%	
134.60	134.69	0.864%	
134.70	134.79	0.912%	
134.80	134.89	0.960%	

134.90	134.99	1.008%
135.00	135.09	1.056%
135.10	135.19	1.104%
135.20	135.29	1.152%
135.30	135.39	1.200%
135.40	135.49	1.248%
135.50	135.59	1.296%
135.60	135.69	1.344%
135.70	135.79	1.392%
Etc	Etc.	Etc

- (d) The cost-of-living allowance percentage provided for above shall be applied in accordance with Section 15.02(a)(1)(a).
- (e) In the event Statistics Canada does not issue the Consumer Price Index on or before the beginning of the pay period referred to in Paragraph 15.02(a)(2), any adjustments required shall be made at the beginning of the first pay period after issuance of the Index.
- (f) No adjustments retroactive or otherwise, shall be made due to any revision which may later be made in the published figures of the Consumer Price Index.
- (g) The continuance of the cost-of-living allowance adjustments is dependent upon the availability of the official monthly Consumer Price Index (1992=100) in its present form and calculated on the same basis as the Index for September 2007.

15.03 Job Classifications and New Job Classifications

(a) The listing of job classifications and pay levels for wage occupations applicable on the effective date of this Agreement are contained in Exhibit "C" which is attached hereto and made a part hereof.

(b) New Job Classifications

When the Company establishes a new job classification coming within the scope of this Agreement the Company will notify the Chairperson of the rate established for such new job. Upon request the Company will meet with three (3) members of the Committee and discuss the basis upon which such rate has been established. If there is no agreement with respect to such rate, a Union policy grievance may be filed in accordance with Article 9.11 hereof. The mutually agreed upon arbitrator shall have an industrial engineering background and shall have authority to determine what the rate shall be but in doing **so** shall be governed by the relative value of the other classifications covered by this Agreement.

15.04 Establishment of Rates of Pay:

- (A) The rates of individual employees hired after the effective date of this agreement shall be determined by the rate chart attached hereto as Exhibit "G-1" Rate Schedules.
- (B) The rate of an employee transferred within the bargaining unit after the effective date of this Agreement shall be as follows:
 - (1) If transferred to a job classification in the same pay level the employee's current rate.
 - (2) If transferred to a job classification in a higher or lower pay level, the rate of pay for that level based upon Exhibit "A" or Exhibit "G-1", Rate Schedules.
- (C) An employee's work assignment is the classification(s) of work to which he/she is assigned as his/her regular

- work and may involve a specific machine(s) process(es), equipment and/or work area(s).
- (D) Definition: Temporary Assignment (not a transfer or formal reclassification): When, upon its completion, the employee is expected to return to his/her former assignment. Except for a Special Duration Job Bid (12.05(B)) a Temporary Assignment will not exceed sixty (60) days.
- (E) When an employee is temporarily incapacitated from performing his/her regular work and is assigned by the Company to another work assignment, such work assignment will be considered a temporary assignment under D above, but will not extend beyond the duration of the period the employee would have been eligible to receive Weekly Indemnity Benefits had the incapacity been temporary total. The employee will be paid the wage rate of the temporary assignment or the wage rate of his/her regular assignment, whichever is higher. If the employee is still so assigned at the end of the period of time for which he/she would have been eligible for Weekly Indemnity had the incapacity been temporary total, the employee will be transferred and reclassified to the work assignment and paid accordingly.

15.05 Continuous Improvement Process/Pay System

A Joint CAW/Deere Continuous Improvement Process has been implemented. Only through continuous improvement can the Company remain competitive and employees enjoy secure employment and premium wages. In order to meet these objectives, the Deere and CAW leadership recognizes the importance of establishing and supporting a Joint CAW/Deere Continuous Improvement Process. This is required to ensure that a culture exists across the business and within the union that promotes continuous improvement. This structured process is outlined in the CAW/Deere Continuous Improvement Process Letter of Understanding.

The Continuous Improvement Pay System (CIPP) rewards teams of employees for helping achieve continuous improvement of the operations to which they are assigned. Continuous improvement requires a cooperative effort from all employees, and provides employees the opportunity to both increase their earnings by sharing in these improvements and maintain a consistent weekly pay level.

A. Outline of the Continuous Improvement Pay System

- (1) A Continuous Improvement Pay Plan provides incentive compensation to a team of employees for achieving continuous improvement on a weekly basis above the Base performance metric(s).
- (2) Weekly Plan Performance is a team's calculated weekly earnings level expressed as a percent. It is determined by increasing (or decreasing) the 115% weekly pay level for the team for the week by 67% of the percentage change in weekly results achieved, compared to each Base performance metric(s). When multiple metrics (e.g., quality, productivity, schedule performance, etc.) are used, each metric will be assigned a percentage weighting factor with the sum of the weighting factors equaling 100%. A Weekly Plan Performance is calculated for each metric as described above and then multiplied by its respective metric weighting factor. These individual metric calculations are then added together to arrive at the total Weekly Plan Performance.
- (3) Pay for an employee's attendance hours while participating in a CIPP application (input hours) within a given week is computed by multiplying the employee's CIPP base rate(s) times the Weekly Pay Level for the week. Weekly Pay Level for each CIPP application will be determined as follows:
 - a. The maximum Weekly Pay Level for a CIPP application is 115%. Weekly hours earned in excess of 115% will be allocated to the CIPP application's Reserve Fund.

- b. When Weekly Plan Performance for a CIPP application is between 100% and 115%, the hours required to build-up earnings to the maximum Weekly Pay Level for the plan's participants will be provided equally from the CIPP application's Reserve Fund Hours and the Company, if hours are available in the Reserve Fund.
- c. When a CIPP application's Weekly Plan performance provides an earnings level that is less than 100% of an employee's input hours times their CIPP base rate(s), the Company will provide build-up hours to a weekly pay level of 100% of their CIPP base rate(s) for a plan participant's input hours in the plan. Weekly earnings will be built-up further to the maximum Weekly Pay Level according to Section 15.05.A.3(b).
- (4) An Average Semester Plan Performance for each CIPP application, weighted by the plan participants' weekly input hours, will be calculated for each Base metric at the end of 26 consecutive calendar weeks (Plan Semester) from the start of the plan and each 26 consecutive calendar week period, thereafter.
 - a. When the Average Semester Plan Performance for a metric is less than 115%, its Base performance metric will not be changed for the next semester.
 - b. When the Average Semester Plan Performance for a metric exceeds the maximum weekly pay level of 115%, its Base performance metric will be adjusted by .33 of the percent of total team improvement in the ratio output over input as compared to the base. However in no case will the base adjustment exceed 2% in any plan semester. For example:

Output hours in a Semester: 9300Input hours in a semester: 10000

Current base: .900

Output/Input =
$$9300/10000 = .930$$

Base Adjustment = .0110

New Base = {base adjustment + 1) X old base

New Base = (1.0110) X .900

New Base = .910

In return for this adjustment, the Company will distribute a Continuous Improvement Award of additional paid hours to the plan participants at the end of the plan semester.

C.

If a Continuous Improvement Award is earned for a metric as provided in 4-b above during a current semester and there was а adjustment as provided in 4-b above, for that metric, each employee's Continuous Improvement Award will be determined by multiplying the employee's input hours for the semester times the percentage the Base was adjusted for the next semester (factored by the Base metric's weighting) times a multiplier based upon the following criteria: For those teams with a base adjustment less than 2%, the employee input hours for the semester will be multiplied by the base adjustment (less than 2%) times 1.5. For those teams with a base adjustment equal to 2%, the employee input hours for the semester will **be** multiplied by the base adjustment **(2%)** times 2. (For example, if an employee had 1000 input hours for the semester and there was a 2% adjustment of the base, and the base is weighted by **100%** because there is only one metric, the Continuous Improvement Award would be calculated as follows:

1000 (the input hours)
 X 2% (the same percentage the base was adjusted)
 X 2 (the additional premium for continuous improvement)

=40 hours.

- d. Remaining Reserve Funds Hours (accumulated weekly Reserve Fund Hours less hours used for build-up) will be distributed to the plan's participants at the end of the Plan Semester. The Reserve Fund Hours, if earned, will be distributed to the plan's participants based on a ratio of each participant's plan input hours to the total plan input hours during the semester.
- e. In the event the number of plan workweeks within a semester is less than twenty-four (24) workweeks the base and performance metric(s) will be adjusted on the basis of .33 of the total team improvement in the outputlinput ratio when compared to the base (not to exceed 2%) times the number of plan workweeks divided by twenty-four (24)

Example:

A CIPP plan has sixteen plan workweeks for the semester and ,33 of the Total Team improvement in output/input as compared to the base was 1.10%.

Base Adjustment = 1.10% x 16/24 = .73%

- (5) Each Base performance metric will be established using historical data, when available, or through comparative data and/or appropriate engineering techniques, such that weekly results equal to the Base performance metric will yield a Weekly Plan Performance of 115%.
- (6) Employees' time for the following situations should be excluded when establishing the Base performance metric and not be included as plan input hours when they occur.
 - a. Paid Lunch Hours
 - Authorized Union Business
 - c. Classroom Training
 - d. Catastrophic downtime
 - **e.** Special projects/assignments outside of plan operation
 - f. Medical assessments/physicals as directed by the Company, scheduled on a regular, predetermined basis. (i.e. monthly, semi-annual or greater – does not include individual First-Aid Treatment).
 - Regularly Scheduled Company meetings (not to include daily or routine Production/Safety meetings)
 - Activities related to Emergency Response Teams and Evacuation

(7) All attendance hours not included as plan input hours will be paid for at the appropriate Wage Rate unless otherwise provided for in this Agreement.

B. Maintenance of CIPP Applications

- (1) In addition to Base adjustments described in Section 15.05.A4(b), adjustments to Base performance metric(s) and/or output standards will be made for the following conditions:
 - a. The Company invests greater than \$50,000, or \$5,000 per plan participant, whichever is less, in a CIPP application area.
 - i. When an investment results in improved performance the adjustments will account for only 80% of the calculated improvement to allow employees time to adjust to the change.
 - When an investment results in deteriorating performance the adjustments will account for the full affect of the deterioration.
 - Both parties agree that a change is necessary to assure the on-going competitiveness of the operation.
 - c. Work is moved into the plan from another area or out of the plan to another area.
 - d. Direct Labour is added to or deleted from the operations.
 - e. Introduction of new or revised products.

When new or revised products are introduced into a plan, output standards for the new and revised products will be established based on the design and methods of manufacture for the new or revised product. At this time, the Company may update the output standards of all products produced by the plan participants. In this case, the ratio of output hours produced

to the plan participants' input hours (Base performance metric) will be adjusted to produce the same Weekly Plan Performance for this metric that was achieved prior to the change in the output standards.

- (2) Output Standards in CIPP applications will be established using procedures such as timestudy, standard data, plant data, video analysis, laboratory analysis, predetermined data systems, part family analysis or any combination thereof.
- (3) In the event that a CIPP application is not producing the desired results, the parties agree to meet, determine the extent of the problem, and develop solutions within the framework of the Continuous Improvement Pay System.

C. Standard Hour Plan Buyout

The SHP Conversion Factor which was established for eligible employees during the term of the 1995-1998 Agreement will be used as it was during that agreement for computing earnings for input hours in CIPP applications and for determining pay in special pay provisions identified in Article XV, Wages, Section 15.06, Special Pay Provisions. The SHP Conversion Factor will also be used in any successor pay plan that does not provide earnings opportunity comparable to the Standard Hour Plan.

15.06 Special Pay Provisions

- (1) In the following special conditions, an employee assigned to a CIPP plan shall be paid his/her Wage Rate X 115% for attendance hours that are not included as plan input hours.
 - (A) Classroom Training (other than employee break-in)
 - (B) Medical assessments/physicals as directed by the Company, scheduled on a regular, predetermined

- basis. (i.e. monthly, semi-annual or greater does not include individual First-Aid Treatment).
- (C) Regularly Scheduled Company meetings (not to include daily or routine Production/Safety meetings)
- (D) Activities related to Emergency Response Teams and Evacuation.
- (E) When temporarily assigned by your supervisor to break-in training in another CIPP Plan while his/her plan is in operation.
- (2) Assignment to Non-CIPP work while his/her regular CIPP plan is in operation and the employee would otherwise be scheduled to work. The employee shall be paid their Average Straight-time Hourly Earnings as computed according to Article XV, Section 10

15.07 Adjustment of Wage Rate – Temporary Assignment

An employee temporarily assigned to perform work other than his/her regular work assignment (not a transfer or formal reclassification and the employee is expected to return to his/her regular assignment upon completion of the work) will have his/her wage rate adjusted **as** follows:

- (A) CIPP assigned employee
 - Wage Rate of the temporary CIPP assignment or the Wage Rate of his/her regular assignment, whichever is higher.
- (B) Non-CIPP assigned employee
 - (1) Assigned to CIPP Work Wage Rate of the temporary assignment or the Wage Rate of his/her regular assignment, whichever is higher
 - (2) Assigned to other Non-CIPP Work wage rate of his/her regular assignment for the length of the shift, then the wage rate of his/her regular assignment or the wage rate of the temporary assignment, whichever is higher for any subsequent full days.

15.08 Daily Work Record

All work performed each day will be recorded by the employee in the manner prescribed by the Company and no work shall be recorded that was not performed on that day. There shall be no change made in such employee's record without his/her knowledge. The record shall, upon request, be available for employee reference.

15.09 Inventory Rate

When an employee is required to do inventory work, i.e., recording, counting, or work incidental to taking inventory, he/she will be paid his/her regular wage rate.

15.10 Computation of Average Straight-Time Hourly Earnings

The method of computing a wage employee's average straight time hourly earnings shall be as follows:

Divide the sum of all money earned during the four (4) calendar weeks preceding such date for hours worked, excluding shift differential premium and overtime penalty pay, by the total of such hours worked. Newly hired employees or employees transferred from salary payroll to wage payroll shall have their average earnings based upon the wage rate of the classification to which they are first assigned until the next regular period for average earnings determination as outlined below.

An employee's average straight-time hourly earnings rate shall be his/her average straight-time hourly earnings for the four (4) week period prior to the notice given to the Union, and shall be effective the first day of the pay period the program is initiated (as set forth in the Average Earnings Letter), and shall be his/her rate for the payment of average earnings for the four (4) weeks that follow introduction. The one (1) week lag shall be maintained during the Agreement. The annual inventory and/or vacation shutdown will be

disregarded in determining the four (4) week computed workweeks.

	Calculate the AE Rate using earnings	AE Rate Calculation Week	
	from:		
Week 1			
Week 2			
Week 3			
Week 4			
Week 5			
Week 6	Weeks 1-4	Week 6	First Week under this program
Week 7		Week 6	
Week 8		Week 6	
Week 9		Week 6	
Week 10	Weeks 5-8	Week 10	
Week 11		Week 10	
Week 12		Week 10	
Week 13		Week 10	
Week 14	Weeks 9-12	Week 14	*****
Week 15		Week 14	
Week 16		Week 14	
Week 17		Week 14	
Week 18	Weeks 13-16	Week 18	
Week 19		Week 18	
Etc.	Etc	Etc	

15.11 Productivity Bonus

A Welland Works Productivity Bonus shall be paid to an individual employee for each actual hour worked, but not for hours paid. The Bonus will be effective on the effective date of this Agreement and shall be \$1.06 for pay levels 1, 2 and 3 and \$0.56 for pay levels 4 through 9 for the agreement period.

ARTICLE XVI - RETIREMENT BONUS

16.01

(A) Eligibility

Effective on the Monday following ratification of this Agreement, each employee while actively employed who possesses thirty (30) or more years of service credit as provided in Appendix "A", John Deere Pension Plan for Hourly and Incentive Paid Employees, Article II, is an "eligible employee" thereafter for a Retirement Bonus under the provisions of this Article but ceases to be "an eligible employee" in the event and at the time of quit, death or discharge except as expressly provided for in this Article.

(B) Bonus

- (1) The Retirement Bonus will be paid to an eligible employee upon his/her retirement. The amount of Retirement Bonus will be \$5,008,00.
- (2) There shall be deducted from the Retirement Bonus any amount required by Canada Customs and Revenue Agency.
- (3) In the event an eligible employee dies before receiving a Retirement Bonus, the amount of Retirement Bonus which such employee would have received if he/she had retired on the date of death will be paid in a lump sum to the person or persons designated in, and in the same manner as provided in, the John Deere Group Life and Disability Insurance Plan for Hourly and Incentive Paid Employees.

ARTICLE XVII - LEAVE OF ABSENCE

17.01 General

"Leave of absence" shall mean an absence from work requested by an employee in writing and consented to by the Company in writing. All requests for personal leave of absence shall be made to the Manager of Human Resources in writing by the employee concerned and a letter shall indicate in full the reason for requesting the leave of absence. Any leave granted shall be in writing covering a specified period of time but in any case not to exceed one (1) month except in special circumstances. The granting or withholding of a leave of absence shall be in the sole discretion of the Company and shall be without pay or any other form of compensation. The employee shall not work in any other position during such leave of absence unless agreed to by the Company in writing. Upon the employee's return to work at the expiry of the leave of absence the Company shall reinstate such employee with accumulated seniority.

17.02 Disability Leave

An employee who, because of illness or injury, whether work related or not, requires absence from work shall furnish evidence of such illness or injury which may upon reasonable grounds include examination by a Company-appointed physician, if requested by the Company. The employee shall furnish supplementary medical evidence of disability, from time to time, as requested by the Company.

17.03 Union Business Leave

The Company will grant leave of absence, without pay or loss of seniority, credited service or continuous employment to not more than six (6) delegated Union members to attend Union conventions or conferences provided the Company is given reasonable notice when the request is made. No person granted leave of absence pursuant to the provisions of this Article XVII shall be absent for more than one (1) week at any given time nor shall any person granted leave

pursuant to this Article XVII be granted more than an aggregate of thirty (30)days leave per calendar year unless elected as a full-time or part-time President, Vice President or Financial Secretary paid by the Local or appointed as a National Representative. Necessary leave will be granted to one (1) employee per year to attend the Labour College of Canada.

17.04 Public Office Leave

The Company will grant a leave of absence without pay or other form of compensation to a seniority employee who is elected to public office. The term "public office" as used in this Agreement shall mean a municipal, provincial or federally elected public office.

17.05 Educational Leave

The Company will grant a leave of absence without pay or any other form of compensation, for a period of not more than one (1) year, without loss of seniority, to an employee who has one (1) year or more of seniority so that the employee can attend a recognized college, university, trade or technical school full-time, provided that the course of instruction is acknowledged by the Company to be related to the employee's employment opportunity with the Company. Before receiving the leave, or an extension thereof, the employee shall submit to the Company satisfactory evidence that the college, university or school has accepted him/her as a student and on the expiration of each semester or other school term, shall submit proof of attendance during such term. Such leave may be extended for additional periods not to exceed one (1) year each.

17.06 Pregnancy and Parental Leave

A pregnancy leave of absence will be granted to any employee who becomes pregnant. A parental leave of absence will be granted to any employee upon the employee becoming a parent. Such leave of absence cannot exceed a twelve (12) month period. The employee shall give at least two (2) weeks' written notice of the date the leave is to begin,

unless special circumstances (as defined in the Employment Standards Act) exist. An employee who wishes to end his/her leave prior to the scheduled date must give at least two (2) weeks' written notice of that date. This provision is not meant to conflict with any obligations which the Company may have under applicable laws.

17.07 Reinstatement at the end of all Leaves

At the end of a leave of absence the employee will be reinstated in his/her former classification or a classification to which he/she is entitled by his/her seniority standing and for which he/she is qualified.

ARTICLE XVIII - JURY AND WITNESS DUTY

18.01 Payment

An employee who is called for jury duty or who is subpoenaed and reports as a witness in court will be reimbursed the difference between his/her normal rate of pay (i.e., the regular hourly rate of the employee as a straight hourly paid worker or, if the employee is an incentive worker, his/her average straight time hourly earnings on a straight time, eight (8) hours per day, forty (40) hours per week basis excluding shift premium and all other premium pay), for necessary time lost because of such jury duty or service as a witness, and the amount of compensation received from other sources for such service. No payment will be made for any day he/she would not have been scheduled to work.

18.02 Eligibility

In order for an employee to be eligible the employee must:

- immediately notify his/her supervisor of the receipt of summons for jury duty or service as a witness, and
- (ii) furnish his/her supervisor with proper evidence of jury duty or service as a witness.

18.03 First and Second Shift Employees

If a first shift or second shift employee is excused from jury duty or service as a witness during the morning, the employee may report for work to finish the day. If the employee fails to reportfor work to finish the day, he/she will be reimbursed only for the actual hours lost from work up to the time he/she was excused from jury duty or service as a witness. If an employee is excused after 12:00 Noon, he/she will not be required to work that day.

18.04 Third Shift Employees

A third shift employee will be excused from work on either the shift preceding his/her service, or the shift immediately following the completion of his/her service at the option of the employee. The employee must notify his/her immediate supervisor of such election at least twenty-four (24) hours prior to being absent from work whenever possible.

18.05 Purpose of Hours Paid

Hours paid under this Article count as hours worked for purposes of determining when the employee has worked in excess of eight (8) hours in a day or forty (40) hours in a week.

18.06 Jury Duty During Vacation

If an employee is entitled to receive jury duty make-up pay and the employee serves during a period of time when the employee is on vacation, the employee will receive both jury service make-up pay and vacation pay.

18.07 Definitions

The term "witness" as used in this Article XVIII shall not include an interested party in a matter, and the term "court" as used in this Article XVIII shall include only a court of record and shall not include an administrative tribunal, arbitrator or arbitration board, or any proceeding including a court in which the Company is a party.

ARTICLE XIX - BEREAVEMENT LEAVE

19.01 Eligibility and Duration

When death occurs in his/her immediate family, i.e., grandmother, grandfather, granddaughter, grandson, father, mother, mother-in-law, father-in-law, brother, sister, spouse, son, daughter, son-in-law, daughter-in-law, stepchild, stepfather, stepmother, stepbrother, stepsister and also the employee's spouse's brother, sister and grandparents, or other relatives residing with the employee, an employee may request to be excused for any three (3) normal scheduled days of work (or for fewer days as the employee may be absent) during the three (3) calendar days (excluding Saturdays, Sundays and holidays) immediately following the date of death provided heishe attends the funeral.

19.02 Conditions of Payment

After making written application the employee will receive pay for any scheduled days of work for which he/she is excused (excluding Saturdays, Sundays and holidays, or in the case of seven-day operations, the sixth and seventh days of the employee's scheduled workweek) provided heishe attends the funeral. Payment will be made at the employee's straight-time hourly rate on the last day worked (or in the case of incentive employees, the employee's average straight time hourly earnings) exclusive of shift and overtime premium. Time paid will not be counted as hours worked for purposes of overtime.

19.03 First Day of Bereavement

An employee may elect to count the day of death rather than the day following death as the first day.

19.04 Delayed Burials

In a delayed burial situation the employee may elect to take compensated days of bereavement so they can attend the burial service. In this instance the days need not be consecutive but the compensated days may not extend beyond the day after the burial service. In order to use this the employee must receive prior approval of his/her supervisor.

19.05 Memorial Services

In the event the funeral is held outside Ontario and if memorial services are held in Ontario, an employee may elect to take compensated days of bereavement provided he/she attends the memorial services but the compensated days may not extend beyond the day after the memorial services.

19.06 Multiple Deaths

In case of multiple deaths, no time is allowed beyond six (6) days following the date, or last date of death.

19.07 Cremation

An employee will be eligible for bereavement pay in instances where the body is cremated if the employee attends a bona fide memorial service at a place of worship or a funeral home.

19.08 Bereavement During Vacation

If an employee is entitled to receive bereavement pay and the excused days fall when the employee is on vacation, he/she will receive both bereavement pay and vacation pay.

ARTICLE XX - EXCUSED PERSONAL ABSENCE

20.01 Eligibility and Payment

Beginning with the effective date of this Agreement and until the first Monday in October of each contract year, all employees with one (1) or more years of continuous employment will be eligible for six (6) days at eight (8) hours per day of excused personal absence to be taken in individual increments of not less than two (2) hours, if the

absence is for a reasonable cause and if the employee gives prior notice to his/her supervisor, where possible, and if the employee has worked some part of the (1 October to 30 September) year. An employee who, as of 1 October of each year thereafter, has not used the entire six (6) days will receive pay for the unused portion in lieu of time off.

An employee or his/her estate will be paid for unused portion as follows:

- (1) at his/her request in eight (8) hour increments; or
- (2) on the regular pay day preceding 1 October; or
- (3) at the time of retirement; or
- (4) at the time of death.
- (5) at the time of quit.

20.02 Computation

The pay allowances for this time will be computed at the employee's average straight-time hourly earnings rate.

20.03 With Vacation and a Holiday

Employees may take this six (6) days or any remaining unused portion of it off as additional time off in connection with their vacations. If a holiday falls within the combined period, it will be treated as a holiday falling during a vacation.

20.04 Personal Illness

If an employee who is eligible for excused personal absence time with pay desires to **do** so he/she can take this time during a personal illness resulting from sickness or accident under the following circumstances:

(1) If the circumstances permit (for example, a planned operation), the employee will notify his/her supervisor prior to leaving work that they are going to be absent and that they want to take excused personal absence time with pay beginning with the first day of such absence.

- (2) If it becomes likely that the employee's disability will prevent his/her return to work prior to 1 October, the employee will be paid in lieu of time off for any unused excused personal absence days.
- (3) Where an employee elects to take excused personal absence time with pay in connection with an illness, the employee is not disqualified from receiving Weekly Indemnity benefits for which he/she qualifies.
- (4) During a period of illness, if an employee is eligible to take vacation time and excused personal absence time with pay, such time must be taken consecutively and not concurrently.

ARTICLE XXI - HOLIDAYS

21.01 Holidays Observed

The Company will observe the following holidays:

New Year's Dav Good Friday Victoria Day (first Monday preceding May 25th)

Canada Day

Civic Day (first Monday in August)

Labour Ďav

Thanksgiving Day Remembrance Day

Christmas Eve Day

Christmas Day

Boxing Day

New Year's Eve Day

Non-recurring Holiday 02Jan2009

21.02 Computation

Employees (not suspended or discharged for good and just cause) will be paid for these holidays (as set out above) if thev:

were scheduled to work in the workweek in which the (i) holiday is observed, if they do not fail to work when

- requested on the holiday (except for a satisfactory reason), and
- (ii) if they work both the work day preceding and following the holiday unless absent for one of the reasons listed below:
 - (1) formal leave of absence for more than three (3) days as provided in Article 17.01 and Article 17.05, Leaves of Absence.
 - (2) plant-incurred injury.
 - (3) jury duty.
 - (4) confining illness of the employee or treatment by a physician or dentist, substantiated by a written statement from the attending physician or dentist.
 - (5) lay-off.
 - (6) death or severe illness in the employee's immediate family. (Immediatefamily is defined as grandmother, grandfather, granddaughter, grandson, father, mother, mother-in-law, father-in-law, brother, sister, spouse, son, daughter, son-in-law, daughter-in-law, stepchild, stepfather, stepmother, stepbrother, and stepsister; and also the employee's spouse's brother, sister, and grandparents.) If requested to do so, the employee will furnish substantiating evidence.
 - (7) a temporary condition of no work available due to such causes as material shortages or trouble with machines or equipment but not including cases arising from occurrences beyond the control of the Company such as floods, fire or other natural causes
 - (8) member of the Honour Guard.
 - (9) subpoenaed as witness.
 - (10) required to appear for examination by a jury commission.
 - (11) attend a funeral.
 - (12) involved in an accident.

- (13) wedding in immediatefamily.
- (14) graduation from senior high school, college or university of a member of immediate family.
- (15) riot situation other than labour disputes.

21.03 Criteria for 'Not Scheduled'

An employee as set out in Article 21.02 above will be considered scheduled for the workweek during which a holiday occurs unless he/she fails to work any part of such workweek because he/she was:

- on lay-off other than an unscheduled lay-off which does not exceed one (1) week;
- (2) in the military service;
- (3) absent because of an illness as set out in Article 21.02(ii)(4) above, including an accident or hospitalization which began more than ten (10) working days prior to the holiday;
- (4) on a leave of absence which began prior to the end of the preceding workweek.

21.04 Hours Paid

NotwithstandingArticle 21.02 above, employees who fail to work all or part of the work day preceding and/or following the holiday for reasons other than set out above will receive holiday pay in an amount determined by totaling the number of hours worked on each day and dividing the result by two (2)with a maximum of eight (8) holiday pay hours for each holiday.

21.05 Exceptions

An employee absent on either the work day preceding or following the holiday (not both) because no work was available because of an occurrence beyond the control of the Company, such as floods, fire, or other natural causes, shall receive holiday pay.

21.06 Day of Observance

All holidays falling on Sunday will be observed on Monday or Tuesday if Monday is also a holiday, and all holidays falling on Saturday will be observed on Friday or Thursday if Friday is also a holiday.

21.07 Computation

Employees who qualify will receive eight (8) hours holiday pay at their regular average straight time hourly earnings rate, plus the appropriate night shift premiums.

21.08 Weekly Indemnity or Workplace Safety and Insurance Benefits

If an employee qualifies for both holiday pay and Weekly Indemnity or Workplace Safety and Insurance Benefits for the same day, the payment for the holiday not worked will be reduced by the amount of such benefit.

21.09 Holidays at Christmas Shutdown

Subject to the other qualifications of this Article XXI, employees not required to work on the first Christmas shutdown day and/or the second Christmas shutdown day will be deemed to have worked those days under this Article for purposes of qualifying for holiday pay for the Christmas Eve, Christmas Day, Boxing Day, New Year's Eve and New Year's Day holidays, provided, however, this provision will not apply if the employee failed to work or was absent for reasons other than those set out in Article 21.02 above on both the last scheduled work day before the Christmas holidays and the first scheduled work day following the New Year holidays.

ARTICLE XXII - VACATION PLAN

22.01 Vacation Year and Weeks of Vacation

The vacation pay for a vacation year (May 1 of the current year to April 30 of the next year) is based upon the

employee's earnings in the fifty-two (52) week period preceding April 15 of the current year, and the length of the employee's continuous employment with the Company as of May 1 of the current year. Wherever in this Article XXII the date May 1 or April 15 appears it shall be deemed to mean the Monday nearest May 1 or April 15 as the case may be.

- (a) An employee having twenty (20) years or more of continuous employment as of 1 May of the current vacation year will receive a vacation of five (5) weeks, with vacation pay computed on the basis of ten percent (10%) of earnings as set out in Article 22.02 hereof.
- (b) An employee having fifteen (15) years or more but less than twenty (20) years of continuous employment as of 1 May of the current vacation year will receive a vacation of four (4) weeks, with vacation pay computed on the basis of eight percent (8%) of earnings as defined in Article 22.02 hereof.
- (c) An employee having ten (10) years or more but less than fifteen (15) years of continuous employment as of 1 May of the current vacation year will receive a vacation of three and one-half (3-1/2) weeks, with vacation pay computed on the basis of seven percent (7%) of earnings as defined in Article 22.02 hereof.
- (d) An employee having five (5) years or more but less than ten (10) years of continuous employment as of 1 May of the current vacation year will receive a vacation of three (3) weeks, with vacation pay computed on the basis of six percent (6%) of earnings as defined in Article 22.02 hereof.
- (e) An employee having one (1) year or more but less than five (5)years of continuous employment as of 1 May of the current vacation year will receive a vacation of two (2) weeks, with vacation pay computed on the basis of four percent (4%) of earnings as defined in Article 22.02 hereof.
- (9 An employee having less than one (1) year of continuous employment as of 1 May of the current vacation year will receive a vacation entitlement as outlined in the ESA, with vacation pay computed on the

basis of four percent **(4%)** of earnings as defined in Article 22.02 hereof.

22.02 Computation

- (a) In computing vacation pay, the employee's earnings will be the sum of his/her straight-time earnings for hours worked during the vacation earnings computation period, i.e., the fifty-two (52) weeks preceding 15 April of the current vacation year plus:
 - (1) any premium or overtime pay for hours worked during that period.
 - (2) any vacation pay based on hours worked during that period.
 - (3) any pay for unworked holidays during that period.
 - (4) any pay for bereavement and/or pay for excused personal absences during that period.
 - (5) any pay for Christmas shutdown during that period.
 - (6) any make-up pay received for jury service.
- (b) If during the vacation earnings computation period an employee has worked and received pay for at least 500 hours, the amount of the following benefits received, if any, during the vacation earnings computation period will be included in computing vacation pay.
 - (1) any Supplemental Unemployment Benefits and any Employment Insurance Benefits received during the period of eligibility for SUB.
 - (2) any Workplace Safety and Insurance Benefits arising out of or in the course of employment with the Company excluding payment for non-economic loss.
 - (3) any Supplemental Weekly Indemnity Occupational Benefits.
 - (4) any Weekly Indemnity Benefits.
 - (5) any payments received for Long-Term Disability.

22.03 Christmas Shutdown

- (a) The Company will schedule a Christmas Shutdown, except for employees who are required to work, beginning on Christmas Eve Day (or the day observed as the Christmas Eve Day holiday) continuing through New Year's Day (or the day observed as the New Year's Day holiday). Employees on the active payroll on 1 December of the year of the shutdown or employees who return to work after 1 December and who are not scheduled to work during this period and who had six (6) months or more of continuous employment as of 1 May of the current vacation year, will be compensated for the two (2) regular work days during this period which are not observed as holidays.
- (b) Employees in the following work areas may be required to work during the Christmas Shutdown period, i.e., Powerhouse, Maintenance, Receiving and Repair Parts Shipping. In those instances where only part of the employees in the above work areas, i.e., Powerhouse, Maintenance Receiving and Repair Parts Shipping, are scheduled to work during this period, employees will be offered work on the following basis:
 - First, to the senior employees in the classification within the department, before,
 - (2) Employees within the classification and department are required to work on the basis of reverse seniority, i.e., the least senior first.
- (c) The compensation for the two (2) days will be on the basis of .83 of a percent of the employee's earnings as set out in Article 22.02, minus required Canada Pension Plan and Income Tax deductions, and will be paid on the regular pay for the week in which the compensated days occur. Employees required to work during the two (2) days described above who are eligible for the .83 of a percent compensation will receive such compensation in addition to their earnings for hours worked during this period.

22.04 Vacation Bonus

- (a) Employees having six (6) months or more of continuous employment as of 1 May of the current vacation year and who are eligible for vacation pay will receive a vacation bonus of one hundred dollars (\$100.00) minus required Income Tax deductions and Canada Pension Plan deductions. The vacation bonus will be paid on the last payday in May.
- (b) The vacation bonus shall not be included as earnings for the purpose of computing vacation pay or any other benefits.
- (c) Employees who are attending school and their employment during the vacation year was limited to periods of school vacation or other periods of interruption in the student's school year will not be eligible to receive a vacation bonus.

22.05 Terminated Employees

In the case **of** an employee whose employment with the Company is discontinued prior to the eligibility date for vacations the provisions of the Employment Standards *Act* of Ontario will determine the amount of vacation pay, if any, to which he/she is entitled.

22.06 Deceased or Retired Employees

An employee who dies or an employee who retires during the vacation year will receive vacation pay provided he/she has worked some part of the vacation year, and this vacation pay will not be less than the entitlement under the provisions of the EmploymentStandardsAct of Ontario.

22.07 Continuous Employment

The term "continuous employment" as used in this Article XXII means the period of time from the date on which the employee reported to work in his/her last employment.

22.08 Vacation Shutdown

- (a) The Company may continue the practice of closing the plant sometime during the summer months (May through September) for vacation. Because this period may vary from year to year employees will be given as much advance notice of the vacation period as possible, but never less than thirty (30) calendar days.
- (b) All employees will be expected to take their vacation during the vacation period. During this period employees may be required to take vacation up to a maximum of four (4) weeks. For example, if the plant is closed for six (6) weeks, employees will only be required to take the first four (4) consecutive weeks as vacation. If the plant is closed for three (3) weeks, then those employees who are eligible for four (4) or five (5) weeks' vacation will be allowed to take the fourth (4th) and fifth (5th) week at the discretion of management with due consideration being given to the wishes of the employee.
- (c) If an employee is laid off on or before the qualifying date, then the employee will have hisher vacation scheduled to begin on the qualifying date. If an employee is laid off after the qualifying date, then the employee will have hisher vacation scheduled to begin on the date of his/her lay-off. If, however, the employee requests hisher vacation to be scheduled at a different period then the provisions of Article 22.10 will not apply.
- (d) An employee who is on lay-off at the time of the qualifying date for vacation may notify the Company prior to the start of the vacation year that he/she does not wish to take hisher vacation at the beginning of the vacation year. The employee's vacation will then be rescheduled.
- (e) An employee eligible for vacation may request at the start of the vacation year hisher vacation entitlement in single day increments up to a maximum of 10 days, except as provided in paragraph (b) above. Single days may be scheduled under the following conditions:
 - Single vacation days must be scheduled with mutual agreement between the employee and the

- supervisor two days in advance of the day requested.
- 2. Employees will be paid one fifth of their weekly vacation total for each single day of vacation.
- Single vacation days must be taken in full day increments.
- 4. Unused single vacation days will be scheduled according to the following schedule if not used by OI January of each year:
 - a. Unused days in excess of five (5) not scheduled by the employee as of 01 January of each year will be scheduled by the supervisor to be taken during the months of January and/or February.
 - b. All other unused days, not scheduled by the employee as of 01 March of each year, will be scheduled by the supervisor to be taken during the months of March and/or April.

22.09 Pay in Lieu of Vacation

An employee will not be entitled to vacation pay in lieu of vacation, except:

- (a) as provided by the *Employment Standards Act* of Ontario, or
- (b) if the employee has worked and received pay from the Company for less than 500 hours during the vacation earnings computation period, the employee at his/her option may elect to waive all or part of their vacation. This waiver must be in increments of not less than one (1) week, except that an employee eligible for three and one-half (3-1/2) weeks may waive the half (1/2) week. Upon receipt of written notice from the employee vacation pay will be paid in lieu of time off.

22.10 Holidays With Vacations

When one of the holidays found in Article 21.01 of this Agreement falls during an employee's vacation, then the employee's vacation will be extended by allowing one (1)

additional day **of** vacation. This extra day of vacation will be the next scheduled working day following the end of the employee's vacation unless this extra day **of** vacation is rescheduled at the discretion **of** the supervisor with due consideration being given to the wishes of the employee. There is no additional vacation pay for the extra day of vacation and the extra day of vacation is fully compensated for by the payment of holiday pay for the unworked holiday falling during his/her vacation.

22.11 Deductions

On ail vacation pay cheques Canada Customs and Revenue Agency requires the deduction of Income Tax and EmploymentInsurance and Canada Pension Plan.

22.12 Receipt of Vacation Pay

Employees will receive their vacation pay on the regular pay day for the week in which the vacation day occurs.

ARTICLE XXIII - BULLETIN BOARDS

23.01

The Company agrees to permit the Union to use designated bulletin boards for the posting of notices of Union meetings, recreational, educational and social affairs of the Union, Union appointments and Union elections and results, but shall not include political material. All such notices must be signed by the proper officer of the Union and submitted to the Manager of Human Resources for approval before being posted.

ARTICLE XXIV - SAFETY AND HEALTH

24.01 Safety Rules and Regulations

- (1) The Company will make adequate provision for the safety and health of all employees during the hours of employment.
- (2) The Company will make every reasonable effort to comply in a timely manner with all relevant legislation pertaining to Occupational Health and Safety.
- (3) The parties agree to co-operate in promoting measures to protect the health and safety of employees.
- (4) (a) If an employee encounters an unsafe condition he/she should report the condition to his/her supervisor immediately. The supervisor will investigate and determine a plan for corrective action.
 - (b) The Company recognizes an employee's right to refuse unsafe work that places him/her or another person in danger. The Company agrees that no employee will be disciplined for properly exercising this right.

24.02 Safety Committee

The Safety Committee will consist of no more than four (4) management members selected by the Company and no more than four (4) members selected by the Union. The duties of the members of the Safety Committee will be as set out in the *Occupational Health and Safety Act.* In addition thereto, the Safety Committee will:

- (1) Investigate all lost time accidents with one (1) Company and one (1) Union member of the Safety Committee and provide a summary of their investigation report and Workplace Safety and Insurance Board Form 7 to the Safety Committee. A complete Form 7 will be provided to the Chairperson or his/her designee upon receipt of a release form signed by the affected employee.
- (2) Be promptly notified of any fatalities, serious injuries or potential serious near misses, resulting from workrelated accidents.

- (3) Meet monthly to review the factory safety performance and make recommendations to Management to improve the safety program.
- (4) Conduct a health and safety inspection of various areas of the factory in conjunction with the monthly Health & Safety meetings.
- (5) Structure a training program for all new employees on safety and health including chemical hazard training.

One (1) Safety Committee member nominated by the Union will receive training to become certified as required by the Ontario Occupational Health and Safety Act. The Company will pay this member of the Committee for time spent in fulfilling the requirements for becoming certified.

24.03 Physical Agents or Toxic Materials

- (a) The Company will disclose the identity of all known physical agents or toxic materials to which workers are exposed.
- (b) The Company will provide to employees who are exposed to potentially harmful agents or toxic materials at no cost to them those medical services, physical examinations and other appropriate tests including audiometric and lung function examinations at a frequency and extent necessary to determine whether the health of such employees is being adversely affected.
- (c) The Company will provide to each employee or his/her physician, upon written request of the employee, a complete report of the results of any such tests or examinations and will review the test results with the employee prior to release to his/her doctor.

24.04 Protective Equipment

The Company will continue to furnish special protective equipment such as special gloves for inert gas welders (not ordinary items of work clothing) and special safety devices (not safety shoes and metatarsals or prescriptions for safety glasses) where they are required by the Company.

24.05 Reimbursement for Safety Shoes

The Company will reimburse an employee for one (1) pair of safety shoes per year provided proof of purchase is supplied to the Company.

The reimbursement will be One Hundred And Forty (\$140) per year.

24.06 Prescription Safety Glasses

The Company will continue its current practice of supplying prescription safety glasses, at no cost to the employee.

ARTICLE XXV - NEW TECHNOLOGY

25.01

In the event of the introduction of new technology which is likely to have a major impact upon employees the Company will give the Union notice thereof and will discuss with the Union the implications of such new technology including ways of helping employees adjust to the change which it occasions.

When new or greater skills are required by affected employees training will be provided to the number of such employees required in accordance with their seniority and qualifications.

ARTICLE XXVI - APPRENTICES AND SKILLED TRADES

26.01 Classifications

Apprentices may be enrolled in the following skilled trades job classifications:

Q-1 Power House Operator

R-5 Machine Maintenance

R-15 Repair Mechanic (Industrial Trucks)

- R-20 Electrician
- R-25 Combination Welder
- R-37 Pipe Fitting, Heating and Ventilating Maintenance
- T-1 Tool and Die Maker
- X-13 Instrument & Test Mechanic

26.02 Eligibility

- (a) Selection of apprentices under these programs shall be made from qualified applicants. When openings exceed the number of qualified seniority applicants, the remaining openings will be filled by applicants from outside on the basis of qualifications.
- (b) Applicants must be physically qualified to perform the work required.
- (c) Applicants must have successfully completed Grade 12 or its educational equivalent..

26.03 Seniority Employees

Notice of openings will be posted on the bulletin boards. Interested seniority employees may apply for the openings.

26.04 Application and Standards

Eligible applicants will be selected through the application of the standards as established by the Joint Apprenticeship Committee which will incorporate aptitude, education and related work experience. The apprenticeship opening(s) will be filled by the senior applicant who attains the designated minimum qualifications.

The established process will be reviewed and approved by the Manager of Human Resources and the Shop Chairperson.

(A) The number of apprentices enrolled in any one job classification shall be limited to fifteen percent (15%) of the total number of skilled workers employed in the particular job classification wherein the employee is to be enrolled provided, however, that the Company may enroll a minimum of four (4) apprentices in each job classification listed in this Article XXVI. Fractional amounts resulting from the application of the above percentage may be increased to the next whole number.

26.05 Seniority

- (a) Enrolled apprentices will be exempt from the seniority provisions of the Agreement and there shall be no obligation on the part of the Company to continue training, if, in the judgment of the Company, the progress of the apprentice is unsatisfactory.
- (b) When apprentices are removed from their training status for any reason, including graduation, and are assigned to other employment in the factory, they shall receive credit for seniority purposes for the time spent in the apprenticeship program.
- (c) Upon completion of the apprenticeship requirements the employee will be classified in the particular Trade job classification in which the employee had been enrolled. In the event there is no need for an additional employee in the classification, any reduction in force will be handled in accordance with the provisions of the Seniority Article. Upon completion of apprenticeship requirements, which includes the completion of a Certificate of Qualification, the apprentice will be granted seniority in their trade equal to the time spent as an apprentice.
- (d) When it is determined by the Company that it is necessary to remove apprentices from a program because of a surplus in an apprenticeship program, the employee with the least number of credited hours will be removed first.
- (e) When it is determined by the Company that it is necessary to increase the number of employees in an apprenticeship program, the employee removed under paragraph (d) above will be returned to the program in

- the order of the employee with the highest credited hours first. Employees removed under paragraph (d) above will be returned to their program before adding additional employees to that program.
- (f) A surplus employee reduced under paragraph(d) of this Article 26.05 will be eligible for Income Security Benefits if he/she is otherwise qualified for such a benefit as provided in Article XIII.
- (g) If layoffs become necessary within the apprentices' trade, apprentices will be reduced in order to maintain a 15% ratio. The apprentice will exercise their seniority rights within the production area.
- (h) The Apprentice may work overtime and will be limited to no more than the hours worked in their trades' overtime group.

26.06 Rates

(a) Apprentices in each of the job classifications covered by these standards will be paid at a progressively increasing schedule of wages, as follows:

1st 1000 hours --65% of the wage rate of the classification 2nd 1000 hours --70% of the wage rate of the classification 3rd 1000 hours --75% of the wage rate of the classification 4th 1000 hours --80% of the wage rate of the classification 5th 1000 hours --85% of the wage rate of the classification 6th 1000 hours --90% of the wage rate of the classification 7th 1000 hours --95% of the wage rate of the classification 8th 1000 hours --95% of the wage rate of the classification

- b) Applicants who are given credit for experience shall be paid the wage rate for the period to which such credit advances them.
- (c) Upon completion of the apprenticeship requirements, the wage rate will be increased to the wage rate for the particular job classification in which the employee had been enrolled.

26.07 Credit

- .(a) The applicant may be granted credit for experience in his/her trade acquired before entering the program. For applicants from the outside, such a work experience must have been gained under an apprentice or other indentured program and not trade or vocational school work. Seniority employees may be allowed credit for previous applicable work experience.
- (b) The Company at the completion of each 1000 hours will review with the employee his/her progress. The Company and the employee will sign off and date the completed skills acquired, if any. The employee's rate of pay, if necessary, will be adjusted at the time the hours are credited.

26.08 Related Classroom Study

(a) Apprentices will be required to complete a related training course through classroom work or by

- correspondence. The Company will determine the type of classroom study applicable in each situation.
- (b) Time spent in related training shall be in accordance with the schedules of work processes and skill sets referred to in Article 26.13. The time and place of such training will be determined by the Company and such time, if used, shall be paid at the apprentice's regular hourly rate. Tuition and course materials fees will be paid by the company.
- (c) In case of failure on the part of any apprentice to fulfill his/her obligation as to satisfactory completion of related classroom study, the Company may suspend or revoke his/her apprenticeship. Any such failure will be reviewed with the Joint Apprenticeship Committee.

26.09 Supervision of Apprentices

- (a) Apprentices shall be under the general direction of the individual charged with coordination of apprenticeship programs and under the immediate direction of the supervisor of the department to which they are assigned. The Coordinator of Apprenticeship Programs is authorized to move apprentices from one department to another in accordance with the predetermined schedule of work training or skill sets outlined. No apprentice may be retained unreasonably on a scheduled work process or skill set once the employee has demonstrated proficiency in that particular skill. When an apprentice feels that he/she is being retained unreasonably, he/she will review their concern with their supervisor, and if not resolved, will have the issue reviewed with the Joint ApprenticeshipCommittee.
- **(b)** Adequate record forms shall be prepared and will be filled in by the supervisor under whom the apprentices receive direct instruction and experience. Supervisors shall make a report at least every thirty (30) days to the Coordinator of Apprenticeship Programs on the work and progress of the apprentices under their supervision.

26.10 Tools

Apprentices shall procure the hand tools needed and as required and specified by the Company. Such tools may be purchased through the Company under a payroll deduction authorization.

26.11 Information

A record will be kept by the Company of time spent on the various work processes. An apprentice may be required to work shift work. Overtime worked by an apprentice shall not reduce the period of apprenticeship unless it is work which is covered by the "Schedule of Work Processes" or "Skill Sets" for apprentices, as the case may be.

26.12 Term of Apprenticeship Programs

The terms of apprenticeship programs shall be as established in accordance with the schedule of work processes and skill sets as outlined in Article 26.13.

26.13 Schedules of Work Processes/ Skill Sets

The Schedules of Work Processes are listed below. The sequence of the specific areas of training shall be determined by the Company on the basis of work available and the progress and needs of the apprentice. The continuity of the hours within an area may be broken and areas of training may be revised or replaced when in the opinion of the Company it is in the best interest of the apprentice to make such changes.

The skill sets for the trades are outlined in the Apprenticeship Training Standards as prescribed by the Workplace Training Branch of the Ontario Ministry of Training, Colleges and Universities (MTCU). The Company and the employee are required to sign and date the skills following each successful acquisition. The Company is required to attest to achievement by signing each skill set within each unit once the employee has demonstrated either competency or a prescribed summary of hours.

SCHEDULE OF WORK PROCESSES FOR APPRENTICES

R-5 Machine Maintenance

Electrical Maintenance		108 Hours
Grinder (external, internal or surfa	ace)	324 Hours
Heat Treat		72 Hours
Hydraulics and Pneumatics		1728 Hours
Lathe		420 Hours
Milling Machines		418 Hours
Repair and Overhaul		3592 Hours
Shaper, Planer or Slotter		180 Hours
Tool Crib		108 Hours
Welding and Fabrication		350 Hours
Related Classroom Study		700 Hours
	TOTAL .	8000 Hours

R-15 Repair Mechanic (Industrial Trucks)

Performs Repair and Maintenance of	
Internal Combustion Engines, Hydraulic and Electrical Systems and Controls	2140 Hours
Transmission, Clutches, Differentials.	540 Hours
Chassis, Suspension, Springs, Axles	200 Hours
Steering and Brakes	200 Hours
Fuel, Oil and Cooling Systems	160 Hours
Tires and Batteries.	40 Hours
Welding and Fabrication	160 Hours
Use and Operation of Specialized	160 Hours
Equipment and Tools	
Crib Functions-Location and	40 Hours
Identification of Supplies and Equipment.	
Related Classroom Study	360 Hours
TOTAL —	4000 Hours

R-20	Electrician		
Electrical Cons	truction		2448 Hours
	owerhouse, su machines and gn		288 Hours
Machine Maint	enance(hydrauli	cs)	216 Hours
Maintenance a	nd Repair		4348 Hours
controls, we equipment, po motors, motor	wer supply, tran repair, generato ttors and pow	treating sformers, rs, hoists, .	700 Hours
		TOTAL	8000 Hours
R-37 Maintenance:	Pipe Fitting	, Heating	and Ventilating
Maintenance:	Pipe Fitting		and Ventilating
Maintenance: Install, Maintai Steam, and Ho Valves, Stea		following: Systems,	and Ventilating 1600 Hours
Maintenance: Install, Maintai Steam, and He Valves, Stea Radiators and Install, Mainta Piping System	n and Repair the ot Water Heating ım Traps, Ai	following: Systems, r Vents, Industrial ss Piping,	
Maintenance: Install, Maintai Steam, and He Valves, Stea Radiators and Install, Mainta Piping System Refrigeration Systems Install and Ma	n and Repair the ot Water Heating Im Traps, Ai Pipe Fabrication ain and Repair as such as Proce	following: Systems, Vents, Industrial SS Piping, Onditioning	1600 Hours

500 Hours

Maintain Heat Treat, Furnaces, Gas Trains and Equipment

OHSA/Safety Training		600 Hours
Related Classroom Studies		720 Hours
Miscellaneous		480 Hours
	TOTAL	8000 Hours
T-1 Tool and Die Maker		
Bench Work and Optional layout, build and inspect)	(including	2750 Hours
Grinder - Surface		300 Hours
Heat Treat		100 Hours
Lathe		1200 Hours
Metallurgy		100 Hours
Milling Machines		950 Hours
Shaping, Planer or Slotter		300 Hours
Power Saw—Contour		100 Hours
Drill Presses		250 Hours
Tool Crib		100 Hours
Tool Design and Follow Up		800 Hours
Tool Grinding		200 Hours
Welding		150 Hours
Related Classroom Study		700 Hours
	TOTAL	8000 Hours
X-13 Instrument	t& Test Mechanic	;
Build, Calibrate and	Try Out	1300 Hours
Instrumentation & Test Electronic	Equipment:	
Electrical		1300 Hours

900 Hours

Hydraulic

Mechanical			700 Hours
Maintain Instrumenta Equipment	tion &	Test	1600 Hours
Familiarization on Instru Equipment: Stress Coat Tests			200 Hours
Dynamometer Tests			208 Hours
Environmental, Vibration	, Sound	Tests	200 Hours
Performance, Efficiency,	Durabili	tyTests	200 Hours
Related Classroom Stud	y		1392 Hours
		TOTAL	8000 Hours

26.14 Joint Apprenticeship Committee

A Joint Apprenticeship Committee composed of two (2) persons appointed by the Company and two (2) persons appointed or elected by the Union shall meet once a month to discuss matters arising pursuant to the provisions of this Article XXVI provided that an agenda is submitted by either party to the other at least three (3) working days prior to the date of the requested meeting.

It shall be the duty of the committee to:

- i) Review the application and selection process & documentation for apprenticeships. It is understood, however, that the final selection and hiring of the apprentices is the sole responsibility of the Company.
- il) See that each prospective apprentice is interviewed and impressed with the responsibilities he/she is about to accept as well as the benefitshe/she will receive
- iii) Accept or reject applicants for apprenticeship subject to final approval by the Company
- iv) Hear and decide on questions involving apprentices which relate to their apprenticeship
- Offer constructive suggestions for the improvement of training on the job which includes the Schedule of Work Processes

- vi) Review the Supervisors monthly report on each apprentice
- vii) Be responsible for the successful operation of the apprenticeship standards in the plant and the successful completion of the apprenticeship by the apprentices under these standards.

The Manager, Human Resources and the Shop Chairperson will review and approve any changes to these duties.

26.15 PROVISIONS FOR SKILLED TRADES

(A) PURPOSE

(1) The purpose of this provision is to define matters dealing with skilled trades classifications covered by this agreement. The provisions of the Collective Agreement and Benefits Plans shall apply to employees in the trades classifications except as specifically modified by the provisions of this provision.

This provision shall apply to employees listed in the trade classifications below:

Q001	Power House Operator
R005	Machine Maintenance
R015	Repair Mechanic - (Industrial Trucks)
R020	Electrician
R025	CombinationWelder
R037	Pipefitting, Heating & Ventilating Maintenance
T001	Tool & Die Maker
X013	Instrument and Test Mechanic

(2) The skilled trades covered by this provision constitute those trades for which an apprenticeship is usually served together with those classifications which form a pari of an apprenticeable trade. A journeyman/journeywoman in any of the trades classifications shall mean any person who:

- (i) has served a bona fide apprenticeship of four (4) years (8000 hours) and possesses proof of such apprenticeship service and has completed a Certificate of Qualifications, or
- (ii) holds a recognized CAW journeyman/journeywoman card in the trade in which recognition is claimed and has completed a Certificate of Qualifications, or
- (iii) has eight (8) years practical and general experience covering all phases of the apprenticeship course applicable to the trade in which journeyman/journeywoman status is claimed and possesses ample proof of such experience and has completed a certificate of Qualifications
- (3) Entry into a trades classification shall be restricted to persons:
 - who qualify as journeymadjourneywoman under the provisions set forth in the immediately preceding paragraphs, or
 - (ii) who qualify for journeyman/journeywoman status through the apprenticeship program outlined in Article 26.13
 - (iii) who provide documents at date of hire proving their claim to journeyman/journeywoman status to the Company

(B) SENIORITY

(1) Employees in trades classifications or who have recall rights thereto as of 30 September 2004 shall retain seniority in their trade as of that date and shall continue to accrue seniority in their trade thereafter. New entrants into a trades classification shall have seniority in their trade only as from the date of entry.

- (2) Employees in trades classifications will exercise seniority within their own trade. For example, if a reduction in the number of TCO1 Tool & Die Makers is required, the T001 Tool & Die Maker with the least trade seniority shall be laid off first and the employee with the most trade seniority shall be the first to be reinstated
- (3) Notwithstanding any other provisions of 30 September 2004, an employee with seniority in a trades classification as of the date of the agreement who is affected by a lay off shall be permitted to bump into a job classification outside of his/her trade with the same seniority date that he/she held immediately prior to attaining seniority in their trade and will be eligible for recall to their trade in accordance with the provisions of paragraphs 1 and 2 above. This provision will be exercised when given notification of layoff.

A trades employee will be permitted a maximum of one (1) such bump/recall following which, such employee shall not acquire or retain seniority rights to bump into any classification outside of his/her own trade. A trades employee who bumps into a job classification outside of his/her trade and refuses recall to his/her trade, will have seniority credited in accordance with the provisions of Article 12.01 but will lose seniority in his/her trade.

The Chairperson may request the Manager of Human Resources to arrange a special meeting concerning problems in connection with the work assignments of employees in trades classifications. Such special meeting may be attended by the skilled trades committeeperson, a representative of management in charge of the trades activities involved and a representative of the Human Resources department. If the meeting doesn't resolve the concern, the issue

may be forwarded to Step two of the grievance procedure.

(C) CONTRACTING OUT

When trades are on layoff the Company will endeavor to have work performed by the bargaining unit. However, due to special skills or short-term completion times, the Company may be required to bring contractors into the factory to perform work. The Company will continue its practice of notifying the Shop Chairperson of Contractors coming into the factory.

(D) TOOL ALLOWANCE

Skilled trades employees and apprentice will be provided an annual tool allowance of \$100.00 in October of each calendar year.

(E) CANADIAN SKILLED TRADES COUNCIL

The Company will deduct Canadian Skilled Trades Council dues from skilled trades employees as may be adopted by such Council and the Company will be notified in writing by the Union of the amount such dues. Dues deductions will be made in January of each year (or upon completion of one (1) month's work in that calendar year) and remitted to the Council Financial Secretary within fifteen (15) working days of the deduction along with a list of names from whom such deductions have been made and the amount of each deduction.

ARTICLE XXVII - PLANT CLOSURE

27.01

In the event of the permanent discontinuance of all or a substantial part of the business carried on by the Company at Welland, the Company will advise the Union of the decision as far in advance as is reasonably possible. Upon the request of the Union representatives of the Company will meet with the Union and discuss the reasons for the closing or discontinuance.

ARTICLE XXVIII - WAIVER

28.01

This Agreement constitutes the entire Collective Agreement between the parties and concludes collective bargaining for its term and can only be amended or supplemented by mutual agreement of both parties.

ARTICLE XXIX - APPENDICES

A The parties have provided the benefits contained in the following Appendices which are made parts of this agreement and subject to all the provisions of this Agreement.

Appendix "1" - Benefits Plans John Deere Limited
Appendix "A" - John Deere Pension Plan for Wage

Employees

Appendix "B" - The Health Benefit Plan for Wage Employees

Appendix "C" - The Disability Benefit Plan for Wage Employees

Appendix "D" - Supplemental Unemployment Benefit Plan Preliminary Provisions

Appendix "D-1"	- Supplemental Unemployment Benefit Plan
Exhibit"B"	 Automatic Short Workweek Benefit Plan
Appendix "F"	- Tuition Refund Plan
Appendix "G"	-John Deere Group Life and Disability Insurance Planfor Wage Employees

Appendix "H" - Profit Sharing Plan

Appendix "I" - John Deere Savings Plan for Wage Employees

ARTICLE XXX - DURATION

30.01 Effective Dates

This Agreement shall become effective 01 October 2007 following the Company receiving notification from the Union that the Agreement has been ratified and shall remain in force and effect and shall not be reopenable save and except as otherwise herein expressly provided until the 30th day of September, 2010, and shall continue automatically thereafter during periods of one (1) year each, unless either party notifies the other party in writing as provided for in clause 30.02 hereof of its desire to negotiate amendments to this Agreement.

30.02 Amendments

Notice that amendments are required shall only be given during the period of not more than ninety (90) days and not less than sixty (60) days prior to the 30th day of September, 2010, or during similar annual periods thereafter. If notice of desire to amend this Agreement is given by either party in accordance with the foregoing, the other party agrees to meet for the purpose of negotiation.

DATED AT WELLAND, ONTARIO THIS 1st DAY OF October, 2007.

FOR THE UNION:	FOR THE COMPANY:
John Gagnon	James Rottinghaus
Scott Anger	Frank Mattei
Gord Digout	Kiersten Crane
Venanzio D'Addario	Pamela Snyder
Roger Demers	DavidHarkin
John Wilgosh	Colin Baker
Ed Kowalczyk	
Thomas R. Napper	
Mike Menicanin	

EXHIBIT "A' Rate Schedule

Pay Level	CIPP RATES	NON-CIPP RATES
1	22.820	25.330
2	22.220	24.665
3	21.205	23.535
4	20.480	22.450
5	19.670	21.815
6	19.005	21.070
7	18.350	20.290
8	17.850	19.555
9	17.360	18.935

EXHIBIT "C" SENIORITY CLASSIFICATIONS

Seniority Unit	Pay Level	Occ Code	Job Title	Group
950	5	B025	Punch Press Operator Hot	A
950	6	C060	Metal Fabricator	Α
963	6	D102	Manufacturing Cell Operator	Α
963	6	D103	Manufacturing Cell Operator	Α
959	5	F001	Manual Arc Welder	Α
959	5	F002	Robotic Welder Operator	Α
915	5	F050	Setup Welding	В
961	6	K008	Assembler (Assembly Check Chart)	A
961	7	K011	Assembler (Assembly Check Chart)	Α
916	6	K013	Assembly Setup	В
920	6	L013	Automated Paint Technician	В

Pg

Seniority Unit	Pay Level	Occ Code	Job Title	Group
903	2	N021	Layout and/or Checker	В
902	2	Q001	Power House Operator	В
902	1	R005	Machine Maintenance	В
902	4	R009	Building Repair and Maintenance	В
902	2	R015	Repair Mechanic (Industrial Trucks)	В
902	7	R019	Maintenance Oiler	В
902	1	R020	Electrician	В
902	3	R025	Combination Welder	В
902	2	R037	Pipefitting, Heating & Ventilating Maintenance	В
900	1	T001	Tool and Die Maker	В
900	3	T009	Tool and Die Repair	В
903	5	U001	Inspector	В

Pg 106

Seniority Unit Pay	Pay Level	Occ Code Job Title		Group
903	4	U012	Inspection- Special Investigator B	
606	8	V002	Industrial Fork Truck Operator B	
905	9	V003	Semi-Trailer Truck Driver	
913	7	V013	Overhead Crane Operator B	
913	7	V015	Steel Storage Stocker B	
911	6	V017	Material Handler B	
905	9	W001	Warehouser or Shipping Checker	
905	7	W002	Warehouser or Shipper	
901	æ	W011	Stores and Tool Crib Attendant B	
206	8	W030	Materials Stocker B	
904	3	X007	Mechanic B	
304	C	onnv	а	

Seniority Unit	Pay Level	Occ Code	Job Title	Group
904	1	X013	Instrument and Test Mechanic	В
904	3	X045	Welder	В
907	8	Z014	Stock Chaser	В
910	9	Z022	Mechanical Floor Cleaner - Driver	В
901	7	Z030	Production Tool Service	В
918	3	Z034	Welding Tryout (Methods Department)	В
911	8	Z043	Inventory Counter or Checker	В

Pg 108

EXHIBIT "F" RED-CIRCLE RATES

A. Individual Red Circle Rates

- So long as an employee listed herein is employed in the job classification(s) listed opposite his/her name, his/her "red-circle" basic rate shall be retained for the duration of the agreement subject to all of the provisions of this Exhibit"F".
- 2. When an employee covered by this Exhibit "F", ParagraphA-1 (above) is reduced from the classification listed opposite his/her name, this ParagraphA will no longer be applicable. If the employee is restored to the classification opposite his/her name, ParagraphA shall be reinstated. An employee will lose all rights under this paragraphA above if the employee loses his/her rights to the classification listed opposite his/her name in this paragraphA.

B. Temporary Assignments

In the application of Article 15.07B(1), the wage rate of their regular assignment will be the wage rate shown on an employee listing showing individual Red-Circle rates, maintained in the Human Resources Department.

Note: An employee listing showing Individual Red-Circle rates will be maintained in the Human Resources Department.

EXHIBIT "G"

SPECIAL PROVISIONS FOR EMPLOYEES HIRED ON OR AFTER 16 MARCH 1999

Section 1. Basic Wage Rates and Total Wage Rates

For each year of work from the date of hire (52 workweeks in which an employee receives pay as interpreted under Appendix "D"), each newly hired employee will have their total wage rate determined as provided for in EXHIBIT "G-1", newly hired employees changing pay levels will remain in the same service step in the schedule.)

Section 2. Productivity Bonus

Newly hired employees will be eligible for the applicable Welland Works Productivity Bonus after each employee has worked six years from the date of hire (312 workweeks for which the employee receives pay as interpreted under Appendix "D").

Section 3. Excused Personal Absence

Newly hired employees with one (1) year of continuous employment as of the first Monday in October will be eligible for one (1) day at eight hours per day of excused personal absence to be taken as provided in Article XX – EXCUSED PERSONAL ABSENCE. For each additional year of continuous employment as of the first Monday in October, each employee will be eligible for an additional day at eight hours per day (until the employee is eligible for all of the days provided in Article XX – EXCUSED PERSONALABSENCE).

EXHIBIT "G-1"

BASIC WAGE AND PROGRESSION TABLE FOR EMPLOYEES HIRED ON OR AFTER 16 MARCH 1999

000000000		20000000		,			,					
AFTER	312	WORK	WEEKS	21.345	20.780	19.830	19.155	18.390	17.765	17.170	16.685	16.250
AFTER	260	WORK	WEEKS	20.520	19.985	19.065	18.420	17.690	17.090	16.505	16.050	15.625
AFTER	208	¥ORK	WEEKS	19.710	19.190	18.310	17.685	16.975	16.400	15.840	15.410	15.005
AFTER	156	WORK	WEEKS	18.885	18.380	17.545	16.945	16.270	15.720	15.180	14.775	14.375
AFTER	104	WORK	WEEKS	18.060	17.585	16.780	16.205	15.570	15.040	14.520	14.125	13.745
AFTER	23	WORK	WEEKS	17.240	16.785	16.020	15.475	14.855	14.360	13.865	13.490	13.125
		MINIMUM		16.420	15.985	15.260	14.735	14.150	13.665	13.205	12.840	12.500
PAY		LEVEL		-	N	ო	4	Ŋ	9	7	80	<u>თ</u>
	AFTER AFTER AFTER AFTER AFTER	AFTER AFTER AFTER AFTER AFTER 52 104 156 208 260	AFTER	MINIMUM WORK WORK WEEKS WEEKS WEEKS	AFTER AFTE	AFTER AFTE	AFTER AFTE	AFTER AFTE	AFTER AFTER AFTER AFTER AFTER AFTER AFTER AFTER S28 260	AFTER AFTE	AFTER AFTER AFTER AFTER AFTER AFTER AFTER S28 260 26	AFTER AFTE

NON-CIPP RATES

PAY		AFTER 52	AFTER 104	AFTER 156	AFTER 208	AFTER 260	AFTER 312
LEVEL	MINIMUM	WORK WEEKS	WORK WEEKS	WORK WEEKS	WORK WEEKS	WORK WEEKS	WORK WEEKS
1	18.225	19.140	20.045	20.960	21.870	22.780	23.690
2	17.745	18.630	19.520	20.405	21.295	22.180	23.065
3	16.935	17.775	18.625	19.475	20.315	21.175	22.010
4	16.155	16.965	17.765	18.580	19.385	20.190	20.995
5	15.695	16.490	17.265	18.050	18.840	19.625	20.405
6	15.160	15.920	16.670	17.435	18.200	18.950	19.715
.7	14.595	15.325	16.055	16.785	17.520	18.250	18.970
8	14.065	14.775	15.475	16.180	16.880	17.585	18.280
9	13.615	14.295	14.985	15.660	16.345	17.030	17.710

Pg 112

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONALAUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERAL WORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: AUTOMATED PAINT TECHNICIANS AND CIPP

During the course of negotiations the parties discussed the difficulties of installing CIPP for the Automated Paint Technicians in the new paint system. The Company will continue to explore opportunities to install CIPP.

While doing so the Company agrees to pay the L-13 Automated Paint Technician in Department 074 a special rate of CIPP rate of Pay level 6 X 115%. If the Company is successful in implementing CIPP for the paint system the Automated Paint Technicians (L-13) will revert to CIPP rate Pay level 6.

DATED AT WELLAND, ONTARIO THIS 1st DAY OF

FOR UNIC	FOR THE COMPANY	
John Gagnon	Scott Anger	James Rottinghaus
Gord Digout Ven	Frank Ma ttei	
Roger Demers	Kiersten Crane	
Ed Kowalczyk	Pamela Snyder	
Thomas R. Nappe	David Harkin	
Mike Menicanin		Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERAL WÖRKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: CONTINUOUS IMPROVEMENT AWARD – OFP PAID TIME OFF

During the course of 2004 and 2007 negotiations, the Company agreed to provide interested employees in New CIPP the ability to convert their Continuous Improvement Award to OFP paid time off according to the calculations in 15.06 A(4) c under the following conditions:

All New CIPP employees working in departments or processes that anticipate a down cycle related to the Company's OFP strategies will be eligible for the paid time off. Rather than taking an immediate cash payment or banking hours until retirement, these employees will be able to convert their Continuous Improvement Award hours to OFP Paid Time Off hours. The Company will schedule OFP Paid Time Off in whole-hour increments for employees during the OFP down cycle of the department or process that the employee is assigned to. In scheduling the time off, the Company will endeavor to allow the time off at the time requested by each eligible employee. However, if this is not possible, the Company will determine the time off schedule. In general, banked CI hours are intended to provide flexibility and stability during periods of low demand due to order fulfillment. However, in periods of unusually low demand, lavoffs are still possible even

when employees may have unusedOFP PaidTime Off hours.

- If an employee would like an immediate cash payment, the Continuous Improvement Award will be paid to the employee.
- If the employee would like to defer the payment, the employee can bank their Continuous Improvement Award hours. The banked hours will either be paid as a cash payment after retirement, scheduled as paid time off immediately prior to retirement, or a combination of the two. Hours paid as a cash payment after retirement, or scheduled as paid time off immediately prior to retirement, will be paid using the employee's CIPP base rate, current at the time of retirement, for the classification in which they worked their final CIPP assignment.
- Because of unanticipated conditions (such as production schedule changes, employee transfers, or layoffs), the Company may be unable to schedule all of an employee's OFP PaidTime OFE at the times requested by the employee. In recognition of this, employees will have an opportunity to have all or part of their banked retirement hours paid out in December of each year. Requests for payout of retirement banked hours will have to be made on a form provided by payroll and submitted prior to the last pay period in November
- No employee may bank more than 1040 hours.
- In the case of death, termination, or discharge, banked hours will be paid out using the employee's CIPP base rate, current at the time of separation, for the classification in which they worked their final CIPP assignment.

DATED AT WELLAND, ONTARIO THIS 1 $^{\rm ST}$ DAY OF OCTOBER, 2007.

FOR UNIO	FOR THE COMPANY:	
John Gagnon	Scott Anger	James Rottinghaus
Gord Digout Ven	Frank Mattei	
Roger Demers	Kiersten Crane	
Ed Kowalczyk	Pamela Snyder	
Thomas R. Napp	er	David Harkin
Mike Menicanin		Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATIONAND

GENERAL WORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: CAW/DEERE CONTINUOUS IMPROVEMENT PROCESS

During the course of negotiations, both parties drew upon the knowledge gained from CIPP and agreed that a structured Continuous Improvement Process is necessary to support a culture that will allow the Company to remain competitive and provide premium wages to its employees.

To obtain maximum employee support, involvement, and rewards, a process called the CAW/Deere Continuous Improvement Process will create a structured work environment, which promotes Continuous Improvement.

Cornerstones of the Cl Process

In order to ensure the success and longevity of the CAW/Deere CI Process, the parties commit to the following cornerstones:

(1) Joint Leadership

The success of the CI Process is dependent upon the leadership and participation of both salaried and CAW represented employees. Joint leadership, CAW & Deere, is essential at all levels and provides the

necessary alignment to ensure CI is successful in improving customer satisfaction and the quality of our workplace and business.

(2) Organizational Structure

The Factory Joint CI Steering Committee leads the CI organizational structure. The JD and CI GROW Team structure supports and drives CI and separates improvement from administrative issues. This cornerstone ensures every employee will have a voice in the CI process.

(3) CI Work Plan and Events

Each CI GROW Team is responsible for developing a Work Plan. The Plan will include goals for safety, quality, delivery and efficiency and the projects and schedule to accomplish them. CI GROW Team progress and their successes will be communicated and recognized.

(4) New Profitable Work

This cornerstone provides additional work through market share increases, new products, expanded product integration, non-traditional work assignments and continuing education and training. This is the stimulus for further CI to occur.

CI Process Support Structure and Duties

The parties recognize the importance of developing and supporting a culture that will allow us to create and sustain an efficient manufacturing environment, which is safe and provides our customers with a high quality product on time. Sustaining these achievements will provide employees enhanced job security and premium wages. To this end, it is recognized that an organizational structure must be in place to support these objectives.

The CI organizational structure includes the following groups along with their roles and responsibilities.

CAW Continuous Improvement Coordinator

This person will be responsible for assisting the CI GROW and JD GROW Teams in their efforts to achieve the CI GROW team goals. The coordinator participates in preparing a report for the CI Steering Committee on the status of CI projects and the CI Process for the factory.

Natural Work Groups/CIPP Teams

CIPP teams have the primary responsibility to continuously improve.

The Natural Work Groups are based on the natural division of manufacturing areas into processes or shifts. Each Natural Work Group selects a CI representative. The Natural Work Groups are responsible for identifying opportunities for CI and assisting in the implementation of improvements in their work areas. The CI Representative carries improvement ideas from the group to the CI GROW Team.

CI Representatives will serve in this capacity for not more than 18 months.

The CI Process, through the implementation of CI projects, helps the Natural Work Group continuously improve at a competitive rate.

Each team, with the assistance of the team supervisor, is expected to develop goals focused upon safety, quality, delivery and efficiency. Actions in support of these goals will be developed and progress will be monitored.

CI GROW Teams

The CI GROW Team consists of CI Representatives from each Natural Work Group, a Supervisor, Manufacturing Engineer and additional support personnel as required.

The CI GROW Team sets goals for safety, quality, delivery and efficiency and develops a work plan and schedule to accomplish these goals. The team is responsible for communicating regularly with the Natural Work Groups.

The CI GROW Team meets twice per month. It also presents projects and goals at quarterly CI X-changes and does not deal with pay or administrative issues.

JD GROW Teams

The members of the JD GROW Team consist of Manufacturing Management, Shop Chairperson, a Shop Committeeperson, Cl Coordinators and additional support personnel as required.

The JD GROW Team meets quarterly with each of their Cl GROW Teams to review work plans, discuss and remove obstacles, and measure progress toward their Cl Goals.

The JD GROW Team is also responsible for recognizing and communicating team successes quarterly.

Factory Joint Continuous Improvement Steering Committee

A Factory Joint Continuous Improvement Steering Committee, consists of the Shop Chairperson, committeepersons, Manager of Human Resources, factory operations management and additional support personnel as

required. The Steering Committee provides guidance and support to JD & CI GROW Teams through training, communication and recognition. The committee monitors progress toward improving factory wide safety, quality, delivery and efficiency performance. In addition, the Committee will monitor team performance at the conclusion of each semester. For those teams performing below 115% the Committee will develop and implement an action plan to improve team performance. The Committee is responsible for assuring the intent of the CI Process Cornerstones is fully implemented and maintained factory wide.

The Factory Joint CI Steering Committee does not deal with pay issues.

DATED AT WELLAND, ONTARIO THIS 1ST DAY OF OCTOBER, 2007.

FOR UNIC	FOR THE COMPANY:	
John Gagnon	Scott Anger	James Rottinghaus
Gord Digout Ven	Frank Matte i	
Roger Demers	Kiersten Crane	
Ed Kowalczyk	Pamela Snyder	
Thomas R. Nappe	David Harkin	
Mike Menicanin		Colin Baker

BE7WEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERAL WORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: CHAIRPERSON COMPENSATION

During the course of negotiations, the Company agreed to provide overtime compensation to the Full-time Chairperson as follows:

- A. Overtime compensation will be based on a determination of the average number of overtime hours worked each workweek by each bargaining unit employee. This average will be determined by dividing the total number of bargaining unit overtime hours worked for each workweek by the total number of bargaining unit employees who worked and received pay in the workweek.
- B. The Chairperson will be paid time and one-half for the number of hours determined in Paragraph A above. Pay will be computed to the nearest one-tenth (.10) of an hour. Lump Sum Payments, as provided in the Current Agreement, will be applied to each overtime hour calculated under Paragraph A above.
- C. No payment will be paid under this interpretation to the Chairpersonfor any absence as a result of vacation, inplant injury for which he received a benefit under

- Worker's Compensation, or non-occupationalillness or injury for which he receives a benefit.
- D. This payment will not be made to any Alternate Chairperson.

DATED AT WELLAND, ONTARIO THIS 1ST DAY OF OCTOBER, 2007.

FOR T UNIO	FOR THE COMPANY:	
John Gagnon	Scott Anger	James Rottinghaus
Gord Digout Vena	Frank Matte i	
Roger Demers	Kiersten Crane	
Ed Kowalczyk	Pamela Snyder	
Thomas R. Nappe	David Hark in	
Mike Menicanin		Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERALWORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: CIPP ORDER FULFILLMENT/DEMAND FLOW SPECIAL PAYMENT PROCEDURES

During the course of negotiations, the parties agreed to the following special payment procedures for employees assigned to CIPP Plans operating in Order Fulfillment/Demand Flow Principles (eg. build and ship to customer demand).

A.

CIPP Input Hours	Plan Performance

Training (except breakin and cross training)

Continuous Improvement Projects such as:
R.C.I.
Process Mapping

Non CIPP Work

CIPP Rate x 115%x SHP

4 Week Average Earnings

4 Week Average Earnings

C. CIPP Plan is in operation and Employee is <u>not scheduled</u> to be in the CIPP Plan because of lack of schedule and has been removed from the Plan for one of the following reasons:

Training (except breakin and cross training)	CIPP Rate x 115% x SHP
Continuous Improvement Projects such as: R.C.I. Process Mapping	CIPP Rate x 115% x SHP
Non CIPP Work	Non CIPP rate of Temporary Assignment or CIPP Rate of Regular Assignment, whichever is higher

D. CIPP Plan is not in operation because of lack of schedule and the Employee has been assigned to one **of** the following activities:

Training (except breakin and cross training)	CIPP Rate x 115% x SHP
Continuous Improvement Projects such as: R.C.I. Process Mapping	CIPP Rate x 115% x SHP
Non CIPP Work	Non CIPP Rate of Temporary Assignment or CIPP Rate of Regular Assignment, whichever is higher

DATED AT WELLAND, ONTARIO THIS 1^{ST} DAY OF OCTOBER, 2007.

FOR T	FOR THE COMPANY:	
John Gagnon	James Rottinghaus	
Gord Digout Vena	Frank Mattei	
Roger Demers	Kiersten Crane	
Ed Kowalczyk	Pamela Snyder	
Thomas R. Nappe	David Harkin	
Mike Menicanin		Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERALWORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: CIPP PLANNED ACTIVITIES AND NEW PRODUCT INTRODUCTION

1. Planned Activities

In general, the time employees spend on determined, planned activities (excluding those identified in Article XV, Sec 15.05.A.(6),) that are of significantly greater magnitude than those incurred or allowed for in the plan's Base productivity ratio, should be considered as non-plan hours and paid at the following appropriate Contractual rate.

- A. If the plan is in operation and the employee(s)would otherwise be scheduled to work, the time should be paid in accordance with Article XV, section 15.06(2)
- B. If the plan is not in operation, employees should be paid the non-CIPP rate of the assignment or the CIPP rate of their regular assignment, whichever is higher for this non-plan time.

If however, it would be beneficial to have the employees minimize the duration and impact of the activity, a predetermined number of output hours may be added ("seeded") to the plans' reported output hours and the time considered as plan input hours. The "seeded" hours will be calculated to provide 115% plan performance for new plans (no semester-to-date plan performance

history available). Current plans will be "seeded" to semester-to-date performance, but in no case will such seeding be less than 115%.

Example A

An activity is scheduled that is anticipated to reduce weekly output to 80 units. The team will return to normal operations at the conclusion of the event.

Normal Week Data

Weekly Schedule = 100 units

Output Standard = 400 Hrs / 100 pcs.

Normal Input hours = 440 hours Current Base productivity metric = .800 Semester to Date Plan Performance = 125.51%

Step 1. Determine input hours affected by activity

Input Hours of lost production = (440 - 351.99) = 88.01

80 units x 400 Hrs / 100 = 320 Output Hours (((((320 Output Hrs/Input Hrs) - .800)/.800)x.67)+1)x115% = 125.51% Solve for Input Hours = 351.99 hours

Step 2. Determine output hours to "seed" at Base for Input Hours of lost Output

(((((Output Hrs./88.01 Input Hrs.)-.800)/.800)x.67)+1) x115%= 125.51% Solve for Output Hours = 80.00 "seed" hours

DisruptedWeek Output Hours = 320.00 (80 units x 400 Hrs/100 units) +80.00 "seed" hours = 400.00 Output Hours

Resultant Weekly Results = 400.00 / 440 = .909 productivity ratio
Weekly Plan Performance = 125.51%

2. Major New Product/Process Introduction

During the production start-up, employees will be producing product at a reduced rate while learning their jobs, trying out tooling, etc. To encourage employees to achieve the specific production requirements during the start-up, employees' time will be considered as plan Input Hours and Output Hours will be "seeded" to the plan to provide plan participants with the opportunity to achieve and exceed the Weekly Pay Level of 115%. The maximum input hours allowed to produce a given schedule during this start-up period must be determined ahead of time. Since Input Hours may vary during startup weeks, "seed" hours will be calculated after actual Input Hours worked during the week are determined. If actual Input Hours exceed the pre-determined maximum Input Hours, "seed" hours will be based on the predetermined maximum Input Hours. The "seeded hours will be calculated to provide 115% plan performance for new plans (no semester-to-date plan performance history available). Current plans will be "seeded to semester-to-date performance, but in no case will such seeding be less than 115%. No output "seed hours will be allowed for units produced in excess of the scheduled production.

New Product/Process Start-up Examples

A new model is being introduced in an assembly operation. Management has planned for a reduced schedule during the first week of this introduction. They have provided, however, enough materials to exceed the planned schedule if employees demonstrate the ability to learn the new work and straighten out tooling and other problems ahead of schedule.

Given conditions:

Current Base = .893
New model output standard = 5000 Hours/100pcs
"Full" production = 5 units/day = 25 units/week
Start-up period = 5 days
Start-up production schedule = 10 units/week
Pre-determined maximum Input Hours = 1400 hours

Example 1

Units produced=10 Input Hours = 1400 Semester to Date Plan Performance= 123.58%

Expected Output Hours = (5000 Hours/100 x 10) = 500 (((((OutputHrs./1400 InputHrs.)-.893)/.893)x.67)+1)x115%=123.58% Solve for Output Hours = 1389.42 Required "seed hours = 1389.42 - 500 = 889.42 "seed hours Allowed "seed" hours/unit produced = 889.42 / 10 = 88.95 Total Output Hours = (5000 Hrs/100 x 10) + (88.95 total Output Hours = 123.58% Weekly Results = 1389.5 / 1400 = .993 Plan Performance = 123.58%

Example2

Units produced = 9 units Input Hours = 1300 Semester to Date Plan Performance = 123.58%

Expected Output Hours = (5000 Hours/100 x 10) = 500 (((((Output Hrs./1300 Input Hrs.).893)/.893)x.67)+1)x115%=123.58%
Solve for Output Hours = 1290.17
Required "seed" hours = 1290.17 - 500 = 790.17 "seed" hours
Allowed "seed hours/unit produced = 790.17 / 10 = 79.017

Total Output Hours = $(5000 \text{ Hrs/}100 \times 9) + (79.017 \times 9) = 1161.15$ Weekly Results = 1161.15 / 1300 = .893Plan Performance = 115.02%

Example 3

Units produced = 10 units Input Hours = 1300 Semester to Date Plan Performance = 123.58%

Expected Output Hours = (5000 Hours/100 x 10) = 500 (((((Output Hrs./1300 Input Hrs.)-.893)/.893)x.67)+1)x115%=123.58% Solve for Output Hours = 1290.17 Required "seed hours = 1290.17 - 500 = 790.17 "seed" hours
Allowed "seed hours/unit produced = 790.17 / 10 = 79.017

Total Output Hours = $(5000 \,\text{Hrs}/100 \,\text{x}\,10) + (79.017 \,\text{x}\,10) = 1290.17$

Weekly Results = 1290.17 / 1300 = .9924 Plan Performance = 123.58%

Example 4

Units produced = 9 units Input Hours = 1450 Semester to Date Plan Performance = 123,58%

Expected Output Hours = (5000 Hours/100 x 10) = 500 (((((Output Hrs./1400*Input Hrs.)-.893)/.893)x.67)+1)x115%=123.58% Solve for Output Hours = 1389.42 Required "seed hours = 1389.42 - 500 = 889.42 "seed" hours

Allowed "seed" hours/unit produced = 889.42 / 10 = 88.942

Total Output Hours = (5000 Hrs/ 100 x 9) + (88.942 x 9)= 1250.48

Weekly results = 1250.48 / 1450 = .862 Plan Performance = 112.36% *Actual Input Hours exceeded maximum allowable Input Hours

Example 5

Units produced = 10 units Input Hours = 1500 Semester to Date Plan Performance = 123.58%

Expected Output Hours = (5000 Hours/100 x 10) = 500 ((((Output Hrs./1400*Input Hrs.)-.893)/.893)x.67)+1) x115%=123.58%

Solve for Output Hours = 1389.42

Required "seed hours = 1389.42 - 500 = 889.42 "seed" hours

Allowed "seed hours/unit produced = 889.42 / 10 = 88.942

Total Output Hours = $(5000 \text{ Hrs/} 100 \times 10) + (88.942 \times 10) = 1389.42$

Weekly results = 1389.42 / 1500 = .926

Plan Performance = 117.87%

Example 6

Units produced = 11 units
Input Hours = 1300
Semester to Date Plan Performance = 123.58%

Expected Output Hours = (5000 Hours/100 x 10) = 500 (((((OutputHrs./1300 InputHrs.)-.893)/.893)x.67)+1) x115%=123.58%

Solve for Output Hours = 1290.17

Required "seed" hours = 1290.17 - 500 = 790.17 "seed" hours

Allowed "seed" hours/unit produced = 790.17 / 10 = 79.017

Total Output Hours = $(5000 \text{ Hrs/} 100 \text{ x } 11) + (79.017 \text{ x } 10^*) = 1340.17$

Weekly results = 1340.17 / 1300 = 1.031

Plan Performance = 126.9%

^{*}Actual Input Hours exceeded maximum allowable Input Hours

*Seed hours per unit will not be applied to any units produced above the pre-determined schedule.

DATED AT WELLAND, ONTARIO THIS 1ST DAY OF OCTOBER, 2007.

FOR THE UNION:		FOR THE COMPANY:
ohn Gagnon	Scott Anger	James Rottinghaus
3ord Digout Venanzio D'Addario		Frank Matte i
Roger Demers	John Wilgosh	Kiersten Crane
Id Kowalczyk		Pamela Snyder
homas R. Napper		David Harkin
/like Menicanin		Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERALWORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: CIPP STEERING COMMITTEE

During the Agreement period ending 30 September 1995, the Company and the Union worked closely together to "pilot" new forms of compensation which encouraged employees to utilize their knowledge, training, experience and talent in initiating changes and managing their workplace on a daily basis to achieve continuous improvement.

During the course of the 1995 and 1998 negotiations, both parties drew upon the knowledge gained from these pilot programs and agreed that they support a new culture that will allow the Company to remain competitive and provide premium wages to its employees. To aid in this transition and obtain maximum employee support and involvement in this process, a Joint CIPP Steering Committee was established. The Committee shall consist of Company representatives and Union Committeepersons representing employees assigned to CIPP Plans. During 2001-2004 contract period, the Company and Union worked on a pilot CIPP Revitalization Program. In 2004 negotiations the parties introduced New CIPP, which is an enhanced version of CIPP Revitalization. It was decided that the CIPP Steering Committee would continue on an "as needed" basis. The Company and Union will appoint their representatives to the Committee, however, one (1) of the Union representatives will be the Chairperson of the Shop Committee, and one (1) of the Company representatives will be the Manager of Human Resources. The role of this committee is as follows:

Factory Joint CIPP Steering Committee

FOR THE UNION:		FOR THE COMPANY:
lohn Gagnon	Scott Anger	James Rottinghaus
3ord Digout Venanzio D'Addario		Frank Matte i
Roger Demers	John Wilgosh	Kiersten Crane
Ed Kowalczyk		Pamela Snyder
ſhomas R. Napper		David H arkin
∕like Menicanin		Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERALWORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: CONTINUITY OF PAY

During the course of negotiations, the parties discussed the difficulties in continuity of pay for union officials when they are on leave of absence for 30 days or less for union business.

- (a) The Company will pay the union officials during such short term leaves of absence as though they were working in the factory.
- (b) The Company will be reimbursed for the wages and related benefit costs of employees on union leave of absence. The Company will provide a monthly invoice to CAW Local 275 and payment is due within 30 days. The related benefit costs include:
 - vacation pay
 - Christmas shutdown pay
 - shift premiums
 - productivity bonus
 - Company portion of CPP and El deductions

- yearly profit sharing bonus (if any)
- Union Paid Education Fund
- Social Justice Fund
- Lump Sum

DATED AT WELLAND, ONTARIO THIS 1ST DAY OF OCTOBER, 2007.

FOR THE UNION:		FOR THE COMPANY
John Gagnon	Scott Anger	James Rottinghaus
Gord Digout Venanzio D'Addario		Frank Mattei
Roger Demers	John Wilgosh	Kiersten Crane
Ed Kowalczyk		Pamela Snyder
Thomas R. Nappe	r	David Harkin
Mike Menicanin		Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERAL WORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: EMPLOYMENT INSURANCE PREMIUM REDUCTION

During the course of negotiations the patties discussed the sharing of the Employment Insurance premium reduction allowed employers with qualified wage loss replacement plans.

The parties recognize that the Employment Insurance premium reduction may be passed on to employees as a group either in the form **of** a cash rebate or in the form of employee benefits.

It was agreed that effective with the first pay period following ratification, the employees share of the rebate will revert to the Company in exchange for the improvements in current benefits already negotiated.

DATED AT WELLAND. ONTARIO THIS 1ST DAY OF OCTOBER, 2007.

FOR THE UNION:		FOR THE COMPANY
John Gagnon	Scott Anger	James Rottinghaus
Gord Digout Venanzio D'Addario		Frank Matte i
Roger Demers	John Wilgosh	Kiersten Crane
Ed Kowalczyk		Pamela Snyder
Thomas R. Napper		David Harkin
Mike Menicanin		Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERAL WORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: EMPLOYEE ASSISTANCE AND HARASSMENT

During the course of negotiations, the parties discussed:

Services for John Deere employees needing alcohol & substance abuse treatment; and

Conflict between John Deere employees based on allegations of harassment.

"Harassment" means engaging in a course of vexatious comment or conduct that is known or ought reasonably to be known to be unwelcome or offensive and is based on a prohibited ground of discrimination.

Every individual has the right to equal treatment and freedom from harassment in employment without discrimination because of race, ancestry, place of origin, colour, ethnic origin, citizenship, creed, sex, sexual orientation, age, record of offences, martial status, samesex partnership status, family status or disability.

In an effort to effectively deal with these difficult situations, the parties have agreed that the Chairperson and the Manager, Human Resources will jointly monitor these issues and address them in an expeditious manner as they arise.

DATED AT WELLAND, ONTARIO THIS 1ST DAY OF OCTOBER, 2007.

FOR THE UNION:		FOR THE COMPANY:
John Gagnon	Scott Anger	James Rottinghaus
Gord Digout Venanzio D'Addario		Frank Mattei
Roger Demers	John Wilgosh	Kiersten Crane
Ed Kowalczyk		Pamela Snyder
Thomas R. Napper		DavidHarkin
Mike Menicanin		Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERALWORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: HEALTH PLAN CHANGES

During the course of 1998 and 2004 negotiations, a number of changes were made to the Health Benefit Plan language with the intent of standardizing benefits and clarifying the application and administration of benefits. The parties agree that unless expressly provided such changes are not intended to result in the elimination of coverage currently provided. The specific benefit plan changes provide enhancements to the current coverage as well as revisions to certain levels of coverage. If such changes would result in the unintentional elimination of coverage previously provided, the Manager of Human Resources agrees to meet with the Chairperson to discuss the resolution of the issue

DATED AT WELLAND, ONTARIO THIS 1ST DAY OF OCTOBER, 2007.

FOR THE UNION:		FOR THE COMPANY
John Gagnon	Scott Anger	James Rottinghaus
Gord Digout Venanzio D'Addario		Frank Matte i
Roger Demers	John Wil <mark>gosh</mark>	Kiersten Crane
Ed Kowalczyk		Pamela Snyder
Thomas R. Napper		David H ark in
Mike Menicanin		Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATIONAND

GENERAL WORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: HOURS OF WORK

During the course of negotiations the parties discussed the opportunities afforded both the employees and the Company with the introduction of alternate work hours in various areas of the factory. The opportunities recognized are better utilization of assets, improved maintenance and increased flexibility, within the Deere Production System principles. It also affords employees longer periods of time off.

- The first was a discussion of an alternate shift schedule for the maintenance area.
 An alternate rotating two week shift schedule with 10 days on and 4 days off. The intent is to perform more maintenance work in house.
- The second was a 4 day 10 hour work schedule with 3 days off for selected production areas in manufacturing.

The parties agree to meet and discuss the implementation of alternate work hours. The implementation of any such proposal will be by mutual agreement. Resolution may involve changes in contract language, the establishment of pilot programs in limited areas or other innovative changes in conjunction with the Deere Production System.

DATED AT WELLAND, ONTARIO THIS $1^{\rm ST}$ DAY OF OCTOBER, 2007.

FOR THE UNION:		FOR THE COMPANY
John Gagnon	Scott Anger	James Rottinghaus
Gord Digout Venanzio D'Addario		Frank Mattei
Roger Demers John Wilgosh		Kiersten Crane
Ed Kowalczyk		Pamela Snyder
Thomas R. Napper		David Harkin
Mike Menicanin		Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONALAUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERAL WORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: LIGHT DUTY COMPENSATION FOR EMPLOYEES DISABLED AS A RESULT OF AN OCCUPATIONAL INJURY OR ILLNESS

During the course of negotiations, the parties discussed the compensation for employees who are partially disabled from an occupational injury or illness, but capable **of** performing light duty work. The rates defined below will be paid for the hours the employee is assigned to perform light duty work in the factory.

Employees will be paid an hourly rate equal to their Average Earnings, as defined in Article 15.10, or the "Annual Wage Ceiling" as defined by WSIB, whichever is less, but in no case will this rate be less than the rate of their regular assignment.

Therapy:

Employees may be allowed paid time off to attend to a therapy appointment provided the therapy is for an establishedwork related injury.

In order for an employee to be permitted to attend appointments during scheduled working hours, the appointment must be pre-approved by the John Deere Claims Management Coordinator. Employees may be paid by John Deere Welland Works for claims which are attended during scheduled working hours.

The appointment will be booked for the entire course of therapy. Employees receiving therapy will be required to attend appointments on a consistent basis regardless of their shift schedule.

Therapy for non-work related injuries was never intended to be covered under this program. Requests for payment of wages to attend non-work related therapy will be denied.

FOR THE UNION:	FOR THE COMPANY:
John Gagnon Scott Anger	James Rottinghaus
Gord Digout Venanzio D'Addario	Frank Mattei
Roger Demers John Wilgosh	Kiersten Crane
Ed Kowalczyk	Pamela Snyder
Thomas R. Napper	David Harkin
Mike Menica nin	Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERAL WORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: LUMP SUM PAYMENT

During the course of negotiations the Company agreed to make Lump Sum Payments to eligible employees. Employees actively at work on the eligibility date will receive a Lump Sum Payment in each of the years of the Agreement in accordance with the following tables:

Eligibility Date	Amount	Payable
30 September 2007	Actual hours worked per individual employee for the period 25 September 2006 - 23 September 2007 times \$0.75 per number of hours worked.	25 October 2007
30 September 2008	Actual hours worked per individual employee for the period 24 September 2007 - 28 September 2008 times \$0.75 per number of hours worked.	30 October 2008

Eligibility Date	Amount	Payable
30	Actual hours worked per	29
September 2009	individual employee for 'the period 29 September 2008 - 27 September 2009 times \$0.75 per number of hours worked.	

An employee shall become eligible for a Lump Sum Payment if he/she is actively at work on the eligibility date above. An employee who retires or dies during the term of this Agreement who, but for such retirement or death, would have had seniority as of the designated eligibility date, shall qualify for the lump sum.

DATED AT WELLAND, ONTARIO THIS 1st DAY OF OCTOBER.2007.

FOR THE UNION:		FOR THE COMPANY:
John Gagnon	Scott Anger	James Rottinghaus
Gord Digout Venanzio D'Addario		Frank Mattei
Roger Demers	John Wil <mark>gosh</mark>	Kiersten Crane
Ed Kowalczyk		Pamela Snyder
Thomas R. Napper		David Ha rkin
Mike Menicani n		Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERAL WORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

FOR THE UNION:		FOR THE COMPANY:
John Gagnon	Scott Anger	James Rottinghaus
Gord Digout Venanzio D'Addario		Frank Mattei
Roger Demers	John Wilgosh	Kiersten Crane
Ed Kowalczyk		Pamela Snyder
Thomas R. Napper		David Harkin
Mike Menicanin		Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERALWORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

FOR THE UNION:	FOR THE COMPANY
John Gagnon Scott Anger	James Rottinghaus
Gord Digout Venanzio D'Addario	Frank Mattei
Roger Demers John Wilgosh	Kiersten Crane
Ed Kowalczyk	Pamela Snyder
Thomas R. Napper	David H arkin
Mike Menicanin	Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERAL WORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: PAID EDUCATION LEAVE

During the course of negotiations, the Company agreed to contribute \$0.02 (two cents) per actual hour worked, to the Union Paid Education Fund, effective with the signing of this Collective Agreement.

The Company agrees to forward a cheque quarterly, payable to the CAW Leadership Fund, to:

CAW- Leadership Fund 205 Placer Court Toronto, Ont., M2H 3H9

Employees selected by the Union to attend courses will be granted a leave of absence without pay or loss of seniority for twenty (20)days time over a twelve (12) month period from the first day of leave.

FOR M E UNION:		FOR THE COMPANY
John Gagnon	Scott Anger	James Rottinghaus
Gord Digout Venanzio D'Addario		Frank Matte i
Roger Demers	John Wilgosh	Kiersten Crane
Ed Kowalczyk		Pamela Snyder
Thomas R. Napper		David Ha rkin
Mike Menicanin		Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERAL WORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: PRESCRIPTION DRUG MAIL ORDER

During the course of negotiations, the parties discussed the establishment of an optional Mail Order Prescription Maintenance Drug Arrangement for wage employees. During the contract period the Company will investigate the feasibility of such a plan. The Chairperson will be informed of the progress of the investigation.

FOR THE UNION:	FOR THE COMPANY
John Gagnon Statt Anger	James Rottinghaus
Gord Digout Venanzio D'Addario	Frank Mattei
Roger Demers John Wilgosh	Kiersten Crane
Ed Kowalczyk	Pamela Snyder
Thomas R. Napper	David Harkin
Mike Menicanin	Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERAL WORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: RETURN FROM TOTAL DISABILITY AFTER VACATION SHUTDOWN

Employees who are totally disabled during the Vacation Shutdown period will, upon their return to work, whether it is regular or modified work, have one week to reschedule, with their supervisor, the vacation they otherwise would have been expected to take during the Vacation Shutdown. During this week, the employee will have an opportunity to reschedule this vacation in one or two week increments, not in single days. Scheduling will be at the discretion of management, with due consideration being given to the employee's wishes. If no agreement is made during the first week, the supervisor will reschedule the vacation at his/her discretion.

FOR THE UNION:		FOR THE COMPANY
John Gagnon	Scott Anger	James Rottinghaus
Gord Digout Venanzio D'Addario		Frank Mattei
Roger Demers	John Wilgosh	Kiersten Crane
Ed Kowalczyk		Pamela Snyder
Thomas R. Napper		David Harkin
Mike Menica nin		Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERAL WORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: SKILLED TRADES AND CIPP

During the course of negotiations, the parties discussed the inclusion of employees in Pay Level 1 though 3 in CIPP applications. The parties agree that issues involving the inclusion of these employees in a CIPP application such as: changes in overtime groups and the selection of employees to be included in the plan will be resolved by mutual agreement of the parties.

FOR THE UNION:	FOR THE COMPANY:
John Gagnon Scott Anger	James Rottinghaus
Gord Digout Venanzio D'Addario	Frank Mattei
Roger Demers John Wilgosh	Kiersten Crane
Ed Kowalczyk	Pamela Snyder
Thomas R. Napper	David Harkin
Mike Me nicanin	Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERALWORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: SOCIAL JUSTICE FUND

During the course of negotiations, the Company agreed to contribute \$0.01(one cent) per actual hour worked to the Social Justice Fund effective with the signing of this Collective Agreement.

The Company agrees to forward the contributions quarterly to:

CAW-Social Justice Fund 205 Placer Court Toronto, Ont., M2H 3H9

Information about the contribution, including the Local Number, the Bargaining Unit covered, the number **of** actual hours used in the calculation and the time period covered **will** be included. This information will be mailed to the following addresses at the same time the contributions are made:

CAW-Social Justice Fund 205 Placer Court Toronto, Ont. M2H 3H9 CAW Local 275 16 Steel St. Welland, Ont. L3B 3L9

Attn: Financial Secretary

FOR THE UNION:	FOR THE COMPANY:
John Gagnon Scott Anger	James Rottinghaus
Gord Digout Venanzio D'Addario	Frank Mattei
Roger Demers John Wilgosh	Kiersten Crane
Ed Kowalczyk	Pamela Snyder
Thomas R. Napper	David Ha rkin
Mike Menicanin	Colin Baker

FOR THE UNION:		FOR THE COMPANY
John Gagnon	Scott Anger	James Rottinghaus
Gord Digout Venan	zio D'Addario	Frank Matte
Roger Demers	John Wilgosh	Kiersten Crane
Ed Kowalczyk		Pamela Snyde
Thomas R. Napper		David H a rkir
Mike Menicanin		Colin Bake

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATIONAND

GENERAL WORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: SPECIAL PAY PROVISIONS FOR PREVENTATIVE MAINTENANCE

During the course of negotiations the parties discussed the opportunities a formal preventative maintenance program affords both the employees and the company. It is understood that a formal preventative maintenance program has little opportunity to be improved upon because of its prescriptive nature.

A formal Preventative Maintenance Program

- Is a regularly scheduled, defined and documented process
- Is work that must be performed exactly as prescribed (has no opportunity for improvement in the process)
- Is not part of normal housekeeping, clean up, or 5S
- Is not normal operation or machine checks that are part of daily or weekly operations to ensure accuracy of machines or quality of parts.

Due to these restrictive conditions within a formal preventative maintenance program it is understood that time spent by CIPP team members on formal preventative maintenancewill be paid at CIPP Rate X 115% as outlined in Article 15.06 (1).

The Operations Manager or his/her designate must approve all preventative maintenance programs covered by **this** letter.

FOR THE UNION:		FOR THE COMPANY	
John Gagnon	Scott Anger	James Rottinghaus	
Gord Digout Venanzio D'Addario		Frank Mattei	
Roger Demers John Wilgosh		Kiersten Crane	
Ed Kowalczyk		Pamela Snyder	
Thomas R. Napper		David Harkin	
Mike Menicanin		Colin Baker	

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERALWORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: SUMMER SHUTDOWN

This letter will confirm our understanding on scheduling employees to work during the inventory and vacation shutdown. The Company will normally schedule junior employees who do not have sufficient vacation entitlement for the whole vacation shutdown. If additional employees are required, the Company will assign work to the senior employees possessing the skills and qualifications, who have signed the shutdown work **list.**

FOR THE UNION:	FOR THE COMPANY:
John Gagnon Scott Anger	James Rottinghaus
Gord Digout Venanzio D'Addario	Frank Mattei
Roger Demers John Wilgosh	Kiersten Crane
Ed Kowalczyk	Pamela Snyder
Thomas R. Napper	David Harkin
Mike Me nicanin	Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERALWORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: SUMMER STUDENT PROGRAM

During the course of the negotiations, the parties discussed applying the Summer Student program to support flexibility in Order Fulfillment. The Summer Student program will cover the four-month period end of April through August, which aligns with the typical summer break period for post secondary schools. Students hired under the Summer Student program will not be subject to the probationary period described in Article 12.01.

In the event that the "peak" production period dictated by Order Fulfillment does not align with the Summer Student program period, the Company will meet with the Union to discuss hiring Temporary Employees. The Company will provide the number of Temporary Employees required, as well as the timeframe for which those Temporary Employees are needed. If the required timeframe exceeds three months, the Company and the Union agree to extend the probationary period for up to an additional three months for these Temporary Employees. The number of Temporary Employees on extension will not exceed 100. At no time will the Company hire Temporary Employees when there are seniority employees, with the required skills, on lay off.

DATED AT WELLAND, ONTARIO THIS 1st DAY OF OCTOBER, 2007.

FOR THE UNION:	FOR THE COMPANY:
John Gagnon Scott Ange	r James Rottinghaus
Gord Digout Venanzio D'Addari	o Frank Mattei
Roger Demers John Wilgos	h Kiersten Crane
Ed Kowalczyk	Pamela Snyder
Thomas R. Napper	David Harkin
Mike Menicanin	Colin Baker

BETWEEN: JOHN DEERE WELLAND WORKS of

JOHN DEERE LIMITED,

Welland, Ontario

AND - THE NATIONAL AUTOMOBILE,

AEROSPACE, TRANSPORTATION AND

GENERAL WORKERS UNION OF CANADA (CAW-CANADA)

and its Local 275

RE: TRAINING

The parties recognize the need to jointly promote training, retraining and personal development to meet the technological advances in the manufacturing environment and improve job skills for CAW represented employees. To this end, the parties agree to establish a team comprised of company and union representatives who will work through the John Deere Welland Works training process to identify, prioritize and participate in the delivery of training programs.

The objectives will be:

- provide opportunities for employees to upgrade their job skills
- improve opportunities for job placement and job retention for employees relative to their seniority
- contribute toward a goal of retaining senior employees at work
- contribute to the competitiveness and well being of the employees and the Company
- to jointly obtain funding for training purposes from various governmental agencies.

FOR THE UNION:		FOR THE COMPANY:
John Gagnon Sc	ott Anger	James Rottinghaus
Gord Digout Venanzio D'Addario		Frank Mattei
Roger Demers John Wilgosh		Kiersten Crane
Ed Kowalczyk		Pamela Snyder
Thomas R. Napper		David Harkin
Mike Menicanin		Colin Baker

	JOINT COMPANY-UNION DARD	175
	Members.	
Section 1.		
Section 2.	Secretary	
Section 3.	Authority and Jurisdiction	
Section 4.	Procedure	179
Section 5.	Chairperson of the Board Selection of	180
Section 6.	Chairperson of the Board-Duties.	180
Section 7.	Request for Review	181
Section 8.	Reports to the Board	183
ARTICLE II	UNION OFFICIALS	184
Section 1.	Service Credit	184
Section 2.	Pension Benefits	184
Section 3.	Life Insurance	185
ARTICLE III	PART-TIME EMPLOYEES	185
Section 1.	Part-Time Employees	185
ARTICLE I	ESTABLISHMENT OF PLAN	186
Section 1.	Purpose	186
Section 2.	Object	186
Section 3.	Cost of Benefits	
Section 4.	Effective Dates	186
Section 5.	Statutory Benefit Defined	187
Section 6.	The Trust Fund and the Trustee	187
Section 7.	Payments to the Trust Fund	188
Section 8.	Administration and Termination	188
Section 9.	Nonencumbrance of Benefits	189
Section 10.	Service Credit	189
Section 11	Vesting	189

Section 12.	Nonparticipation	189
Section 13.	Denial of Benefits	190
Section 14	Administration	190
ARTICLE II	PLAN FOR COMPUTING AND	
MAINTAININ	G EMPLOYEES SERVICE CRE	
•••••		190
Section 1.	Company / Employee Defined	190
Section 2.	Plan Participation and Starting Date	.191
Section 3.	Service Credit Records	191
Section 4.	Service Credit Defined	191
Section 5.	Time of Crediting	192
Section 6.	Partial Year's Service Credit	192
Section 7.	Occupational Injury	193
Section 8.	Nonoccupational Injury or Illness	193
Section 9.	Layoff	194
Section 10.	Military Service	194
Section 11.	Accumulation in Any Unit	194
Section 12.	Transfer	194
Section 13.	Crediting Time Spent at Other Unit	s 195
Section 14.	Termination of Participation,	
	Employment and Service Credit	
Section 15.	Re-establishment of Service Credit	
Section 16.	Original Unit Defined	
ARTICLE III	ELIGIBILITY FOR RETIREMEN	
AND AMOU	NT OF PENSIONS	. 197
Section 1.	Normal Retirement - Age 65	
Section 2.	Early Retirement	199
Section 3.	Total and Permanent Disability	
Retirement		
Section 4	Automatic Retirement	204

	Section 5.	Deferred Vested Pension	.205
		Commencement and Termination of the fits	
Δ	RTICLE IV		
	Section 1.	Survivor Benefitsfor Retired Emplo	
	Section 2.	Survivor Benefitsfor Employees	
	Section 3. Survivor Ber	Commencement and Termination of the comme	
	Section 4.	Death Benefits for Employees and ployees	.210
Δ	RTICLE V	SUPPLEMENTAL ALLOWANCE	
	Section 1.	Eligibility for Supplemental Allowand	ce211
	Section 2.	Amount of Supplemental Allowance	.212
	Section 3. Computing A	Assumptions and Adjustments in Amount of Supplemental Allowance	.214
	Section 4.	Payment of Allowance	.214
	Section 5.	Discharged Employees	
Æ	ARTICLE VI	RETIRED EMPLOYEES	
	Section 1.	Minimum Benefits	216
	Section 2.	Formula Pensions	218
	Section 3.	Disability Pensioners	218
	Section 4. Death of Re	Restoration of Pension Benefits After tired Employee's Spouse	
	Section 5. the Event of	Election of Pension Benefit Increase Divorce	
	Section 6. Retirement	Survivor Benefit Election After	221
	Section 7.	Supplemental Allowance of Retired	

ARTICLE VI UNDER THE	I MAXIMUMBENEFITS PAYAB E PLAN	
Section 1.	General	.225
Section 2.	Definitions	.225
Section 3.	Lifetime Retirement Benefits	.226
Section 4.	Bridging Benefits	.228
Section 5.	Retirement Benefits Before Age 65.	
Section 6.	Benefit Accrual Rate	
Section 7.	Service Credit	
ARTICLE I	GENERAL PROVISIONS	232
Section 1.	Preamble	.232
Section 2.	Effective Dates of Coverage	.232
Section 3.	Termination	.233
Section 4.	Cost of Benefits	.233
Section 5. Termination	Amendment. Modification.and Plan	.234
Section 6.	Funding	
Section 7.	Claim for Benefits	.234
Section 8.	Denial of Benefits	.235
Section 9.	Definitions	.235
ARTICLE II UNDER THI	CONTINUATION OF COVERAG	
Section 1.	Leave of Absence	.236
Section 2.	Layoff	237
Section 3. Leave of Al	Illness. Accident. Maternity or Parer bsence	
Section 4. Monthly Ins	Employee Receiving Life Insurance stallments	238
Section 5.	Retirement	238
Section 6	Survivina Spause	238

ARTICLE III		239
ARTICLE IV HOSPITAL E ONTARIO HI	enefit Contribution Rates MEDICAL, SURGICAL AND EXPENSE BENEFITS UNDER EALTH INSURANCE PLAN (OF	IIP)
Section I. ARTICLE V EXPENSE B	O.H.I.PSUPPLEMENTAL HOSPITAL ENEFITS EMPLOYEES AND T	240 HEIR
DEPENDEN Section 1.	TS For Daily Room and Board Expen	
Section 2.	Chronic Care	241
ARTICLE VI EXPENSE B	ADDITIONAL TREATMENT SENEFITS EMPLOYEES AND T	HEIR
Section 7.	Prosthetic Device Benefits Durable Medical Equipment Ambulance Licenced, Certified Practitioner Special Duty Nursing Health Coverage outside province anada Diagnostic Services	244 245 245 246 e of 247
	BENEFITS	
Section 1. Section 2. Section 3.	Eligibility Effective Date Covered Catastrophic Medical Ex	248 pense
		749

Section 4.	Indemnity Limits	249
Section 5.	Exclusions and Limitations	250
	EXCLUSIONS AND LIMITATION OF THE PLANTAGE OF T	_AN
Section 1.	Exclusions and Limitations	
Section 1.	Nonduplication of Benefits	
	Coordination of Benefits	
Section 3.		
Section 4.	Subrogation Administrative Benefit Guidelines	
Section 5.		
ARTICLE IX	PRESCRIPTION DRUG EXPE	N2F
	TS	255
Section 1.	Eligibility Amount of Benefit	
Section 2.	Exclusions and Limitations	
Section 3.		
ARTICLE X	DENTAL EXPENSE BENEFITS S AND THEIR DEPENDENTS	
Section 1.	Eligibility	
Section 2.	Effective Date	
Section 3.	Hospital Expenses	
Section 4.	Indemnity Limit	
Section 5.	Covered Dental Expenses	
Section 6.	Treatment Plan	
Section7.	Limitations	
Section 8.	Exclusions	
Section 9.	Definitions,	264
Section 10.	Coordination With Other Dental	00=
- xnense Be	enefits	265

	VISION CARE EXPENSE MPLOYEES AND THEIR	
DEPENDEN'	TS	265
Section 1.	Eligibility	265
Section 2.	Effective Date	265
Section 3.	Covered Vision Care Benefits	266
Section 4.	Indemnity Limits	266
Section 5.	Exclusions	267
ARTICLE XII	_	
	AN FOR EMPLOYEES AND TH	
Section 1.	Eligibility	267
Section 2.	Effective Date	268
Section 3.	Covered Hearing Aid Expense Be	
Section 4.	Indemnity Limits	
Section 5.	Coordination of Benefits	269
Section 6.	Services Not Covered	269
ARTICLE I	ESTABLISHMENT OF PLAN	271
Section 1.	Preamble	271
Section 2.	Claims Covered Under New Plan	271
Section 3.	Effective Dates of Coverage	271
Section 4.	Physical Examinations	271
Section 5.	Termination	272
Section 6.	Cost of Benefits	272
Section 7. Termination	Amendment, Modification, and Pl	an 272
Section 8.	Funding	
Section 9.	Application for Benefits	272
Section 10	Denial of Renefits	273

ARTICLE II	CONTINUATION OF COVERAC	
Section 1. Section 2. ARTICLE III	Leave of Absence LayoffAMOUNT OF BENEFIT	273 273
Section 1. Section 2. ARTICLE IV DURATION	Determination of Earnings Benefit Amount COMMENCEMENT AND OF BENEFITS	275
ARTICLE V	Weekly Indemnity Employment Insurance Program Layoff Disability Benefits Long-Term Disability Benefits al and Nonoccupational EXCLUSIONS AND LIMITATION	279 279 281 NS
Section 1. Section 2. ARTICLE I	Exclusions Nonduplication of Benefits PRELIMINARY PROVISIONS	287 287
Section 1. Plan Section 2. Section 3. ARTICLE 16	Continuation and Amendment of the Term of Preliminary Provisions Governmental Rulings	289 289 290
	Eligibility for a Regular Benefit Conditions with Respect to Layoff Disputed Claims for Employment Benefits	292

Section 1	Regular Benefits	294
	mployment Insurance Benefit and Cion	
	. Insufficient Credit Units for a Full B	
	Benefit Overpayments	
	Withholding Tax	
	CREDIT UNITS AND DURATIO	
Section 1.	General	298
Section 2.	Accrual of Credit Units	298
Section 3.	Forfeiture of Credit Units	299
Section 4.	Cancellation & Credit Units	300
Section 5. A	Armed Services	301
OF ELIGIBIL	APPLICATION. DETERMINATION. DETERMINATION OF THE PROCEDURE OF THE PROCEDUR	RES
Section 1.		
	Determination of Eligibility	
	g ,	
	ADMINISTRATION OF THE PLA	
Section 1.	Powers and Authority of the Comp	
Section 2.	Determination of Dependents	
Section 3. Certain Cor	To Whom Benefits Are Payable in nditions	304
Section 4.	Nonalienation & Benefits	304
Section 5.	Applicable Law	305
ARTICLE VI	MISCELLANEOUS	
Section 1	General	305

Section 2.	Amendment and Termination of th	
ARTICLE VI	DEFINITIONS	
ARTICLE I		311
Section 1.	Automatic Short Week Benefits	311
Section 2. Benefit	Eligibility for An Automatic Short V	
Section 1.	Purpose	
Section 2.	Eligibility	
Section 3.	Approved Schools and Courses	314
Section 4.	Application for Approval	315
Section 5.	Amount of Tuition Aid	
ARTICLE I	ESTABLISHMENTOF PLAN	316
Section 1.	Preamble	316
Section 2.	Cost of Benefits	316
Section 3.	Nonencumbrance of Benefits	316
Section 4. Termination	Amendment. Modificationand Pla	
Section 5.	Funding	
Section 6.	Application for Benefits	317
Section 7.	Denial & Benefits	
ARTICLE II	ELIGIBILITY AND AMOUNT O	
Section 1.	Eligibility of Employees and ement c Insurance	
Section 2.	Group Life Insurance	
Section 3. Insurance	Accidental Death and Dismembe	rment
Section 4.	Total and Permanent Disability	

Section 5. Bridge Survi	Transition Survivor Income Benefit and ivor Income Benefit
Section 6.	Continuance of Insurance325
Section7.	Reduction of Insurance326
Section 8.	Termination of Insurance326
Section 9.	Exclusions & Limitations328
Section 10.	Designation of Beneficiary329
Section 11. Insurance	Privilege of Obtaining Individual329
Section 12.	Physical Examinations330
Section 13.	Common Law Spouse330
Section 1.	Type of Plan and Purpose332
Section 2.	Definitions332
Section 3.	Eligibility and Participation334
Section 4.	Amount of Benefit334
Section 5.	Payment of Benefit338
Section 6.	Miscellaneous339
Section 1.	Purpose342
Section 2.	Eligibility342
Section 3.	Enrollment342
Section 4.	Payroll Deductions343
Section 5.	Registered Retirement Savings Plans 343
Section 6. Plan	Voluntary Termination of Participation in344
Section 7. in Plan	Automatic Termination of Participation344
Section 8.	Amendment or Termination of Plan344

APPENDIX "1" BENEFIT PLANS

ARTICLE 1 JOINT COMPANY-UNION PENSION BOARD

Section 1. Members

- A Joint Company-Union Pension Board, Hereinafter called the Pension Board, shall be established. consisting of six members, three of whom shall be appointed by the Company and three of whom shall be appointed by the Union from among the employees in the bargaining unit represented by the Union. The Company shall compensate each Union appointee for work time lost in any regular meeting of the Pension Board, as herein provided, up to but not in excess of two hours per month, and, in the event of a special meeting requested by the Company members, the Company shall compensate each Union appointee for work time lost in such special meeting, as herein provided, up to but not in excess of two hours per month, in each case computed on the same basis as pay for grievance time. The parties shall complete the appointment of their respective members within thirty (30) days from the date of signing the Collective Bargaining Agreement.
- B. Either party may at any time remove a member appointed by it and may appoint a new member to fill any vacancy among its appointed members. Each party shall notify the other party in writing of the members or replacements that they appoint before such member or replacement may serve on the Pension Board and before such appointment shall become effective.
- C. The members appointed by either party may, by giving prior notice to the members appointed by the other party, have one advisor attend any meeting to advise them and to assist in presenting their views to the Pension Board as hereinafter provided. In such an event, the members appointed by the other party may also have one advisor

present in the same capacity. Neither advisor shall be considered a member of the Pension Board nor shall have a right to cast a vote.

Section 2. Secretary

- A. Immediately upon the appointment of the Pension Board members by the parties, the Company shall appoint a Secretary to the Board who shall serve until such time as he/she may resign, or die, or until removed by a majority vote of the Pension Board. The Secretary shall attend all Pension Board meetings, but shall not be a member of the Board and shall not have any right to vote at any Board meeting.
- B. In the event of the removal d the Secretary by resignation, death, or written notice of removal by the Pension Board, the Company shall appoint a new Secretary.
- **C.** The duties of the Secretary shall include the following:
 - (1) To receive on behalf of the members of the Board, the information to be furnished to the Pension Board by the Company as hereinafter provided.
 - (2) To receive on behalf of, and at the next regular meeting to distribute to the members of the Pension Board, a copy of any notice by the Company to an employee of that employee's normal or postponed retirement date.
 - (3) To receive on behalf of the Pension Board any written pension applications filed in triplicate (as set out in Exhibit "P-4," Application for Pension, on forms furnished by the Company) concerning any member(s) of the bargaining unit represented by the Union for review at the next regular Board meeting. A copy of Exhibit "P-4," Application for Pension, and a pension calculation estimate will be furnished to the Union members at least five working days in advance of the regular Board meeting.
 - (4) To receive and present to the Pension Board at the next regular meeting, employee requests for review (as set out in Exhibits "P-6" and "P-2" on forms

- furnished by the Company) which are appealed to the Board.
- (5) To secure additional information requested by the Pension Board for the review and disposition of cases presented to it.
- (6) To notify employees and to authorize the Trustee to make pension payments in the detailed amounts as directed by the Pension Board. Such notice and authorization will be in the amounts as set forth on "P" Exhibits.
- (7) To take the official minutes and keep all official Pension Board records, such minutes being subject to approval by the members of the Pension Board, sufficient copies of which will be furnished to the members of the Pension Board. "Sufficient" shall mean six copies for the three members appointed by each party.
- (8) To notify the Pension Board members prior to any monthly meeting date, when there is business to be considered by the Pension Board. Written notice of meetings, including copies of all "P" Exhibits to be acted upon, will be furnished to Pension Board members at least five working days in advance of the regular Board meeting.
- (9) To furnish to the Union members one copy of all "P" Exhibits signed by Pension Board members.
- (10) Notify the Chairperson, at the direction of the Board, of any meeting at which a tie vote is to be resolved.
- (11) To perform such other incidental duties as the Pension Board, within its powers, directs.

Section 3. Authority and Jurisdiction

- A The Pension Board, in connection with employees in the bargaining unit represented by the Union, shall have, but be limited to, the following authority and jurisdiction:
 - To consider any dispute involving the correctness of any determination of service credit, as hereinafter

- provided which is challenged within the prescribed time limits.
- (2) To consider employee pension applications and employee requests for review as to (1) the employee's amount of service credit, (2) age, (3) the average annual earning for the purpose of computing the amount of monthly pension payable, (4) the amount of monthly pension benefit, if any, payable and (5) disapprove or approve for payment or make the corrections necessary for such approval of payment. Disapproval by the Board of pension applications shall be recorded on the form as set out in Exhibit "P-5" "Notice of Rejection of Application for Pension," and a copy shall be transmitted to the employee by the Secretary of the Board.
- (3) To consider notice of employee terminations under Article III, Section 5, of the John Deere Pension Plan for Wage Employees and to determine and notify such employees of their vested rights under Article III, Section 5, of the John Deere Pension Plan for Wage Employees on the form as set out in Exhibit "P-3."
- (4) To direct the Secretary to secure any additional information required for the review and disposition of cases presented to it.
- (5) To direct the Secretary to forward the authorization to make pension payments in the detailed amounts as directed by the Pension Board to the Trustee and to forward to the employee the "Notice of Retirement." Such transmittal shall be made by the Secretary on the form as set out in Exhibit"P-8."
- **B.** The members of the Board, by mutual agreement may appoint a physician or an approved clinic or the staff of an approved hospital who will determine any dispute between the members of the Board over whether or not any employee in the bargaining unit represented by the Union is, or has remained totally and permanently disabled as defined by the Plan.

- C. The Pension Board shall have no authority to alter, change, detract from or add to any of the provisions of the John Deere Pension Plan for Wage Employees or this Appendix "1" and must interpret and apply the provisions of the John Deere Pension Plan for Wage Employees only insofar as such interpretation and application are necessary to make a determination of the individual case under consideration.
- D. The Pension Board's determination of matters properly referred to it as herein provided shall be final and binding on both parties to this Collective Bargaining Agreement and the employees affected thereby.
- E. The Board may, from time to time, review its forms, methods of distribution of necessary information to employees and its operations, such as evidence for verification of age, birth certificate, employment records, etc., and make changes therein consistent with all other provisions of the Plan and this Agreement with the approval of the Union and the Company. The review provided for in this Section shall not be within the jurisdiction of the Chairperson, and he/she shall have no authority to vote upon such matters.

Section 4. Procedure

- A Two Company members and two Union members shall constitute a quorum of the Pension Board, and no business of the Pension Board may be transacted without a quorum. At all meetings of the Pension Board, the Company members shall have a total of three votes and the Union members shall have a total of three votes. The vote of any absent member shall be cast by the members present who were appointed by the same part that appointed the absent member.
- **B.** Decisions of the Pension Board shall be by a majority vote of the Pension Board.
- C. In the event of a tie vote by the members, the Pension Board shall direct the Secretary, as hereinafter provided, to notify the Chairperson of the tie vote and to request his/her attendance at the next regular meeting.

However, if it is impossible for the tie vote to **be** resolved before the pensioner's payment(s) become due, the applicant will be paid or not paid as originally determined (in the case of an employee request for review); and in other cases, after disposition of such cases by the participation of the Chairperson and effective **as** of the date of retirement as **determined** by the Board.

- D. The Pension Board shall meet once each month at a time to be agreed upon by the members of the Board in an office to be furnished by the company, unless there is no business for the Board to consider.
- E. Although there is no official business to be considered, a regular monthly meeting of the Board can still be held, or a special meeting of the Board may be called by unanimous vote of the members.

Section 5. Chairperson of the Board--Selection of

- A. Within thirty (30) days after their selection, the members of the Board, by unanimous vote, shall appoint an impartial umpire who shall act as a member and Chairperson of the Board in connection with those issues on which that person has been notified to attend, and who shall serve until such time as he/she may resign or dies or until he/she may be discharged by any three (3) members of the Board. In the event the Chairperson is discharged, the discharging members shall do so by notifying the Chairperson and the other three (3) members in writing, stating the reasons therefore.
- **B.** The fees and expenses of the Chairperson shall be shared equally by both parties to this Agreement.

Section 6. Chairperson of the Board-Duties

The Chairperson of the Board shall function as follows:

A. In the event of a tie vote by the members, the Board Chairperson shall attend meetings, be considered a member of the Board and have the right to cast the deciding vote, and then only in connection with issues within the authority and jurisdiction of the Board.

- **B.** The Chairperson shall have the same authority and jurisdiction as other members of the Board on issues properly to be considered by such chairperson, except that he/she shall also act as Chairperson of the Board during its consideration of such issues.
- C. The Chairperson of the Board, before voting on such issues, shall give the members of the Board who so desire an opportunity to present their claims and views in such manner as they may elect.
- D. The Chairperson, in considering such issues, may determine the relevancy of any evidence presented.
- E. The decision of the Chairperson shall be reduced to writing, and the concurring and/or dissenting votes of the members shall be noted thereon. Any dissenting member may file an explanation of the minority vote with the Chairperson's decision.
- F. The Chairperson's decision shall set out his or her vote, and shall also explain it by citing the provisions of the Plan, the facts, and reasoning on each issue involved.
- G. The Chairperson's decision, along with the explanations of the minority votes, if any shall on such issues constitute the decision of the Board and a copy shall be furnished to each member of the Board, the Board secretary, and to each party to this Agreement.

Section 7. Request for Review

- A. Employee request for review over eligibility may be referred to the Pension Board as follows:
 - (1) Such requests for review over the eligibility of an employee to receive a pension, or over the amount of such pension, must be referred in writing and signed by the employee affected. It must be referred to the Secretary of the Pension Board at the office of the Company within thirty days from notice to the employee by the Secretary of the Pension Board as to the employee's retirement, or within thirty days from the Secretary's notice of the Pension Board's decision on any application for pension. Such requests for review must be filed

with the Secretary on Exhibit "P-6" furnished by the Company.

- B. Employee requests for review over total and permanent disability may be referred to the Pension Board as follows:
 - (1) Such requests for review may be filed by any employee over whether he/she is totally and permanently disabled as defined by the John Deere Pension Plan for Wage Employees and as determined by the Pension Board. Such requests for review must be in writing, signed by the employee, and filed with the Secretary of the Pension Board within thirty days from the date of the Company's written notice on the matter to the employee, on Exhibit "P-6" furnished by the Company.
- C. Employee requests for review of service credit may be referred to the Pension Board as follows:
 - (1) Any employee who, upon receiving from the Company a notice, "Statement of Service Credit Not Earned," on Exhibit "P-1" and as provided for in Article II of the John Deere Pension Plan for Wage Employees may within thirty days from receipt of such notice file with the Secretary of the Pension Board a Request for Review of such statement on Exhibit "P-2" furnished by the Company.
 - (2) Any employee who, upon receiving from the Company a, "Notice of Denial of Re-establishment of Service Credit," on Exhibit"P-14" and as provided for in Article II of the John Deere Pension Plan for Wage Employees may within thirty days from receipt of such notice file with the Secretary of the Pension Board a Request for Review of such statement on Exhibit "P-2" furnished by the Company.
- D. Questions concerning pensions, including any requests for review, shall be handled only as outlined in this Appendix "1" in place of the review procedure provided in the John Deere Pension Plan for Wage Employees,

Appendix "A", Article I, Section 13-B and shall not be filed or handled as grievances or be subject to arbitration under the terms of the Collective Bargaining Agreement between the parties, and no arbitrator shall have authority to undertake consideration of any such questions or request.

E. In the event the seniority and employment of any employee is broken under the terms of the Collective Bargaining Agreement; and if there is any complaint on the part of the employee, the issue, including any question of the employee's inability to perform efficiently work for the Company, will be determined through the Grievance Procedure (including arbitration) as provided for in the Collective Bargaining Agreement. However, the employee being so removed may be eligible for benefits under the John Deere Pension Plan for Wage Employees: and any questions within the jurisdiction of the Pension Board shall be determined by the Pension Board.

Section 8. Reports to the Board

- A. The Company will furnish to the Secretary for transmittal to the Pension Board the following information with respect to the operation of the John Deere Pension Plan for Wage Employees:
 - (1) A monthly statement showing the name, age and amount of service credit of employees in the bargaining unit represented by the Union, if any, who have filed pension applications with the Secretary since the last meeting of the Pension Board.
 - (2) The complete annual report of the actuarial valuation of the plan prepared by the enrolled actuary for the plan.
 - (3) An annual report from the Administrator showing the names of retired employees of the bargaining unit and the total of the disbursements to each during the year.

- (4) Such information as to the age, sex, earnings, and service of employees covered by the John Deere Pension Plan for Wage Employees, and amount of pensions and supplemental allowances by age groups as the Board may reasonably require, but in no event shall the Company be required to furnish the Board with any data not furnished by the Company to the actuary.
- **B.** The Company will provide the Pension Board with such clerical assistance and office supplies as the Pension Board Secretary may require in the performance of his/her duties

ARTICLE II UNION OFFICIALS

Section 1. Service Credit

A. In the event that any regular full-time National Union Representative on leave from the Company or a Company employee who serves as a Local President, Financial Secretary, Chairperson, Shop Committeeperson, or Union Time Study Representative does not work 500 hours in any anniversary year, as provided for in Article II of the John Deere Pension Plan for Wage Employees, then any leave from active employment for such National Union business or for Local Union business, under the provisions of Article XVII, Section 17.03, "Leaves of Absence" of the Collective Bargaining Agreement between the parties shall be treated as time worked for purposes of qualifying for a year of service credit for such year.

Section 2. Pension Benefits

A. The Local Union representatives listed above who lose time during regular scheduled work hours (not to exceed eight hours in any scheduled workday or forty hours in one scheduled workweek) for conducting Union business and for which they are not paid by the Company, but are paid by the Union, will, only for the purpose of computing pension benefits, be credited with such earnings on the basis of their average straight-time

- hourly earnings. The procedure for administering this provision will be determined by the Company.
- B. Any regular full-time National Union Representative on leave from the Company, who retires on or after 1 November 1989, and who has accrued service credit as provided in Section 1-A above, shall at retirement have his/her average monthly earnings, as determined on the effective date of such leave, as defined in Appendix A, Article III, Section 1-C, adjusted on the effective date of each applicable cost-of-living allowance(s) and general wage increase(s), if any, which become effective immediately after commencement of such leave and prior to the date of retirement.

Section 3. Life Insurance

A Local Union representative listed above who is absent from work for Local Union business as set out in Article XVII, Section 17.03, "Leaves of Absence" of the Collective Bargaining Agreement between the parties may request that such absence be treated as time worked for the purpose of earnings brackets and qualifications under the John Deere Life and Disability Insurance Plan for Wage Employees.

ARTICLE III PART-TIME EMPLOYEES

Section 1. Part-Time Employees

Employees covered by the Collective Bargaining Agreement that work not more than 24 hours per week are considered Part time employees and will be eligible for only the minimum amount of benefits allowed by law.

APPENDIX "A" JOHN DEERE PENSION PLAN FOR WAGE EMPLOYEES

ARTICLE I ESTABLISHMENT OF PLAN

Section 1. Purpose

The purpose of this Plan, which is to be known as the John Deere Pension Plan for Wage Employees is to promote the mutual interests of John Deere Limited and its subsidiaries, hereinafter designated as the Company, and the Wage employees of the Company, except employees of retail establishments in Canada.

Section 2. Object

The object of this Plan is to provide disability and retirement benefits for employees, supplementing benefits payable under the Old Age Security Act or other statutory benefits. The benefits payable under this Plan are at all times subject to the requirements, minima and maxima imposed by the Ontario Pension Benefits Act RSO 1990 and regulations there-under, the Income Tax Act (Canada) and regulations thereunder and any other legislation, binding directive or administrative rules of any governmental authority to which the Plan is or may become subject as the same may be amended from time to time. Hereinafter such legislation and regulations will be referred to as "Applicable Legislation".

John Deere Limited shall be the Administrator of this Plan.

Section 3. Cost of Benefits

Except as hereinafter provided, the cost of providing benefits under this Plan will be borne by the Company.

The expenses incurred in the operation of this Plan shall be paid from the Trust Fund unless paid by the Company.

Section 4. Effective Dates

The effective date of this Plan shall be 1 April 1955, with respect to employees of the Company in active service on or

after such date. Except as otherwise provided, the Plan as amended herein shall be effective from 1 October 2007.

The Fiscal Year of this Pension Plan commences on the 1st day of January and ends on the 31st day of December.

Section 5. Statutory Benefit Defined

"Statutory benefit" as used in this Plan means unreduced benefits from the Old Age Security Act, the Canada or Quebec Pension Plan or any equivalent federal or provincial law providing old age benefits." The statutory benefit under each act or law providing statutory benefits shall be considered to commence on the date on which the retired employee first receives or shall become eligible to receive statutory benefits under such act or law. Any changes in the amount of the statutory benefit under such act or law after the retired employee first receives or becomes eligible to receive such statutory benefits shall not affect benefits under this Plan. Any reduction or cessation of statutory benefits because the retired employee has disqualified himself or herself shall be disregarded, and the amount of statutory benefit considered under this Plan shall be the amount that would have been payable to the retired employee except for such disqualification.

Section 6. The Trust Fund and the Trustee

A trustee shall be designated by the Company and a Trust Agreement executed between the Company and such Trustee, under the terms of which a Trust Fund will be established to receive, hold, invest, and reinvest contributions made by the Company, interest and other income, and to pay the retirement benefits provided for in Article III, Article IV, Article V and Article VI of this Pian.

The Company will determine the form and terms of any such Trust Agreement, may remove any Trustee or select any successor Trustee, and may modify, amend or alter any such Trust Agreement at any time.

The investments of the Trust Fund shall be restricted to the securities and loans prescribed by Applicable Legislation and all investments shall be made in accordance with the terms

of the Trust Agreement and the Statement of Investment Policies and Goals.

Section 7. Payments to the Trust Fund

The Company will, as specified below, contribute such amounts, as are determined by an actuary or firm of actuaries chosen by the Company and qualified through Fellowship of the Canadian Institute of Actuaries, hereinafter referred to "the Actuary", necessary to provide the pension, ancillary and other benefits in accordance with Applicable Legislation.

In particular, such amounts include payments made in respect of the normal cost of benefits under the Plan and such special payments for the liquidation of any going concern unfunded actuarial liabilities and solvency deficiencies in accordance with and within the limits permitted by Applicable Legislation.

Contributions required in respect of the normal cost of benefits shall be paid into the Trust Fund in monthly installments within thirty days after the month for which contributions are payable.

Special Payments towards any going concern actuarial liabilities or solvency deficiencies shall be paid into the Trust Fund in monthly installments.

Section 8. Administration and Termination

Except as otherwise specifically provided, this Plan shall be administered by the Company and shall be subject to amendment, modification or termination by action of the Board of Directors of the Company at any time. If this Plan is ever terminated, the assets comprising the Trust Fund shall be used and applied first towards the provision of pensions and other benefits of employees, former employees, pensioners, spouses, dependents and estates accrued in accordance with these provisions with respect to service up to the date of such termination in an equitable manner determined by the Company, based on the advice and recommendation of the Actuary and subject to compliance with applicable legislation as the same may be amended

from time to time. If those assets on the date of such termination exceed the amount required to provide for all such accrued pensions and other benefits, the excess shall revert to the Company. In no event shall any part of those assets revert to the Company until provision has been made for all such accrued pensions and other benefits, and in no event shall such provision be made that would result in benefits to an employee being in excess of the maxima determined in accordance with Article VII.

Section 9. Nonencumbrance of Benefits

Assignment, pledge or encumbrance of any benefits under this Plan will not be permitted or recognized and benefits shall not be subject to attachment, execution, garnishment, or other legal process, provided, however, that an employee's interest in monies payable under the Plan are subject to assignment by an order under the Ontario Family Law Act, 1986 or by a domestic contract as defined in that Act or any such similar legislation of another province of Canada, if appropriate.

Section 10. Service Credit

All service credit will be determined in accordance with the Company's Service Credit Plan as set forth in Article II.

Section 11. Vesting

No employee shall acquire a vested interest (except as provided by Section 5 of Article 111) under the pension provisions of this Plan until retirement or eligibility therefore, and in either case, such interest shall be only as defined in Article III, Article IV and Article V.

Section 12. Nonparticipation

No employee who is first hired after attainment of his/her sixty-fifth (65) birthday shall become a participant in or acquire rights to any benefits under this Plan.

Section 13. Denial of Benefits

- A. When benefits provided by this Plan are denied in full or in part, the employee or eligible beneficiary shall receive a written notice of the reason or reasons for such denial.
- B. Except as otherwise specifically provided, a request for review of the denial of benefits may be submitted within 60 days of such denial, in writing, to the Plan Administrator. The Plan Administrator will reply to such request within 60 days.

Section 14. Administration

The Administrator shall provide every employee covered by this Plan a written explanation of the terms and conditions of the Plan and amendments thereto, and information concerning his/her rights and duties under the Plan, together with a description of the benefits available to him/her under the terms of the Plan and any other information required by Applicable Legislation.

A copy of all documents and any other information prescribed by Applicable Legislation may be examined by any employee covered by the Plan at any reasonable time at such place as the Company may designate.

Except as otherwise specifically provided, in order for any employee to receive pension benefits pursuant to any provisions of the Plan, he/she must file a written pension application on forms furnished by the Administrator. Such pension application shall state the reason for retirement and the date upon which the proposed retirement is to become effective and shall be made not less than sixty days in advance of such date.

ARTICLE II PLAN FOR COMPUTING AND MAINTAINING EMPLOYEES' SERVICE CREDIT

Section 1. Company/Employee Defined

A. Company Defined

Only for the purpose of determining service credit hereunder the word "Company" as used herein shall include John Deere

Limited and its subsidiaries, its parent Deere & Company, a Delaware Corporation, and the various domestic and foreign subsidiaries of Deere & Company

B. Employee Defined.

For the purposes of this and subsequent Articles, "Employee" should be taken to mean all wage employees of the Company. All reference to male gender shall be taken as applying on an identical basis to the female gender.

Section 2. Plan Participation and Starting Date

The starting date of any employee's participation in the Plan shall be the date on which the employee first reports to work for the Company in his/her last continuous employment with the Company.

Section 3. Service Credit Records

A present employee's participation in the Plan, continuous employment and service credit shall be as established by the Company's service credit records.

The service credit record for the individual should indicate both his/her Past Credited Service, this being service credited prior to 1 January 1992, and his/her Future Credited Service, this being service credited after 31 December 1991.

The sum of the Past Credited Service and Future Credited Service of an individual is his/her Service Credit.

The remaining sections of this Article deal only with, the calculation of Future Credited Service; the Past Credited Service in respect of each individual having been accrued in accordance with the Plan Text and legislative restrictions as they applied prior to 1 January 1992.

Section 4. Service Credit Defined

When an employee's starting date for participation and service credit have been established in accordance with the provisions of Section 3 (or are established under Section 2 in the case of an employee hired after the date of this Plan), then on each succeeding anniversary of his/her starting date he/she will be credited with one year of continuous

employment and one year of participation, and will also be credited with one year of service credit, provided he/she has worked 500 hours or more for the Company in that anniversary year; however, no service credit shall be credited after the employee attains age 69.

When total hours worked within an anniversary year are less than 500, he/she will not receive service credit for that year, and this fact shall be recorded in writing and a copy will be furnished to the employee.

Section 5. Time of Crediting

Five hundred or more hours worked within an anniversary year will not be credited to the employee as a year's service credit until the end of that anniversary year.

Section 6. Partial Year's Service Credit

- A. For the purpose of computing pension benefits, an employee retiring or attaining age 69 prior to the end of an anniversary year shall receive service credit at the rate of 8-1/3 percent of one year's service credit for each month from the employee's last anniversary date to the date of retirement or to the date the employee attains age 69, whichever is earlier, providing such time would have been creditable had the employee's continuous employment continued to the next anniversary date.
- B. For the purpose of computing pension benefits, an employee whose employment with the Company is terminated on or after 1 January 1977 and prior to the end of an anniversary year shall receive service credit at the rate of 8-l/3 percent of one year's service credit for each month from the employee's last anniversary date to the date of termination or to the date the employee attains age 69, whichever is earlier, providing such time would have been creditable had the employee's continuous employment continued to the next anniversary date.

Section 7. Occupational Injury

Time lost due to injury sustained in the active employment of the Company will be credited as time worked subject to the following conditions:

- a) the injury sustained prevents the employee from performing the same work that he/she was doing immediately prior to becoming disabled,
- the Administrator is satisfied, on the basis of written certification from a medical doctor, licensed to practice under the laws of a province of Canada, that the period is a period of disability,
- c) the period of time credited is equal to that which the employee would have worked but not to exceed a maximum of forty (40) hours in any work week and, in any event, not beyond the date that the individual attains Normal Retirement Age.

Section 8. Nonoccupational Injury or Illness

- A Time lost on or after 1 April 1971 and prior to attainment of Normal Retirement Age due to non-occupational injury or illness and subject to the conditions imposed by Article II, Section 7, Paragraphs a) and b), for which the employee receives:
 - (1) a weekly indemnity benefit, will be credited as time worked but not to exceed a maximum of forty (40) hours in any one work week: and
 - (2) a long term disability benefit, will be credited at the time the employee becomes eligible for a normal or early retirement benefit as time worked in an amount equal to that which the employee would have worked but not to exceed a maximum of forty (40) hours in any one work week.
- B. An employee receiving benefits as described in A(2) above or under Section 3-C of Article III who recovers and is subsequently re-employed shall be credited with the service credit he/she had when the disability commenced. In addition, for the period of such disability, such employee will at the date of re-

employment receive credit as time worked in an amount equal to that which the employee would have worked but not to exceed a maximum of forty (40) hours in any one work week.

Section 9. Layoff

Time lost on or after 1 January 1971 due to layoff for which the employee receives a Supplemental Unemployment Benefit will be credited as time worked in an amount equal to that which the employee would have worked, but not to exceed a maximum of 40 hours in any one workweek. The maximum period of time will be determined on the basis of one month for each four credit units the employee was credited with as of the date of his/her layoff, but not to exceed 12 months. Further limitations on the total amount of additional service credit which can be accrued under this section are imposed by Section 7 of Article VII.

Section 10. Military Service

An employee who left his/her employment with the Company and immediately entered the service of the Armed Forces of Canada, will be considered as having been employed and his/her military service will be credited toward service credit up to a maximum of 40 hours per week and in the same manner as set out in Section 4, provided he/she has reemployment rights under any existing law, and does become reemployed under the provisions of the law. Further limitations on the total amount of additional service credit which can be accrued under this section are imposed by Section 7 of Article VII.

Section 11. Accumulation in Any Unit

Participation in the Plan and service credit may be accumulated by an employee in any unit **of** the Company, subject to all the provisions herein.

Section 12. Transfer

When an employee is formally transferred from one unit of the Company to another unit of the Company, the unit from which the employee is being transferred will furnish to the new unit a written statement of the employee's service credit status and the new unit will furnish a copy of same to the employee.

Section 13. Crediting Time Spent at Other Units

An employee working for one unit of the Company who is on layoff and subject to recall to another unit of the Company shall, when recalled and reemployed by the original unit, receive credit for time spent at the other unit, or units, for the purpose of participation, continuous employment and service credit in accordance with the provisions of the Plan. If the employee is not recalled to the original unit within the period in which the employee is subject to recall, or having been recalled, he/she notifies both the original unit and the current employing unit, within five working days, that he/she has elected to remain in the current employing unit and thereby forfeit recall rights to the original unit, then the employee will be considered a formal transfer to the new unit as set out in Section 12.

Section 14. Termination of Participation, Continuous Employment and Service Credit

An employee's participation in the Plan, continuous employment and service credit will be terminated by any one of the following occurrences:

- A. Death or retirement or determination of permanent and total disability as specified in Article III, Section 3, of this Plan and Article II, Section 4 of the John Deere Group Life and Disability Insurance Plan for Wage Employees, except that an employee who has been determined to be totally and permanently disabled shall receive service credit as provided in Section 8-A-(2) and further provided in the event such employee recovers and is subsequently reemployed shall have continuous employment reinstated as though he/she had been continued on leave of absence during the period of total and permanent disability.
- **B.** Quitting the job, either with or without notice.

- C. Discharge, provided, however, if it is later determined that the employee was discharged without good and just cause and the employee is reinstated, then service credit will also be reinstated.
- **D.** Failure on the part of the employee to return to work on the required date after a leave of absence.
- E. Failure on the part of a laid-off employee to return to work within five working days when notified to do so by the Company at the last known address, unless prevented from doing so by a reason satisfactory to the Company.
- **F.** Failure to return to work from military service within the period during which he/she has reemployment rights under any existing law.
- G. Subject to Section 14 above, any absence from active employment other than formal leave of absence in writing granted by the Company, military service as set out in Section 10, or plant-incurred injury as set out in Section 7, for a period of time equal to the employee's employment prior to such absence, or for a period of five consecutive years, whichever is the lesser, provided, however, that no employee who has completed the probationary period will lose continuous employment and service credit unless such absence exceeds two years.

Section 15. Re-establishment of Service Credit

- A. Subject to Paragraph C below, any present employee or any employee hired after 1 January 1988 whose participation, continuous employment and service credit is terminated will, if rehired, have the service credit held at the time of termination reestablished.
- B. Subject to Paragraph C below, any former employee who is rehired after 1 January 1988 whose participation, continuous employment and service credit was terminated will have the service credit held at the time of termination reestablished.
- C. Notwithstanding the foregoing Paragraphs A and B, service credit of a former employee who is rehired will not be reestablished if the former employee requested a

transfer payment in accordance with Section 5-G of Article III and such transfer payment was made. Furthermore a former employee, who is rehired shall not be credited with additional service credit while in receipt of a pension benefit.

Section 16. Original Unit Defined

The words "original unit" as used in Section 13 shall mean the unit at which an employee's starting date for participation, continuous employment and service credit was established as set out in Section 2 or the last unit to which he/she has been formally transferred.

ARTICLE III ELIGIBILITY FOR RETIREMENT AND AMOUNT OF PENSIONS

Section 1. Normal Retirement - Age 65

- A. An employee on attaining the Normal Retirement Age of age 65 in the employment of the Company and having at least 2 years of participation in the Plan is eligible for normal retirement benefits.
- **B.** The amount of benefits allowed monthly upon retirement will be the formula pension in (1) below, but not less than the minimum pension in (2) below and subject to the maximum limits imposed by both Section 3 (A) and Section 5 of Article VII.
 - (1) Formula Pension. The sum of:
 - a. 0.5 percent of the first \$350 or less of the average monthly earnings as defined in Paragraph C of this Section for each year of service credit up to and including 25.
 - b. 1.5 percent of the excess over \$350 of the average monthly earnings as defined in Paragraph C of this Section for each year of service credit up to and including 25.
 - c. 1.5 percent of the average monthly earnings as defined in Paragraph C of this Section for each year of service credit in excess of 25.

- (2) Minimum Pension.
 - a. For retirements on or after 30 September 2007 \$45.55 per month for each year of service credit.
- C. The average monthly earnings figure used for the purpose of computing a formula pension in accordance with Section 1-B-(1) shall be the total of all straight-time wage payments plus vacation pay, personal absence pay and pay for unworked holidays, received in the highest 5 consecutive anniversary years during the 10 anniversary years immediately prior to the earlier of retirement or attainment of age 69 in each of which service credit was received, divided by 60, but in no event shall the average monthly earnings figure exceed \$2,750.

If, in this 10-year period, there is any full anniversary year (or years) in which service credit was not received, that year shall be omitted and the 11th year preceding retirement, or any additional anniversary years in the same order, i.e., the 12th year, etc., shall be used to obtain 10 anniversary years in which service credit was received.

In the event service credit was received on account of time lost due to compensable accident in one or more of the 10 years as outlined above, the amount of temporary total, partial permanent and total permanent compensation will be considered as earnings.

In the event service credit was received on account of military service in one or more of the 10 years as outlined above, such years shall be omitted and the 11th year preceding retirement, or any additional anniversary years in the same order, i.e., the 12th year, etc., shall be used to obtain 10 anniversary years in which service credit was received for other than military service. In the event such employee has less than 5 years of service credit where earnings are credited, the average monthly earnings to be used in determining retirement benefits shall be determined by using the earnings in the years of service credit wherein earnings are credited, divided by the number of months in these years.

Section 2. Early Retirement

- A. An employee may retire at his/her option, if such employee
 - (1) has attained age 55 but not 65 and who at that time has at least 2 years of participation in the Plan, or
 - (2) has 30 or more years of service credit.
- B. In such event, an employee retiring on or after 01 January 1992 shall be entitled to benefits in a reduced amount based upon service credit and earnings prior to such retirement and computed in accordance with Section 1 of this Article. Subject to the maximum limits imposed by both Section 3 b) and Section 5 of Article VII, such reduced amount shall be the greater of:
 - (1) The formula pension as determined in Section 1-B-(1) of this Article, based upon the service credit and earnings prior to such retirement reduced by 1/3% for each month the employee is under age 62 at the date of retirement.
 - (2) The minimum pension as determined in Section 1-B-(2) of this Article, based upon service credit prior to such retirement reduced by 1/3% for each month the employee is under age 62 at the date of retirement except that an employee who retires with 30 years of service or whose age and service credit total 85, such reductions shall not apply.
- C. An employee retiring prior to Normal Retirement Age under Section 2-A of this Article may elect, at the time of retirement, to receive benefits commencing at any time up to Normal Retirement Age, in lieu of the benefits under Section 2-B of this Article. The amount of such benefits will be computed in accordance with Section 2-B of this Article subject to the minimum pension as determined in Section 1-B-(2) of this Article with appropriate reduction under Section 2-B based upon the date benefits commence. The date these benefits are to commence after retirement but prior to Normal Retirement Age, as herein provided, shall be designated by the employee at the time of retirement and is subject to change by the employee provided notice is received

not less than 60 days prior to the designated commencement of benefits.

- D. An employee retiring prior to age 62 on or after 31 October 1989 and before 1 January 1992 who has 30 or more years of service credit or whose combined years of age and service credit total at least 85 shall, upon reaching age 62 have his/her minimum pension, which was determined in accordance with Section 1-B-(2) and reduced in accordance with Section 2-B-(2), redetermined without reduction because of age.
- E. An employee retiring on or after 1 October 2007 who does not elect to receive a Supplemental Allowance under Article V, Section 2A or 2B, shall receive a Supplemental Transitory Benefit commencing at the date of commencement of benefits under Section 2-C of this Article III and payable until attainment of age 65, of \$16.00 per month for each year of service credit, not to exceed \$480.

The Supplemental Transitory Benefit shall be reduced by, 1/2% for each month the employee is under age 62 at the date of commencement of benefits, except that for an employee with 30 years of service, or whose age and service credit total 85, such reductions shall not apply.

If a retired employee entitled to receive a Supplemental Transitory Benefit has earnings after retirement in excess of \$18,000 in any calendar year, a penalty equal to one times the amount by which such earnings exceed the amount permitted shall be charged against each succeeding Supplemental Transitory Benefit which the employee would otherwise be entitled to receive until the full amount of such penalty is satisfied, it being understood that penalties and charges shall be cumulative if appropriate. Each retired employee receiving a Supplemental Transitory Benefit may be required to certify that his/her earnings have not been in excess of the permitted amount, to furnish verification of the amount of such earnings, or to authorize the Company to make inquiries regarding these earnings from the Department of National Revenue, Taxation. The amount of any overpayments of Supplemental

Transitory Benefit made after a retired employee shall have ceased to be entitled to receive such allowance, in whole or in part because of excess earnings, may be deducted from future monthly retirement benefits payable under the plan, provided, however, that the employee may make repayment in a lump sum amount.

F. Special Early Retirement

- (1) In connection with actions taken by the Company with respect to the 2007 Collective Bargaining Agreement, the special early retirement benefits described in paragraph (2) below shall be paid to employees eligible for such benefits who elect to retire during one of the following temporary Special Early Retirement windows:
 - beginning on 01 October 2007 and ending on 31July 2008;
 - beginning on 01 October 2008 and ending on 31July 2009;
 - beginning on 01 October 2009 and ending on 31July 2010.

An employee shall be eligible for such benefits if, on their retirement date, he/she has attained at least 30 years of service or is at least age 55 with 10 years of service. During each of these temporary Special Early Retirement windows the Company will identify a retirement date which is in alignment with low production periods. Employees electing to retire during one of these temporary Special Early Retirement windows must elect to retire on the retirement date, identified by the Company. Such retirement dates will be communicated no less than 3 months in advance of the date.

(2) An employee who is retired under the provisions of paragraph (1) above shall be entitled to benefits determined in accordance with the preceding paragraphs of this Section 2, with the following two exceptions:

- (a) The minimum pension, as described in Section 1-B-(2), shall instead be equal to \$50.05 per month for each year of service credit accrued up to their retirement date.
- (b) For each of the Special Early Retirement windows defined in Section 2-F-(1) above, the Supplemental Transitory Benefit, as determined in Section 2-E of this Article III, shall instead be calculated using an amount of \$17.00 per month for each year of service, not to exceed \$510.00, paid in accordance with the provisions of Section 2-E of Article III
- (3) For each of the Special Early Retirement windows defined in Section 2F(1) above, the Supplemental Allowance, as determined in Section 2C of Article V, shall instead be calculated using a Total Monthly Benefit of \$2,650. An employee retiring with less than 30 years of service credit shall have this Supplemental Allowance adjusted by multiplying the Total Monthly Benefit by the employees' service credit and dividing by 30; the reduction described in paragraph (2) of Section 2D of Article V shall not apply to employees retiring under Section 2F(1) of this Article III. Section 4 of Article V shall continue to apply to the Supplemental Allowance described above.
- (4) The benefits provided under this Section 2F d Article III shall not be made available to any employee who terminates from the Plan for any reason other than retirement or death, or who does not meet the eligibility conditions under Section 2F(1).
- (5) The Company may from time to time designate by plan amendment other dates on which an employee may elect to retire under the provisions of this paragraph F.
- (6) All benefits payable under this paragraph F are subject to the maximum limits imposed by Sections 3, 4 and 5 of Article VII."

Section 3. Total and Permanent Disability Retirement

- A. An employee shall be eligible upon retirement to receive disability benefits hereinafter provided, if such employee becomes totally and permanently disabled on or after 1 May 1987, has not attained age 65 and has at least 10 years' service credit.
- B. An employee shall be deemed to be totally and permanently disabled when the Administrator is satisfied, on the basis of a written certification from a medical doctor, licensed to practice under the laws of a province of Canada or of a place where the member resides, that the employee is suffering from a physical or mental impairment that prevents him/her from engaging in any employment for which he/she is reasonably suited by virtue of his/her education, training or experience and that can reasonably be expected to last for the remainder of his/her lifetime.
- C. In the event of retirement because of total and permanent disability, an employee shall be entitled to receive benefits in the following amounts:
 - (1) Total and Permanent Disability Benefit: the formula pension in accordance with Article III Section 1, Paragraph B(1) but not less than the minimum pension in Article III Section 1, Paragraph B(2), based upon service credit at the time of such retirement including time not credited the employee while on Long Term Disability.
 - (2) An Additional Temporary Benefit commencing at early retirement until attainment of age 65 or eligibility for an unreduced statutory benefit, whichever first occurs, and subject to the maximum limits imposed by Section 4 of Article VII as follows:
 - a. For retirements on or after 30 September 2007, \$35.80 per month for service as defined in (1) above, not to exceed \$1074.00, reduced, but not below zero, by the excess, if any, of the formula pension over the minimum pension as calculated for the preceding paragraph (1).

- D. An employee retired because of total and permanent disability under the 1955 Plan and the 1955 Plan as amended thereafter shall have the benefit redetermined in accordance with Section 1 of this Article at age 65.
 - Service credit applicable to this redetermined benefit shall for employees who became totally and permanently disabled prior to 01 January 1992 include service credit for the anniversary years during which disability benefits were received. For employees who became totally and permanently disabled after 01 January 1992, service credit applicable to this redetermined benefit shall exclude service credit for the anniversary years during which disability benefits were received. The average monthly earnings shall be for the highest 5 consecutive anniversary years during the 10 anniversary years (in each of which service credit was received) immediately prior to the last day of work preceding total and permanent disability.
- E. Benefits under this Plan on account of total and permanent disability shall be payable and retirement shall commence as of (1) the first day of the seventh month following the commencement of the employee's continuous incapacity to work because of such disability, or as of (2) the first day of the month after the employee is found to be totally and permanently disabled as set out in Section 3-B of this Article, whichever is later.

Section 4. Automatic Retirement

- A. Any employee shall be deemed to retire on the last day of the month in which his/her 69th date of birth occurs and benefit payments shall commence in accordance with Section 6-A of this Article whether or not such employee ceases active employment.
- **B.** In the event of Automatic Retirement hereunder, such employee shall cease to receive service credit or accrue further benefits under this Plan.

Section 5. Deferred Vested Pension

- A. Any employee whose employment shall terminate on or after 1 January 1988 and who is not eligible for any other type of retirement benefit under this Plan shall be eligible upon making application therefor to receive a deferred vested pension benefit if at the time of such termination the employee has 2 or more years of participation in the Plan.
- B. The former employee shall be entitled to benefits computed in accordance with Section 1 of this Article except that the average monthly earnings shall be for the highest 5 consecutive anniversary years during the 10 anniversary years (in each of which service credit was received) immediately prior to termination.
- C. The earliest commencement date of this pension shall be the first day of the month following the month in which such former employee attains age 55. The pension shall be payable on the first day of the month following the month in which such former employee makes application on or after the earliest commencement date. If application is made at age 65 or later, the pension will be computed on a normal retirement basis. If application is made before age 65, the pension will be the greater of
 - (i) the pension computed on the normal retirement basis and reduced by ½% for each month the former employee is under age 65; or
 - (ii) a reduced pension the commuted value of which is equal to the commuted value of the pension deferred to age 65 and payable on the normal retirement basis.

However, the resulting pension shall be limited to the pension computed on the normal retirement basis and reduced in accordance with the requirements of Regulation 8503(3)c of the Income Tax Act.

D. The provisions for the Survivor Benefit in Section 1 of Article IV shall be applicable to a former employee who is eligible for deferred vested pension benefits as defined in this Section. Such Survivor Benefit shall not be effective prior to the date of application for deferred

- vested pension and in no event, earlier than the earliest commencement date as defined in Paragraph C above.
- E. The application must be submitted not earlier than 60 days prior to the earliest commencement date. The Company will mail notices to former employees who would be eligible to receive deferred vested pension benefits but who have not made applications. The notice to such former employee will be sent on or about the 90th day prior to the employee's 55th, 65th and 69th birthdays, and will be mailed to the last address shown on the Company record.
- F. If an employee terminates employment with a deferred vested pension and is later reemployed, he/she will not be eligible for a deferred vested pension benefit during any such period of reemployment. Unless the employee requested a transfer payment in accordance with Paragraph G below and such transfer payment was made, upon any such reemployment service credit shall be reestablished, and eligibility for retirement benefits shall be determined upon the basis of the remaining sections of this Article rather than upon this Section (unless the employee shall again be terminated upon the circumstances described in this Section, i.e., prior to qualification for early retirement).
- G. Commutation of Vested Retirement Benefit

An employee who is entitled to a deferred vested pension benefit pursuant to this Section and who has not attained age 55 may elect in full settlement of such entitlement to have that benefit commuted and transferred to

- (i) the registered pension plan of the employee's new employer,
- (ii) a registered retirement savings plan designated by the employee, provided such plan includes any locking-in restrictions or other provisions required by applicable legislation, or
- (iii) an insurance company for the purchase of a life annuity that will commence not earlier than the employee's 55th birthday.

Section 6. Commencement and Termination of Pension Benefits

- A. Benefit payments shall be made monthly. The first payment will be payable as of the first day of the month following retirement.
- **B.** The last payment will be made on the first day of the month of the retired employee's death.

ARTICLE IV SURVIVOR BENEFITS

Section 1. Survivor Benefits for Retired Employees

- A. For the purposes of this Article, the term "Spouse" will mean a person who is residing or co-habiting with the Employee, former Employee, or retired Employee, at the date of his/her death, or earlier retirement, provided that:
 - (a) the person has been married to, and residing with, the Employee, former Employee, or retired Employee for at least one year, or
 - (b) the person is not married to, but is living in a conjugal relationship with the Employee, former Employee or retired Employee
 - continuously for a period of not less than three years or
 - (ii) in a relationship of some permanence, if the person and/or the Employee are the natural or adoptive parents of a child.
- B. For an Employee retiring under Sections 1, 2, 3 and 4 of Article III on or after 1 January 1988, a Survivor Benefit shall be payable on the Employee's death after retirement to his/her spouse during such person's further lifetime. The Survivor Benefit for an employee who retired through Total and Permanent Disability incurred prior to 1 January 1992 is payable following redetermination of the Employee's pension as described in Section 3-D of Article III. Should such Total and Permanent Disability retirees die before their pension

benefit is redetermined then they are entitled to alternative survivor benefits calculated in accordance with Sections 2 and 4 of this Article IV. The provisions of this paragraph will be effective on the first day of the month following retirement.

For Employees retired prior to **01** October **1995**, the Regular Surviving Spouse's benefit shall be **55%** of the monthly benefit (excluding any Additional Temporary Benefit or Supplemental Allowance) payable to the retired employee except that the survivor Benefit payable to the surviving Spouse of an employee who retired prior to age 62 on or after **31** October **1989** and who had completed 30 or more years of service credit or whose combined years of age and service totaled at least **85** at retirement shall be not less than **55%** of the minimum pension calculated in accordance with Article III Section 1-B-(2) without reduction in accordance with Article III Section 2-B-(2).

For Employees retiring on or after 30 September 1995, the Regular Surviving Spouse's benefit shall be 60% of the monthly benefit (excluding any Additional Temporary Benefit, Supplemental Transitory Benefit or Supplemental Allowance) payable to the retired employee except that the survivor Benefit payable to the surviving Spouse of an employee who retired prior to age 62 and who had completed 30 or more years of service credit or whose combined years of age and service totaled at least 85 at retirement shall be not less than 60% of the minimum pension calculated in accordance with Article III Section 1-B-(2) without reduction in accordance with Article III Section 2-B-(2).

The retired employee's monthly benefit, however, shall be decreased by ½% for each full year in excess of 10 years that such Spouse is younger than the retired Employee.

C. In the event that the retired employee has elected deferral of benefits in accordance with Section 2-C of Article III, the surviving Spouse may elect to have the Survivor Benefit redetermined as though the retired employee's benefits were to commence on the first of the month following such retired employee's death.

Section 2. Survivor Benefits for Employees

- A. The surviving Spouse will be eligible for a monthly Survivor Benefit, provided the employee:
 - (1) dies on or after 31 December 1979, and
 - (2) had met the age and service requirements for pension benefits under Section 1, 2 or 4 of Article III on the date of death.
- The Survivor Benefit will be equal to the amount which B. would have been paid to the employee's surviving Spouse had the employee retired on the date of death but calculated on the basis of the emplovee's accumulated service credit as at December 31, 1986. Service credit as applicable to this paragraph B and the following paragraph C includes credit for any period, prior to 31 December 1986, during which the individual was a Total and Permanent Disability retiree who became disabled prior to 1 January 1992 and whose benefit had not been redetermined, in accordance with Section 3-D of Article III. at the date of his/her death. Benefits will commence as provided in Section 3 of this Article
- C. A surviving Spouse who is not eligible for benefits under Paragraph A of this Section will be eligible for a monthly Survivor Benefit provided that the Employee dies on or after 30 April 1987 with 10 or more years of service credit.

The Survivor Benefit will be equal to the amount which would have been paid to the surviving Spouse had the employee lived and retired at the employee's earliest retirement date but calculated on the basis of the employee's accumulated service credit as at December 31, 1986. The earliest retirement date shall be determined by the total service credit attained at the time of death plus the employee's projected age at which such employee would first be eligible to retire under Sections 1, 2 or 4 of Article III.

Section 3. Commencement and Termination of Survivor Benefits

- A. Benefit payments shall be made monthly. The first payment will be payable as of the first day of the month following:
 - (1) the retired employee's death in the case of benefits under Section 1:
 - (2) the death of an employee in the case of benefits under Section 2-A;
 - (3) the month in which the employee's earliest retirement date occurs in the case of benefits under Section 2-C.
- B. The last payment will be made on the first day of the month of the Spouse's death.

Section 4. Death Benefits for Employees and Former Employees.

- A. In addition to the monthly Survivor Benefit described in Section 2 above, a Death Benefit will be payable upon the death on or after 1 January 1988 of:
 - (1) any employee who has 2 or more years of participation in the Plan; or
 - (2) any former employee whose employment terminated after 1 January 1988 provided such employee was entitled to a deferred vested pension under Section 5 of Article III and who did not elect Settlement of such entitlement by a transfer payment under paragraph G of the said Section.
- B. The Payee of the Death Benefit will be the Spouse, if any, of the Employee or former Employee, otherwise the beneficiary designated in writing by the Employee or former Employee, otherwise the estate of the Employee or former Employee.
- C. If the payee of the Death Benefit is the Spouse of the employee or former employee, the form of the payment of the Death Benefit will be, according to the election of the Spouse either

- (1) a lump sum payment; or
- (2) an immediate annuity payable for the lifetime of the Spouse; or
- (3) a deferred annuity commencing on a date selected by the Spouse but not later than the 65" birthday of the Spouse and payable for the subsequent lifetime of the Spouse.

If the payee of the Death Benefit is the designated beneficiary or estate of the employee or former employee, the form of the payment of the Death Benefit will be a lump sum.

D. The amount of the Death Benefit if paid in a lump sum will be the commuted value of the vested deferred pension payable to the former employee or which would be payable to the employee in accordance with Section 5 of Article III if his/her employment had terminated on the date of death, calculated on the basis of the service credit accumulated during the period from 1 January 1987 to the date of death. Service Credit, as applicable to this paragraph, includes credit for any period after 1 January 1987 during which the individual was a Total and Permanent Disability retiree who became disabled prior to 1 January 1992 and whose benefit has not been redetermined, in accordance with Section 3-D of Article III, at the date of his/her death. If the Death Benefit is paid in the form of an immediate or deferred annuity to the surviving Spouse, the monthly annuity payments shall be such that the commuted value of the annuity is equal to the lump sum payment described above.

ARTICLE V SUPPLEMENTAL ALLOWANCE

Section 1. Eligibility for Supplemental Allowance

A. An employee who on 30 April 1987 is eligible to retire and shall subsequently retire before age 65 under Section 2 or Section 3 of Article III may elect to receive a monthly Supplemental Allowance in addition to other retirement benefits under this Plan hereinafter provided in Section 2-A, 2-B or 2-C of this Article V.

- B. An employee who subsequent to 30 April 1987 but prior to September 30, 1998 is eligible to retire and shall retire before age 62 under Section 2 or Section 3 of Article III may elect to receive a monthly Supplemental Allowance in addition to other retirement benefits under this Plan hereinafter provided in Section 2-B or 2-C of this Article V.
- C. An employee who subsequent to September 30,1995 becomes eligible to retire and shall retire before age 60 under Section 2 or Section 3 of Article III may elect to receive a monthly Supplemental Allowance in Addition to other retirement benefits under this Plan hereinafter provided in Section 2-C of this Article V.

Section 2. Amount of Supplemental Allowance

- A Subject to both the provisions of the other Sections of this Article V and the maximum limits imposed by Section 4 of Article VII, the Supplemental Allowance of an employee eligible under Section 1-A of this Article V shall be a monthly amount which when added to benefits payable under Article III of the Plan shall equal a Total Monthly Benefit of \$950.00 for an employee retiring with 30 or more years of service credit on or after 30 April 1987.
- **B.** Subject to both the provisions of the other Sections of this Article V and the maximum limits imposed by Section 4 of Article VII, the Supplemental Allowance of an employee eligible under Section 1-A or 1-B of this Article V shall be a monthly amount which when added to benefits payable under Article III of the Plan shall equal a Total Monthly Benefit for an employee retiring with 30 or more years of service credit as follows:

For Retirement		Total Monthly Benefit
On or After	And Prior To	
30 September 2001	30 September 2002	\$2,400.00
30 September 2002	30 September 2003	\$2,450.00
30 September 2003		\$2,500.00

C. Subject to both the provisions of the other Sections of this Article V and the maximum limits imposed by Section 4 of Article VII, the Supplemental Allowance of an employee eligible under Section 1-A, 1-B or 1-C of this Article V shall be a monthly amount which when added to benefits payable under Article III of the Plan shall equal a Total Monthly Benefit for an employee retiring with 30 or more years of service credit as follows:

For Retirem On or After	ents And Prior To	Total Monthly Benefit
30 September 2007		\$2,500.00

- D. For an employee retiring with less than 30 years of service credit under Paragraph A or B or C of this Section, the Total Monthly Benefit shall be determined by
 - Dividing the applicable Total Monthly Benefit as determined in Paragraph A or B or C of this Section by 30,
 - (2) If the employee is under age 60 at the time of retirement, the amount determined in (1) above shall be reduced by 1% for each month he/she is under age 60, and

(3) The result shall be multiplied by the employee's service credit which for the purposes of benefits provided in Section 3 of Article III shall include time not credited the employee while on Long Term Disability.

Section 3. Assumptions and Adjustments in Computing Amount of Supplemental Allowance

- In the case of an employee who retires at his/her option Α. under Article III, Section 2-A of the Plan, the monthly Supplemental Allowance shall be computed in accordance with the provisions of Section 2 of this Article on the assumption that the retirement benefits under Article III of the Plan would commence immediately after retirement; and the amount computed shall be reduced for any month prior to age 65. in the case of benefits under Section 2-A of this Article or to age 62 in the case of benefits under Section 2-B of this Article or to age 60 in the case of benefits under Section 2-C of this Article, for which the employee becomes or could have become eligible for unreduced statutory benefit. The amount of reduction shall be as follows:
 - (1) For retirements on or after 30 September 2007, \$31.30 per month for each year of service credit, not to exceed \$939.00.
- B. In the case of an employee retiring under disability retirement under Section 3 of Article III of the Plan, the monthly Supplemental Allowance shall be computed in accordance with Section 2 of this Article except that service credit shall be as defined in Section 3-C-(1) of Article III and on the assumption that the monthly benefit under Article III of the Plan includes a temporary benefit provided under Section 3-C-(2) of Article III whether or not it actually includes the temporary benefit.

Section 4. Payment of Allowance

The Supplemental Allowance of an employee entitled to such allowance shall become payable on the first day of the first

month after (1) employment shall have terminated, and (2) he/she shall have filed application for a retirement benefit, and shall be payable on the first day of the month in each month thereafter until and including the first day of the month in which the retired employee attains age 65 in the case of benefits under Section 2-A of this Article to age 62 in the case of benefits under Section 2-B of this Article, or to age 60 in the case of benefits under Section 2-C of this Article. dies. is reemployed by the Company, or the retirement benefits under Article III of the Plan cease for any other reason, whichever shall occur first: provided, however, that if a retired employee entitled to receive a Supplemental Allowance has earnings after retirement in excess of \$18,000 in any calendar year, a penalty equal to one times the amount by which such earnings exceed the amount permitted shall be charged against each succeeding Supplemental Allowance which the employee would otherwise be entitled to receive until the full amount of such penalty is satisfied, it being understood that penalties and charges shall be cumulative if appropriate. Each retired employee receiving a Supplemental Allowance may be required to certify that his/her earnings have not been in excess of the permitted amount, to furnish verification of the amount of such earnings, or to authorize the Company to make inquiries regarding these earnings from the Department of National Revenue, Taxation. The amount of any overpayments of Supplemental Allowance made after a retired employee shall have ceased to be entitled to receive such allowance, in whole or in part because of excess earnings, may be deducted from future monthly retirement benefits payable under the plan, provided, however, that the employee may make repayment in a lump sum amount.

Section 5. Discharged Employees

A discharged employee shall not be eligible to receive the Supplemental Allowance, unless the Company finds, in the case of an employee eligible for regular early retirement under Article III, Section 2-A, that such discharge was for reason which should not result in his/her being ineligible to receive the Supplemental Allowance.

ARTICLE VI RETIRED EMPLOYEES

Section 1. Minimum Benefits

A. The Minimum Pension calculated in accordance with Article 3 Section 1-B -2 payable on or after 01 October 2007, to an employee or to the surviving spouse in the event of survivor benefits retired in accordance with the provisions of the 1955 Plan and the 1955 Plan as amended thereafter, will be continued on the basis of the following amounts per month per year of service credit:

FOR EMPLOYEES RETIRED		
Prior To 31 December 1973		\$16.00
On or After	For Each Year of Service	17.75
31 December 1973	Credit Through 30	
And Prior To	For Each Year of Service	16.00
31 December 1976	Credit in Excess of 30	
On or After 31 December 1976 And Prior To 31 December 1979		19.50
On or After 31 December 1979 And Prior To 31 August 1983		20.00
On or After 31 August 1983 And Prior To 31 October 1989		22.00

On or after 31 October 1989 and prior to 30 September 1992	29.30
On or after 30 September 1992 and prior to 30 September 1995	32.35
On or after 30 September 1995 and prior to 30 September 1998	35.45
On or after 30 September 1998 And prior to 05 June 2000	40.50
On or after 05 June 2000 And prior to 30 September 2004	45.55
On or after 30 September 2004 and prior to 30 September 2007 and who did not retire under the Special Early Retirement Window	45.55

50.05

On or after 30 September 2004 and prior to 30 September 2007 who retired under the Special Early Retirement Window

B. An employee who exercised the election provided in Article III, Section 6-C of the 1974 Plan shall be deemed to have retired under the provisions of the 1971 Plan for the purposes of continuing benefits under this Article.

Section 2. Formula Pensions

A. The Formula Pension benefit payable on or after 1 October 2007 to an employee, or to the surviving spouse in the event of survivor benefits, retired prior to 30 September 2007 will be continued on the same basis as they were payable prior to 1 October 2007.

Section 3. Disability Pensioners

- A. The benefits payable on or after 1 October 2007 to disability pensioners retired prior to 30 September 2007 in accordance with the provisions of the 1955 Plan and the 1955 Plan as amended thereafter will be continued on the basis of the applicable amount shown in the table in Section 1-A of this Article and service credit at the time of disability retirement.
- B. The Additional Temporary Benefit payable on or after 1 October 2007 for disability pensioners retired prior to 30 September 2007 who are currently ineligible for statutory disability benefits will be continued on the basis of the following amount:

		<u> </u>	tional ry Benefit
For Retirements On or After	For Retirements And Prior to	Per Month Per Year of Service Credit	Subject to A Maximum of
	31 Dec 1980	\$14.00	\$350.00
31 Dec 1980	31 Dec 1981	\$15.00	\$375.00
31 Dec 1981	31 Aug 1983	\$16.00	\$400.00
31 Aug 1983	31 Apr 1987	\$16.00	\$480.00
30 Apr 1987	31 Oct 1989	\$18.00	\$540.00
31 Oct 1989	30 Sep 1992	\$23.60	\$708.00
	30 Sep 1995	\$27.25	\$817.50
	30 Sep 1998	\$29.50	\$885.00
	05 Jun 2000	\$30.40	\$912.00
05 Jun 2000	30 Sep 2004	\$31.30	\$939.00
30 Sep 2004	30 Sep 2007	\$35.80	\$1074.00

Section 4. Restoration of Pension Benefits After Death of Retired Employee's Spouse

A. An employee who retired under the provisions of Section 1, 2, 3 or 4 of Article III of the Plan prior to 31 December 1970 and who elected a Joint and Survivor Option at the time of such retirement but whose designated spouse predeceased or shall predecease the retired employee may have the monthly basic pension benefit increased effective the first day of the month following the month in which the Company receives satisfactory evidence of the spouse's death. Such increase will be computed as follows:

- For employees who retired on or after 1 June 1968 and prior to 31 December 1970 - \$0.34 per month per year of service credit;
- (2) For employees who retired prior to 1 June 1968 \$0.31 per month per year of service credit.
- **B.** In the event that a retired employee who is receiving the basic pension benefit increase provided in Paragraph A of this Section makes an election under Section 7 of this Article, such increase shall be discontinued on the effective date of such election. If the designated spouse under the election under Section 7 shall predecease the retired employee, the provisions of Paragraph A shall again be effective.

Section 5. Election of Pension Benefit Increase in the Event of Divorce

In lieu of receiving a reduced amount of any increase in benefits otherwise payable under this Article VI, on or after 1 January 1977, in order to provide an increase in the amount of joint and survivor benefit otherwise payable, an employee who retired prior to 31 December 1970, and who is divorced by court decree from the designated spouse for whom the joint and survivor option is in effect, may elect to receive the full amount of such increase adjusted for early retirement. To make such election the retired employee must complete a form approved by the Company and file such form with the Company, accompanied by evidence satisfactory to the Company of a final decree of divorce, in which case such election shall become effective with respect to benefits falling due for months commencing the first day of the month following the month in which the Company receives such completed election form and final decree of divorce. In the event of such effective election, the benefit payable on 1 July 1982 and thereafter shall be continued

Section 6. Survivor Benefit Election After Retirement

- A. An employee who retired prior to 31 December 1979 under the 1955 Plan or the 1955 Plan as amended thereafter and for whom no survivor option is in effect may elect a Survivor Benefit option by filing a written application with the Company provided:
 - (1) The employee was not married at retirement and has subsequently married, or
 - (2) The employee has had a Survivor Benefit provision in effect and has remarried.
- **B.** An employee who retires on or after **31** December **1979** for whom no Survivor Benefit is in effect may file written application with the Company for a Survivor Benefit provided:
 - (1) The employee was not married at retirement and has subsequently married, or
 - (2) The employee has had a Survivor Benefit provision in effect and has remarried.
- Such Survivor Benefit shall become effective with C. respect to benefits falling due for months commencing with the first day of the month following the month in which the Company receives an application, but in no event before the first day of the month following the month in which the retired employee has been married to the designated spouse for one year. No Survivor Benefit under Paragraph A of this Section shall become effective for a retired employee whose application form is received by the Company after the first day of the month in which the retired employee has been married to the designated spouse for one year. The Survivor Benefit under this Section shall be provided under the terms of the Plan in effect at the time of the employee's retirement
- D. For an employee retiring on or after January 1, 1992, election of the option shall not result in any change to the amount of pension payable to the retired employee.

Section 7. Supplemental Allowance of Retired Employees

- A. The Supplemental Allowance payable to a retired employee on or after 1 October 2007 will be provided as an amount which when added to the benefits payable under Section 1 or Section 2 of this Article, adjusted for early retirement, shall equal the Total Monthly Benefit as follows:
 - (1) For an employee who retired prior to 30 September 2007 with 30 or more years of service credit, the Total Monthly Benefit will be continued as follows:

FOR EMPLOYEES RETIRED	BENEFITS PAYABLE 1 NOVEMBER 1989 AND THEREAFTER
Prior To 31 Dec 1976	\$830.00 Payable to Age 65
On or After	
31 Dec 1976	\$855.00 Payableto
And Prior To 31 Dec 1978	Age 65
On or After 31 Dec 1978 And Prior to 31 Dec 1979	\$890.00 Payableto Age 65
On or After 31 Dec 1979 And Prior To 30 Apr 1987	\$920.00 Payableto Age 65
On or After 30 Apr 1987	\$950.00 Payableto Age 65
if Eligibleto Retire On or Before 30 Apr 1987	(Retired in accordance with Article V Section 1 A)
On ar After 30 Apr 1987 And Prior To 31 Oct 1989	\$1,200.00 Payableto Age 62
if Eligibleto Retire After 30 Apr 1987	(Retired in accordance with Article V Section 1B)
On or after 31 Oct 1989 and Prior to 30 Sep 1992 if Eligibleto Retire After 30 Apr 1987	\$1700.00 Payable to Age 62 (Retired in accordance with Article V Section 1B)

On or after 01 Oct 1992	\$2000.00 Payable to
And Priorto 30 Sep 1995	Age 62
if Eligible to Retire	(Retired in accordance with
After 30 Apr 1987	Article V Section 1B)
·	
and Prior to 30 Sep 1998	Age 62
if Eligible to Retire After	(Retired in accordance with
30 Apr 1987	` Article V Section 1B)
but prior to 30 Sep 1998	
On or after 01 Oct 1995	\$2100.00 Payableto
And prior to 30 Sep 1998	Age 60
if Eligibleto Retire after	(Retired in accordance with
30 Sep 1995	Article V Section 1C)
On or after 01 Oct 1998	\$2300.00 Payable to
And prior to 05 Jun 2000	Age 62
■ If Eligible to Retireafter	(Retired in accordance with
30 Apr 1987 but prior to	Article V
30 Sep 1998	Section 1B)
On or after 01 Oct 1998	\$2300.00 Payableto
And prior to 05 Jun 2000	Age 60
If Eligible to Retire after	(Retired in accordance with
30 Sep 1995	Article V
On an offen OF I.m. 8000	POEOO OO Doyoble to
On or after 05 Jun 2000	\$2500.00 Payable to Age 62
and prior to 30 Sep 2004	Age 62 (Retired in accordance with
If Eligibleto Retireafter	Article V
30 Apr 1987 but prior to	I Alucie v
On or after 05 Jun 2000	\$2500.00 Payable to
and prior to 30 Sep 2004	Age 60
If Eligibleto Retireafter	(Retired in accordance with
30 Sep 1995	Article V Section 1C)
On or after 01 Oct 2004 and	\$2500.00 Payable to
	Age 62
	(Retired in accordance with
	` Article V
	Section 1B)

On or after 01 Oct 2004 and	\$2500.00 Payableto
prior to 30 Sep 2007	Age 60
who did not retire under the	(Retired in accordance with
Special Early Retirement	Article V
Window If Eligible to Retire	Section 1C)
after	•
30 Sep 1995	
On or after 01 Oct 2004 and	\$2650.00 Payableto
prior to 30 Sep 2007	Age 62
who retired under the Special	(Retired in accordance with
Early Retirement Window	Article V Section 1B)
If Eligibleto Retireafter	
30 Apr 1987 but prior to	
30 Sep 1998	
On or after 01 Oct 2004 and	\$2650.00 Payableto
prior to 30 Sep 2007	Age 60
who retired under the Special	(Retired in accordance with
Early Retirement Window If	Article V Section 1C)
Eligible to Retire after	
30 Sep 1995	

- (2) For an employee who retired on or after 31 December 1973 with less than 30 years of service credit, the Total Monthly Benefit shall be determined by:
 - Dividing the applicable Total Monthly Benefit as determined in A-(1) or (2)of this Section by 30,
 - b. If the employee was under age 60 at the time of retirement, the amount determined in (2)-a shall be reduced by 1% for each month he/she was under age 60, and
 - c. The result shall be multiplied by the employee's service credit.

ARTICLE VII MAXIMUM BENEFITS PAYABLE UNDER THE PLAN

Section 1. General

Benefits payable under the Plan are, at all times, subject to the constraints and limitations imposed on it by Applicable Legislation. In particular, the Income Tax Act and Regulations restrict the amount that any individual may receive from a registered pension plan by way of benefits. This article sets out the maximum benefits which may be paid from the Plan.

Section 2. Definitions

For the purposes of the Article, the following definitions will apply. Terms not defined in this Section but appearing in the body of this Article should assume the meanings prescribed to them in previous Articles of this Plan Text:

"Bridging Benefits" are retirement benefits provided to a Member under the Plan, for a period ending no later than a date determined at the time benefits commence to be paid, and provided under Article III, Section 2-E, Section 3C(2) and Article V of this Plan."

"Compensation" means the total of all straight-time wage payments plus vacation pay, personal absence pay and pay for unworked holidays received during an anniversary year. For periods of unpaid service credit, Compensation shall be assumed to be not less than the rate of Compensation in effect immediately prior to commencement of the unpaid period.

"Defined Benefit Limit" for a calendar year means \$1,722.22 for years before 2005 and, for years after 2004, one-ninth of the Money Purchase Limit for the year.

"Highest Average Compensation" is, the average of the highest 3 consecutive anniversary years Compensation paid to the employee by the Company during the 10 anniversary years immediately prior to the earlier of retirement or attainment of age 69.

"Lifetime Retirement Benefits" provided to a Member of the Plan are retirement benefits which, are provided under Sections 1B, 2B, 3C(1), 5B and 5C of Article III of this Plan Text, and once they commence to be paid, are payable to the Member until the Members death, unless the benefits are commuted or payment of benefits is suspended.

"Member" is an employee of the Company who is eligible to receive benefits provided under the Plan.

"Money Purchase Limit" for a calendar year is defined by Section 147.1 (1) of the Income Tax Act.

"Pensionable Service" of the Member of the Plan means the sum of his/her Past Credited Service up to 31 December 1991, not to exceed 35 years, plus his/her Future Credited Service, on or after 1 January 1992, as determined under Article II. Section 3.

"Public Pension Benefits" means amounts payable on a periodic basis under the Canada Pension Plan, a provincial plan as defined in Section 3 of the Canada Pension Plan, or Part I of the Old Age Security Act, but does not include disability, death or survivor benefits provided thereunder.

"Year of Commencement" is the calendar year in which the member's retirement benefits commence to be paid.

Section 3. Lifetime Retirement Benefits

- A The amount (expressed on an annualized basis) of Lifetime Retirement Benefits payable to the Member must not exceed the amount determined by the sum of:
 - (i) FxG

where

- F is the lesser of
 - (a) 2% of the Members Highest Average Compensation, and
 - (b) The Defined Benefit Limit for the Year of Commencement, and
- G is the Pensionable Service of the Member after 31 December 1991 and:

- (ii) the lesser of 1 or 2 as follows:
 - 2% of the average of the highest 3 consecutive years' remuneration paid to the employee by the Company multiplied by the number of years of his/her pensionable service, subject to a maximum of 35 years, prior to 1 January 1992;
 - the Defined Benefit Limit multiplied by the number of years of pensionable service, subject to a maximum of 35 years, prior to 1 January 1992.
- B. If Lifetime Retirement Benefits commence to be paid to a Member before age 60, then the amount (expressed on an annualized basis) of Lifetime Retirement Benefits payable for each calendar year may not exceed the sum of:
 - (i) the amount determined by the formula:

where

- X is the amount (expressed on an annualized basis) of Lifetime Retirement Benefits payable to the member for the year if the benefits were determined without a reduction computed by reference to the member's age, duration of service, or both, and without any similar reduction but based on pensionable service after 31 December 1991.
- Y is the number of months in the period from the particular day on which lifetime benefits commence to be paid to the Member to the earliest:
 - (A) the day on which the Member attains 60 years of age,
 - (B) the day on which the Member has 30 years of Service Credit, (calculated as if the Member had continued in employment with the company)
 - (C) the day on which the aggregate of the Member's age (measured in years including any fraction of a year) and years of Service

(calculated as if the Member had continued in employment with the company) is equal to 80 and

(D) the day on which the member became Totally and Permanently Disabled.

and

(ii) the amount of pension which, is Actuarially Equivalent to the unreduced pension calculated in accordance with A (ii) above if payable from the earlier of age 60 and age at date of disability as a single life annuity with a guaranteed period of payment of ten years or a 60% joint and last survivor annuity, which ever is appropriate.

Section 4. Bridging Benefits

The amount of the bridging Benefits may not exceed the greatest of:

A. The amount determined by the formula:

where

- ▲ is the amount (or a reasonable estimate thereof) of Public Pension Benefits that would be payable to the Member for the month in which the Bridging Benefits commence to be paid to the Member if
 - (a) the member were 65 years of age throughout that month.
 - (b) that month were the first month for which Public Pension Benefits were payable to the Member.
 - (c) the Member were entitled to the maximum amount of benefits payable under the Old Age Security Act, and
 - (d) the Member were entitled to that proportion, not exceeding one, of the maximum benefits payable under the Canada Pension Plan (or a provincial plan as defined in section 3 of the Canada Pension Plan) that the total of the Member's remuneration for the three calendar

years in which the remuneration is the highest is of the total of the Year's Maximum Pensionable Earnings for those three years (or such Pensionable Earnings as is acceptable to the Minister),

B is.

- (a) except where clause (b) is applicable, the number of months, if any, from the date that the Bridging Benefits commence to be paid to the Member to the date that the member attains 60 years of age, and
- (b) where the Member is totally and permanently disabled at the time at which the Bridging Benefits commence to be paid to the Member, nil,

🚅 is

- (a) except where clause (b) is applicable, the lesser of 10 and the aggregate of all amounts each of which is the duration (measured in years, including any fraction of a year) of a period that is Service Credit of the Member under the Plan, and
- (b) where the member is Totally and Permanently Disabled at the time at which the Bridging Benefits commence to be paid to the member, 10,

the amount determined by the formula:

$$[(D \times E/F) + (G \times (1 - E/F))]$$

where:

- **D** is the amount determined for A under Section **4(A)** of this Article based on the terms and conditions of the Plan on January **1**, 1992.
- **E** is the Member's Past Credited Service.
- **E** is the Member's Service Credit
- A is the amount determined under Section 4(A) of this
 Article but limited to the amount of Bridging Benefits
 payable in accordance with the terms and

conditions of the Plan as they existed on December 31, 1991.

C. Bridging Benefits vested in a member as of December 31, 1991.

Section 5. Retirement Benefits Before Age 65

From the time at which the benefits commence to be paid to the time at which the member attains 65 years of age, hereinafter referred to as the "Bridging Period, the amount (expressed on an annualized basis) of retirement benefits payable to the member for the Bridging Period in respect of service after 31 December 1991 must not exceed the amount determined by the formula:

$$(A \times B) + (0.25 \times C \times D/35)$$

where

- is the Defined Benefit Limit for the Year of Commencement.
- **B** is the aggregate of all amounts each of which is the duration (measured in years, including any fraction of a year) of a period that is Pensionable Service of the member, subsequent to 31 December 1991.
- is the average of the Year's Maximum Pensionable Earnings for the Year of Commencement and for each of the two immediately preceding years, and
- **D** is the lesser of 35 and the amount determined for B.

Section 6. Benefit Accrual Rate

The benefit accrual rate or the equivalent benefit accrual rate, **as** the case may be, must not exceed 2% per annum of the Members' Compensation.

Section 7. Service Credit

Periods treated as time worked for Future Credited Service purposes in accordance with Sections 9 and 10 of Article II shall be limited for each employee to 5 years with an additional 3 years in respect of periods of parenting defined

as a period within 12 months of the birth or adoption of a child.

APPENDIX "B" THE HEALTH BENEFIT PLAN WAGE EMPLOYEES

ARTICLE I GENERAL PROVISIONS

Section 1. Preamble

John Deere Welland Works of John Deere Limited, hereinafter referred to as the Company, will provide the Health Benefit Plan for Wage Employees, hereinafter referred to as the Plan. The Plan shall make available coverage for semi-private hospital accommodation expense, supplementary and catastrophic medical expense, prescription drug expense, dental expense, vision care expense and hearing aid benefit expense.

Section 2. Effective Dates of Coverage

- A. Employees shall have John Deere Supplemental Health coverage effective the first day of the month following attainment of seniority, unless otherwise stated in this Collective Bargaining Agreement.
- B. Employees shall notify the Company within thirty-one (31) days of the date their dependency status changes (unless prevented from doing so because of reasons satisfactory to the Company) between "employees with no dependents" and "employees with one or more dependents" in order to be properly covered and classified. Upon such proper notification, coverage shall be deemed effective from the date the employee acquired the dependent.
- C. Evidence of good health is required, except for dental expense coverage, if an employee makes a Late Application for coverage on any person. An application is considered late when an employee:
 - applies for coverage on any eligible person(s) more than 31 days after having been eligible, or
 - (2) reapplies for coverage on any eligible person(s) whose coverage had earlier been cancelled.

When an employee applies for coverage after having previously waived benefits under this Plan because the employee was covered for similar benefits under the employee's Spouse's plan, an application is considered late:

- if application for coverage is more than 31 days after benefits terminated under the Spouse's plan, or
- (2) if application for coverage, and benefits under the Spouse's plan have not terminated.

Section 3. Termination

- A. Employee coverage referred to in Section 1 of this Article will terminate when the employee's employment terminates subject to the provision of this Plan.
- B. Coverage on any dependent will cease automatically:
 - The date the dependent becomes covered as an employee of the Company, or
 - (2) if the dependent is a spouse, on the date of divorce, or
 - (3) if the dependent is a child when any such child fails to meet the definition of dependents as set forth in Section 9 of Article I or,
 - (4) on the date of termination of employee coverage.

Section 4. Cost of Benefits

A The Company will pay the benefit amounts for all covered expenses once any applicable deductibles have been satisfied. Payments are subject to any maximum amounts identified within the Health Benefit Plan. Lifetime maximums apply to all periods combined in which an individual is covered under this plan. Claim amounts that will be applied to the maximum are the amounts paid for covered expenses after applying deductibles and co-payment percentages, where applicable. Except as otherwise specifically provided, the cost of providing benefits under this Plan will be

borne by the Company and no contribution to the Plan shall be made by any employee, retired employee or beneficiary.

B. Expenses are covered if they:

- Are medically necessary for the treatment of an illness or injury and are recommended by a Physician; and
- Are incurred for the care of a person while he/she is covered under this plan: and
- 3) Are reasonable taking all factors into account.
- Are reasonable and customary, as determined by the Administrative Service Provider; and
- Are not covered under the Provincial plan or any government-sponsored program;

Section 5. Amendment, Modification, and Plan Termination

Except as otherwise specifically provided, John Deere Limited may at any time amend, modify, or terminate the Plan, provided however, that no change shall reduce the amount of any benefit to which an employee, retired employee, or beneficiary shall be entitled in respect to claims incurred prior to the effective date of such change.

Section 6. Funding

Except as otherwise specifically provided, benefits shall be provided through an insurance company selected by John Deere Limited, a fund established by John Deere Limited, or from the general assets of John Deere Limited.

Section 7. Claim for Benefits

In order for any employee, retired employee or beneficiary to receive benefits pursuant to any provisions of this Plan, such employee, retired employee or beneficiary must file a written claim within twelve (12) months from the date the expense was incurred, with exceptions noted below. The necessary form(s) for such purpose will be supplied by the Company.

Upon termination of a person's benefits under this plan, submission of the claim must be within the earlier of twelve (12) months from the date the expense was incurred and 90 days from the termination of plan benefit coverage.

Section 8. Denial of Benefits

When a claim for benefits is denied in full or in part, the claimant shall receive a written notice of the reason or reasons for such denial.

Section 9. Definitions

The term "Dependent" as used herein is limited to:

- A. The employee's spouse, as designated by the employee, who is residing in Canada, and will only include one person, defined as follows;
 - (i) the employee's legal spouse, or
 - (ii) the employee's common law spouse as defined in Appendix G, Article II, Section 13.
- B. The employee's unmarried natural or adopted children, or step-children of the employee or the employee's spouse (provided that the spouse's step-child(ren) reside with and are in the legal guardianship of the employee) or those for whom the employee has legal guardianship, or the employee has legal responsibility for health care, or legal adoption proceedings have been initiated who are
 - under nineteen years of age, or nineteen years of age but under twenty-six years of age if a full-time student, and
 - (ii) residing in Canada, and
 - (iii) not employed on a full-time basis, and
 - (iv) not eligible for coverage under this or any other group benefits plan and

- dependent on the employee for more than one-half of their support
- C. A child who is incapacitated on the date when coverage would normally terminate. Any such child will continue to be eligible as a dependent provided the child was covered under this Plan immediately prior to that date. A child is considered incapacitated if the child is incapable of engaging in any substantially gainful activity and is dependent on the employee for support, maintenance and care due to a mental or physical handicap. Proof of the child's condition may be required as often as reasonably necessary.

ARTICLE II CONTINUATION OF COVERAGE UNDER THIS PLAN

Section 1. Leave of Absence

- A. An employee not actively at work due to leave of absence except maternity or parental leave of absence (See Section 3), may continue coverage under this Plan for a maximum of twelve (12) months on the following basis:
 - (1) During the first (6) months of such leave, the employee will pay fifty (50) percent of the premium and the Company will pay the remaining fifty (50) percent of the premium. During the subsequent six (6) months of the leave, the employee will be required to pay the full premium.
 - (2) Employees not actively at work due to leave of absence for Local Union business may continue coverage under this Plan beyond twelve (12) months for the duration of the leave by paying the full premium by the first of each month.
- **B.** If an employee is granted a leave of absence due to a clinically anticipated disability based on the natural course of the employee's diagnosed condition and if such employee continues coverage under this Plan during such approved leave of absence as provided in Paragraph A-(1) of this Section, upon medical

certification satisfactory to the Company from the employees attending physician that the employee is totally disabled, coverage for the employee will be continued under this Plan as provided in Section 3 of this Article

Section 2. Layoff

- An employee with seniority but not actively at work due to layoff will have coverage, subject to the Nonduplication of Benefits provision of Section 2 of Article VIII of this Plan, continued under this Plan for a period of time which will be determined by the number of credit units credited to the employee at the date of layoff, in accordance with the provisions of the Supplemental Unemployment Benefit Plan, but in no event will such period be less than six (6) months or exceed twelve (12) months from the date of layoff. One month's coverage will be provided for each four (4) credit units the employee is credited with as of the date of layoff but not to exceed twelve (12) months.
- **B.** Following the expiration of the maximum number of months for which the coverage under this Plan was continued under (A) above, the employee may continue coverage for a period of twelve (12) additional months by payment of the full premium without interruption.
- C. Notice will be given to employees at least sixty (60) days prior to termination of their coverage under this Plan notifying them of the date coverage will terminate as provided in Paragraph A above and of the provisions for continuation of coverage as provided in Paragraph B above.

Section 3. Illness, Accident, Maternity or Parental Leave of Absence

The coverage of an employee not actively at work because of illness, accident, maternity or parental leave of absence will be continued under this Plan at no cost to the employee during such illness, accident or leave of absence or for a period of time equal to the employee's continuous

employment, whichever is lesser, but in any event not less than fifty-two (52) weeks.

An employee or a retired employee will have coverage continued while eligible to receive:

- (i) Long-Term Disability Benefits as set forth in the John Deere Disability Benefit Plan for Wage Employees, or
- (ii) Disability Retirement Benefits as set forth in the John Deere Pension Plan for Wage Employees.

Section 4. Employee Receiving Life Insurance in Monthly Installments

An employee receiving Life Insurance in monthly installments under the John Deere Group Life and Disability Insurance Plan for Wage Employees will have the benefits outlined in this Plan continued by the Company without cost to the employee. While such benefits are in effect for such employee, they will be continued for the employee's eligible dependents.

Section 5. Retirement

An employee retiring under the John Deere Pension Plan for Wage Employees will have the benefits outlined in this Plan continued by the Company without cost to the employee. While such benefits are in effect for the retired employee, they will be continued for the employee's dependents without cost.

Section 6. Surviving Spouse

- A. The surviving spouse of an employee not eligible to retire under the John Deere Pension Plan for Wage Employees at the time of death and who is receiving a Transition Survivor Income Benefit and not eligible for a Bridge Survivor Income Benefit will have coverage outlined in this Plan continued without cost while eligible to receive such Transition Survivor Income Benefits.
- **B.** The surviving spouse of an employee not eligible to retire under the John Deere Pension Plan for Wage Employees at the time of death and who is eligible to

receive both a Transition and Bridge Survivor Income Benefit will have coverage outlined in this Plan continued without cost while eligible to receive such Transition and Bridge Survivor Income Benefits.

- C. The surviving spouse of a retired employee or the surviving spouse of an employee who was eligible to retire under the John Deere Pension Plan for Wage Employees at the time of death will have the benefits outlined in this Plan continued by the Company without cost provided such spouse was covered under the Health Benefit Plan. While such benefits are in effect for the surviving spouse, they will be continued for such spouse's dependent children who were eligible as dependents at the time of death.
- D. If an employee dies as a result of a work incurred accident or illness, the surviving spouse not eligible under Paragraph C above, will have the benefits outlined in this Plan continued by the Company without cost. Such benefits will cease on the first to occur of the surviving spouse's remarriage, or death.
- E. If an employee dies as a result of a work incurred accident or illness and there is no surviving spouse, the legal guardian of the employee's eligible dependent children may continue the benefits outlined in this Plan for such dependents by payment of the full monthly premium.
- **F.** For the purpose of A, B, C, and D above, coverage will apply to the surviving spouse and the employee's or retiree's eligible dependents as defined in Article I, Section 9. No benefit will be payable with respect to a pregnancy commencing after the death of the employee.

ARTICLE III

Section 1 Benefit Contribution Rates

When an employee is required to make benefit contribution payments to continue coverage, such payments must be made prior to the first of the month for which it is paid.

ARTICLE IV MEDICAL, SURGICAL AND HOSPITAL EXPENSE BENEFITS UNDER ONTARIO HEALTH INSURANCE PLAN (OHIP)

Section I. O.H.I.P.

Ontario Health Insurance is a comprehensive Governmentsponsored plan of health insurance for residents of Ontario. It provides a wide scope of benefits for medical and hospital services, plus additional benefits for the services of certain other health practitioners. Any portion of a medical or surgical bill covered by OHIP is not covered under any John Deere sponsored plan. John Deere Supplemental Insurance Plan will insure the difference, but only up to and including the amount shown in other sections of the John Deere Limited Health And Accident Insurance Plan.

ARTICLE V SUPPLEMENTAL HOSPITAL EXPENSE BENEFITS EMPLOYEES AND THEIR DEPENDENTS

Section 1. For Daily Room and Board Expense

- A. Benefits will be paid to an employee if a covered employee or a covered employee's dependent shall be confined upon the recommendation of a physician as a resident patient in a hospital on account of accidental bodily injury or sickness for an amount equal to the difference between the regular direct cost for standard ward and semi-private accommodation.
 - Charges will be paid by the plan for up to 365 days for each continuous period of disability. Successive periods of confinement for the same or related causes will be considered one period unless they are separated by a return to work (or by 60 days in the case of a retiree).
- **B.** Benefits will not be paid for any confinement following the date the attending physician or the employee or the employee's dependent determine that such confinement is no longer necessary.

C. No payment shall be made under this section for any confinement in a chronic care bed or unit of a hospital or in any other bed or unit by whatever name called where patients do not require active medical treatment for the acute phase of illness or disability.

Section 2. Chronic Care

A covered employee or a covered employee's dependent eligible for benefits under Section 1-A of this Article because of confinement in a hospital on account of accidental bodily injury or disease for a period of five (5) days or more and transferred on orders of a physician within fourteen (14) days of release from such hospital on account of the same bodily injury or sickness to a chronic care bed of a hospital or any other bed by whatever name called where patients do not require active medical treatment but do require continued medical supervision will be eligible for benefits under this The amount payable for any day of such confinement where not eligible for benefit under Section 1 of this Article will be equal to the difference between the regular direct cost for standard ward and semiprivate accommodation, up to a maximum of \$10,000 per confinement.

Section 3. Exclusions and Limitations

- A. No payment shall be made under Section 1 for:
 - non-emergency hospital care outside the person's province of residence,
 - (ii) a period of confinement to a hospital outside of Canada.
- **B.** No payment shall be made under Section 2 for any day of confinement for which the employee or employee's dependent is not eligible for services under a Hospital Insurance Act or a provincial hospital plan.

ARTICLE VI ADDITIONAL TREATMENT EXPENSE BENEFITS EMPLOYEES AND THEIR DEPENDENTS

Section 1. Prosthetic Device Benefits

A. Eligibility

Prosthetic device benefits will be payable if a prosthetic device is received on or after 1 October 1975 by a covered employee or a covered employee's dependent as a result of accidental bodily injury or sickness on the order of a physician when payment for such device is not otherwise covered under the Plan. Payment will be made for the actual amount charged for such device up to the reasonable and customary charge. Payment may be made directly to the provider or supplier of such device.

B. Covered Expenses

- (1) Charges for artificial eyes, limbs and breast prostheses (includes surgical brassieres up to two (2) per calendar year).
- (2) Charges for external prostheses, braces (other than foot braces), trusses, collars, leg orthosis, casts and splints.
- (3) Charges up to \$150 per covered person for the following expenses when recommended by a physician or podiatrist:
 - i) stock-item orthopaedic shoes, subject to a maximum of one pair per calendar year;
 - modifications or adjustments to stock-item orthopaedic shoes or regular footwear terminal devices and external prostheses.
- (4) Charges for 1 pair of custom-made orthopaedic shoes per calendar year which are:
 - iii) constructed by a Certified Orthopaedic Footwear Specialist; and

- iv) required because of a medical abnormality that, based on medical evidence, cannot be accommodated in a stock-item orthopaedic shoe or a modified stock-item orthopaedic shoe.
- (5) Charges for casted, custom-made orthotics which are recommended by a physician or podiatrist, up to 1 pair/3 yrs for adults and 1 pair/yr for children 16 yrs of age and under, to an annual maximum of \$400 per covered person.
- (6) Post-surgical lenses customarily used during convalescence from eye surgery. In addition, combinations of prosthetic lenses are covered when determined to be medically necessary by a physician to restore essentially the vision provided by the crystalline lens of the eye.
- (7) Charges for ileostomy, colostomy and incontinence supplies.
- (8) The cost of oxygen.
- (9) Charges up to \$250.00 per covered person per lifetime for wigs where required for hair loss resulting from chemotherapy or other similar medical treatment.
- (10) Charges for surgical stockings (such as Jobst or equivalent), up to a maximum of two (2) pair per calendar year.

Replacements of unusable prosthetic devices or repairs of these devices when furnished on a physician's order, and supplies and equipment not having any use other than in connection with the use of the prosthetic device and which are necessary for the effective use of the prosthetic device will also be covered.

C. Exclusions

Dentures, other dental appliances, hearing aids and glasses and contact lenses prescribed to correct visual defects are excluded. Also excluded are nondurable

items such as garments, tongue depressors, swabs and elastic support bandages.

Section 2. Durable Medical Equipment

A. Eligibility

Durable Medical Equipment Benefits will be payable if durable medical equipment is received on or after 1 October 1975 by a covered employee or a covered employee's dependent as a result of accidental bodily injury or sickness on the order of a physician for use, when not confined as an inpatient in a hospital, nursing home, or any other institution for the treatment of such accidental bodily injury or sickness, or to improve the functioning of a malformed body member when payment for such equipment is not otherwise provided for under the Plan

B. Payment

Payment will be made for the actual amount charged for the rental of such equipment up to the reasonable and customary charge. Payment may **be** made directly to the provider or supplier **of** the equipment. The company may approve purchase of such equipment if it can reasonably be assumed that the duration of need is such that the rental price would exceed the purchase price, or said items cannot be made available on a rental basis.

C. Covered Expenses

- (1) Manual hospital beds, respiratory and oxygen equipment and other durable medical equipment normally found only in hospitals;
- (2) Wheelchairs, crutches, canes, walkers;
- (3) Inhalators, traction equipment, nebulizers and suction machines, toilet aids, circulatory aids and neuromuscular stimulators, and glucose monitors.

Benefits will not be paid for special features or equipment such as motor drive beds and wheelchairs requested by the patient for personal comfort or convenience unless medically necessary.

D. Exclusions

"Durable Medical Equipment" does not include dentures; hearing aids: eyeglasses; contact lenses or equipment which is primarily and customarily used for nonmedical purposes such as heat lamps: air conditioners and other devices and equipment used for environmental control or to enhance the environmental setting in which the patient is placed such as room heaters, humidifiers, dehumidifiers and other equipment which basically serve comfort or convenience; special pad or mattress to prevent decubitus ulcers, (except in case of advanced neurological disorders) and bed bath types of equipment which basically are utilized for hygienic purposes; prosthetic devices; any other item or device which does not stand repeated use such as , face mask, irrigating kits, ace bandages, in the care and treatment of the patient.

Section 3. Ambulance

Benefits will be paid to an employee on behalf of a covered employee or a covered employee's dependent for payment of any portion of the charge for licensed ambulance service provided in the employee's province of residence, including air ambulance service to or from the nearest hospital where adequate treatment is available.

Section 4. Licensed, Certified Practitioner

Benefits will be paid to a covered employee for charges made by the following Licensed, Certified Practitioners for necessary treatment of himself or a covered dependent on account of accidental bodily injury or sickness, or examination to determine if such a condition exists, equal to the actual expense to the employee, up to the maximum amounts described below:

- (1) Naturopath, Chiropractor, Osteopath, Podiatrist and Acupuncturist: \$15.00 for each visit with a maximum of \$450.00 per person per calendar year.
- (2) Reasonable and customary charges for services of a licensed Psychologist, Speech Therapist,

Physiotherapist and Massage Therapist shall be payable up to a maximum of \$450.00 per person per calendar year per practitioner.

Section 5 Special Duty Nursing

- A. Benefits will be paid to an employee if a covered employee or a covered employee's dependent shall receive care and treatment within the practice of nursing that is performed in the employee's home by a graduate Registered Nurse (R.N.) or a Registered Nursing Assistant (R.N.A.) (or equivalent designation), who has completed an approved medications training program, where treatment is determined medically necessary and is ordered by the attending physician.
- **B.** Benefits provided under this Section will be subject to the following conditions:
 - (1) Benefits shall be paid in an amount not to exceed 80% of the first \$10,000 of the reasonable and customary charge for such care and treatment, and 100% of the reasonable and customary charge for such care and treatment in excess of \$10,000. This benefit is subject to a \$10,000 maximum per person per calendar year and to a maximum total benefit under this Section of \$70,000 for any person for his or her lifetime.
- **C.** Charges for the following services are not covered:
 - Service provided primarily for custodial care, homemaking duties or supervision;
 - (2) Service performed by a nursing practitioner who is an immediate family member or lives with the employee or the employee's dependent;
 - (3) Service performed while confined in a hospital, a nursing home, or any similar institution; and
 - (4) Service which can be performed by a person of lesser qualification, a relative, friend or a member of the household.

Section 6. Health Coverage outside province of residence/Canada

- A. No payment shall be made under this plan for nonemergency hospital care outside the person's province of residence.
- **B.** No payment shall be made under this plan once a medical emergency no longer exists, per item **D** below, and the person can be safely moved to a Hospital in the person's place of residence.
- C. No payment shall be made under this plan for any day for which the employee or employee's dependent is not eligible for services under a Hospital Insurance Act or a provincial hospital plan.
- D. Charges incurred for treatment required as a result of a Medical Emergency arising while temporarily outside the province of residence are covered under this plan, provided that the covered person who receives the treatment is also covered by the Provincial Plan during from the province of residence. absence A Medical Emergency is a sudden, unexpected injury which occurs or an unforeseen illness which begins while a covered person is traveling outside his/her province of residence and requires immediate medical attention. Such emergency no longer exists when, in the opinion of the attending physician and supporting medical evidence, the covered person is able to return to his/her province of residence.
- E. Eligible expenses under this Plan for treatment outside the province of residence will be limited to a maximum of \$1,000,000 per person and will cover the following expenses:
 - (1) Physician services;
 - (2) Charges for the difference between ward and semiprivate room and board rates up to the maximums stated in this Plan had the person been confined to a hospital in the person's province of residence;
 - (3) The cost of hospital services; and

- (4) Hospital charges for out-patient treatment.
- (5) Licensed ambulance services, includiring air ambulance, to transfer the person to the nearest medical facility or Hospital where adequate treatment is available; and
- (6) Medical evacuation for admission to a Hospital or medical facility in the province where the person normally resides.
- F. Covered Expenses will be limited to reasonable and customary charges less the amount payable by the Provincial Plan, or which would have been payable had proper application been made. All other charges incurred while outside the province of residence are payable under the appropriate Covered Expense on the same basis as if they were incurred in the province of residence.
- G. The plan provides Travel Assistance coverage to covered employees and covered dependents as outlined in the ManuAssist explanatory brochure in effect at the time of the claim.

Section 7. Diagnostic Services

Charges for microscopic & laboratory tests & other similar diagnostic services will be covered.

ARTICLE VII CATASTROPHIC MEDICAL EXPENSE BENEFITS

Section 1. Eligibility

A. Benefits will be paid to an employee if a covered employee or a covered employee's dependent incurs Catastrophic Medical Expense as hereinafter provided.

Section 2. Effective Date

A. The benefits provided in this Article shall be effective for disabilities commencing on or after 6 June 1983.

Section 3. Covered Catastrophic Medical Expense

- A. Subject to the limitations and provisions hereinafter contained, if a covered employee or a covered employee's dependent shall be necessarily confined as a resident patient of a hospital after exhaustion of the benefits under Article V, Section 1-A of this Plan, benefits will be paid to the employee for the following necessary health care services in respect to such continuing disability on account of accidental bodily injury or sickness:
 - In-patient hospital care beyond 365 days for an illness or injury;
 - Necessary special duty nursing services, rendered outside of hospital, by a person as described in Article VI, Sections 5(A) and 5 (C).
- B. If a covered employee or a covered employee's dependent necessarily receives special duty nursing services by a person as described in Article VI, Sections 5(A) and 5(C), subject to the limitations and provisions hereinafter contained, benefits will be paid to the employee for charges for such services to the extent that such charges are reasonable and customary for the area and type of service, are not rendered by a member of the employee's family, and the first \$100 of such charges in a calendar year are paid by the employee.

Section 4. Indemnity Limits

- A. Payment for charges for health care services described in Section 3-(A)-1 of this Article for in-patient hospital care beyond 365 days shall be 80% of the first \$10,000 of such charges in respect of any one (1) disability, and 100% of such charges in excess of \$10,000, subject to a \$10,000 maximum per person per calendar year and total benefit under this Article of \$70,000 for any person for his/her lifetime.
- B. Total payment for charges for health care services described in Article VI, Section 5 and Section 3-(A)-2 of

this Article for special duty nursing shall be 80% of the first \$10,000 of such charges in respect of any one (1) disability, and 100% of such charges in excess of \$10,000, subject to a \$10,000 maximum per person per calendar year and total benefit under this Article of \$70,000 for any person for his/her lifetime.

Section 5. Exclusions and Limitations

A. No payment shall be made under this Article in respect to charges for services for which benefits are otherwise provided under any health benefit program provided through the Company to employees and their dependents.

ARTICLE VIII EXCLUSIONS AND LIMITATIONS APPLICABLE TO ALL ARTICLES OF THE PLAN

Section 1. Exclusions and Limitations

- **A.** No payment shall be made in any event with respect to:
 - (1) Charges incurred for which benefits are payable in accordance with the provisions of any Workplace Safety and Insurance law.
 - (2) Charges incurred for which benefits are provided under any Governmental programs.
 - (3) Charges which would not have been made if no coverage had existed or for charges the employee is not legally required to pay.
 - (4) Charges related to war, insurrection, the hostile action of any armed forces or participation in a riot or civil commotions.
 - (5) Charges related to the conviction of a criminal offence (excluding impaired driving or attempted assault).
 - (6) Charges for periodic check-ups, broken appointments, third party examinations, travel for health purposes or completion of claim forms.

- (7) Charges for services or supplies received from a medical or dental department maintained by an employer, association or trade union.
- (8) Charges for services or supplies which are required only for recreation or sports.
- B. Reimbursement for eligible expenses will be limited to the reasonable and customary charges for such expense or to the cost of any medical equipment or device that adequately meets the covered person's fundamental medical needs.

Section 2. Nonduplication of Benefits

- A. This provision shall apply to an employee whose coverage is continued during a period of layoff provided he/she is entitled to benefits as a covered employee of another employer.
- B. If any benefit shall be provided under any other group benefit policy, or any other group plan by whatever name called, on account of hospital, surgical and medical expenses covered under this Plan and in connection with any injury, sickness or pregnancy an amount equal to the sum of (1) the total benefits provided through such policy or plan and (2) the total cash value computed on an equitable basis of all services and supplies furnished through such policy or plan under provisions thereof which provide for the furnishing of services and supplies rather than payment in cash shall be deducted from the amount which otherwise would be payable under this Plan on account of such injury, sickness or pregnancy.

Section 3. Coordination of Benefits

A. Coordination

Benefits for covered expenses payable under this Plan shall be coordinated as set out below with the following "other group plans":

- (1) other employment-relatedgroup benefit plans;
- any other arrangement of coverage for individuals in a group, whether on an insured or uninsured basis,

including any pre-payment coverage, capitation plan, franchise pian or services plan;

(3) individual travel insurance plans (which are coordinated in accordance with the guidelines provided by the Canadian Life and Health Insurance Association).

The term "other group plans" does not include school insurance.

B. How Claims are Coordinated

Benefits payable under this Plan will be reduced so that no more than 100% of eligible expenses are jointly paid by this Plan and all other plans which come before this Plan in the Order of Benefit Payment.

For the purpose of this section, eligible expenses are as defined in each plan document, before any applicable payment limitations, such as deductibles, coinsurance and maximums, are applied. An expense is eligible only to the extent that it is reasonable and customary.

C. Order of Benefit Payment

The following rules establish the Order of Benefit Payment. These rules are applied from first to last until an order is established.

- (1) The other group plan that does not contain a Coordination of Benefits provision shall be considered primary.
- (2) If all the other group plans contain a Coordination of Benefits provision:
 - a. The plan of the employer of the covered person on whose behalf the expenses were incurred shall be primary. If the person is an employee under more than one plan, the following order applies:
 - 1. the plan where the person is an active full-time employee;
 - 2. the plan where the person is an active parttime employee;
 - 3. the plan where the person is a retiree.

- b. In the case of a covered dependent who is the spouse of a covered person, if "a" above does not establish which plan is primary, the plan under which he/she is a dependent shall be primary.
- c. In the case of a covered dependent who is the child of a covered person, expenses should be submitted first to the plan of the parent whose birthday falls earlier in the calendar year. If both parents share the same birthday, expenses should be submitted in alphabetical order of the parents' first names.
- **d.** If the parents are separated or divorced:
 - 1. claims should be submitted first to the plan of the parent with custody of the child;
 - **2.** then the plan of the spouse of the parent with custody of the child;
 - **3.** then the plan of the parent not having custody of the child;
 - **4.** then the plan of the spouse of the parent not having custody of the child.
- e. For dental accidents, extended health care plans with accidental dental coverage determine benefits before dental plans.
- f. If the Order of Benefit Payment cannot be established by the above rules, benefits will be pro-rated among the plans in proportion to the amounts that would have been paid under each plan had there been coverage by that plan only.

D. Right of Recovery

Whenever benefit payments have been made under this Plan and by the other group plan which are in excess of the amount of allowable expenses, the Company shall have the right to recover such overpayments. Recovery of overpayment, if any, may be made at a later date from any person to, for or with respect to whom such payments were made, any insurance companies and any other organizations.

E. Release of Information

Any person claiming benefits under this Plan must authorize the release of such information as may be necessary to implement this Section 3.

Section 4. Subrogation

In the event of any payments to the employee for services under the Health Benefit Plan provided in Appendix B for which the employee may have a cause of action against **a** third party the Company and its Carrier will have their interests subrogated in this regard. This will entitle the Company and its Carrier to be reimbursed for any amount that the employee recovers in payment for the above services from any third party.

The employee will execute and deliver such instruments and papers as may be required and do whatever else is necessary to ensure such rights. The employee may take no action which may prejudice the subrogation rights.

The subrogation rights referred to above do not apply to an individual policy purchased by the employee.

Section 5. Administrative Benefit Guidelines

There may be some expenses, including drugs not legally requiring a prescription, that are not explicitly covered in this agreement but are covered now or may be covered in the future by administrative practice by agreement of the parties or due to changes in accepted administrative guidelines, benefit standards, technology or other similar reasons. The Administrator of this Plan has the right to amend the coverage for any expenses not explicitly contained in this agreement in order to conform with accepted benefit standards and guidelines.

ARTICLE IX PRESCRIPTION DRUG EXPENSE BENEFITS EMPLOYEES AND THEIR DEPENDENTS

Section 1. Eligibility

Prescription drug expense benefits **as** set out herein will be payable if a covered employee or a covered employee's dependents incur expenses for eligible prescription drugs and medicines as a result of an injury not entitling him to benefits under any Workplace Safety and Insurance or occupational disease law, or of sickness not entitling him/her to benefits under any such law **as** follows.

Charges will be payable for:

- any drug or medicine that is dispensed by a licensed Pharmacist and which by law or convention requires the written prescription of a Physician or Dentist;
- (2) the following non-prescription drugs, when prescribed by a Physician: Actifed, Aveeno, Acetaminophen, Benadryl, Claritin, Dimetapp, Dimetane, Duo C.V.P., Hismanal, Maxenal, Ornade, Peritrate, Phenegran, Pro-banthine, Sudafed, Tavist;
- (3) injectable medications;
- (4) life-sustaining drugs;
- (5) oral contraceptives, intrauterine devices and diaphragms; and
- (6) standard syringes, needles and diagnostic aids required for the treatment of diabetes (charges for cotton swabs, rubbing alcohol, and similar equipment are not covered).
- (7) drugs used in the treatment of sexual dysfunction subject to a monthly maximum of 8 pills and an annual maximum of \$1200.

Charges will not be payable for:

(1) preventative vaccines (oral or injected);

- (2) the administration of serums, vaccines, or injectable drugs;
- (3) drugs, biologicals and related preparations which are intended to be administered in hospital on an in-patient basis or out-patient basis and are not intended for use at home;
- (4) dietary supplements, health foods and nutritional products, vitamins (oral or injected);
- (5) anti-obesity drugs and weight-loss programs: and
- (6) treatment of varicose veins

Section 2. Amount of Benefit

- A. Payment for covered Prescription Drugs shall be made after a co payment of two dollars (\$2) has been assessed and shall not exceed the actual charge to the employee for the reasonable and customary charge for such drugs and medicines.
- **B.** The deductible in respect of each prescription drug expense equals the amount **of** the dispensing fee applied to the cost of purchasing the prescription drug from the pharmacy that is greater than \$8.00.
- C. The term drugs and medicines as used herein shall mean drugs and medicines legally requiring a prescription, except as indicated in Article VIII, Section 5 or as may be specifically agreed to by the parties.

Section 3. Exclusions and Limitations

- A. No payment shall be made in any event with respect to charges for drugs and medicines administered or obtained while the employee or dependent is confined to a hospital.
- B. No payment shall be made for charges for any prescription refill in excess of the number specified by the Physician or any refill dispensed after one year from the date of the Physician's latest order.

C. Payment of covered expenses under this benefit, which may be purchased in large quantities, will be limited to the purchase of up to a 3 month's supply at any one time.

ARTICLE X DENTAL EXPENSE BENEFITS EMPLOYEES AND THEIR DEPENDENTS

Section 1. Eligibility

- A. Benefits will be paid to an employee if a covered employee or a covered employee's dependent incurs Dental Expenses as hereinafter provided and such expenses are incurred on or after the effective date of coverage for such covered employee.
- B. Notwithstandingany other provision of the Health Benefit Plan, an employee who was hired on or before 31 December 1979 will be covered for the benefits provided in this Article.
- C. Any employee hired on or after 1 January 1980 will be covered for benefits provided in this Article on the first day of the month following one (1) year of continuous employment.

Section 2. Effective Date

The Benefits provided in this Article will become effective 1 January 1980.

Section 3. Hospital Expenses

A If a covered employee or a covered employee's dependent is confined as a resident patient in a hospital for medically necessary treatment in connection with injuries or disease of a dental nature, the daily room and board and miscellaneous fee expenses incurred during such confinement will be payable under the provisions of Article IV and V.

Section 4. Indemnity Limit

- A. The maximum benefit payable in any one (1) calendar year for benefits under this Article, except benefits described in Section 5-A-(1) and 5-A-(5) will be one thousand, four hundred dollars (\$1,400)
- B. The maximum benefit payable in connection with orthodontic treatment will be one thousand, five hundred dollars (\$1,500) for all such expenses incurred during the life time of the covered person. The one thousand, five hundred dollars (\$1,500) is applicable only to treatment on or after 01 October 1998.
- C. The maximum benefits payable in Paragraphs A and B above will apply separately to each covered employee and to each covered employee's dependent.

Section 5. Covered Dental Expenses

- A. Benefits will be paid to the employee subject to the limitations and provisions hereinafter contained in an amount equal to the actual expense to a covered employee of the reasonable and Customary charge of a licensed dentist, a dental hygienist working under a dentist's supervision or a denturist operating under the scope of their license for the dental services and supplies received by a covered employee or a covered employee's dependent for the necessary treatment as hereinafter listed.
 - (1) One hundred percent (100%) of the reasonable and customary charge for:
 - a. Complete oral examinations, one per 2 calendar years
 - **b.** Full mouth x-rays, one per 2 calendar years
 - c. Recall examinations, two per calendar year
 - d. Bite wing x-rays, twice per calendar year
 - e. Routine diagnostic and laboratory procedures
 - f. One unit of light scaling and one unit of polishing twice per calendar year,
 - g. Flouride treatment, twice per calendar year
 - h. Oral hygiene instruction, initial plus one recall

- i. Space maintainers (excluding appliances placed for orthodontic purposes)
- j. Fillings, (amalgam, silicate, acrylic and composite), retentive pins and pit and fissure sealants. Replacement fillings are covered only if:
 - The existing filling is at least 12 months old and required due to significant breakdown of the existing filling or recurrent decay; or
 - II. The existing filling is amalgam and there is medical evidence indicating that the patient is allergic to amalgam
- Pre-fabricated full-coverage restorations (metal and plastic)
- Minor surgical procedures, simple extractions, and post surgical care
- m. Complicated extractions including impacted and residual roots
- Consultation, anaesthesia and conscious sedation
- Denture repairs, relines and rebases, only if the expense is incurred later than 3 months after the date of the initial placement of the denture
- p. Injection of antibiotic drugs when administered by a Dentist in conjunction with dental surgery
- (2) One hundred percent (100%) of the reasonable and customary charge for:
 - a. Surgical procedures not included in Section
 5.A. (1) of this Article (excluding implant surgery)
 - b. Periodontal services for treatment of diseases of the gums and other supporting tissue of the teeth, including:
 - Scaling not covered under Section 5.A. (1) of this Article, and root planning, up to a combined maximum of 16 units per calendar year;
 - ii. Provisional splinting; and

- iii. Occlusal equilibration, up to a maximum of 8 units per calendar year
- c. Endodontic services (which include root canals and therapy, root amputation, apexifications and periapical services). Root canals and therapy are limited to one initial treatment plus one re-treatment per tooth per lifetime. Retreatment is covered only if the expense is incurred more than 12 months after the initial treatment
- Inlays (covering at least 3 surfaces provided the tooth cusp is missing).
- Crowns and onlays (only when function is impaired due to cuspal or incisal angle damage caused by trauma or decay).
- (3) Fifty percent (50%) of the reasonable and customary charge for:
 - a. Initial provision of full or partial removable dentures
 - **b.** Replacement of removable dentures, provided the new dentures are necessary due to one of the following:
 - i. A natural tooth is extracted and the existing appliance cannot be made serviceable
 - ii. The existing appliance is at least 60 months old and cannot be made serviceable
 - iii. The existing appliance is temporary and within 12 months of its installation it is replaced by a permanent denture. The total amount payable for both the temporary and permanent dentures is the amount which would have been allowed for permanent dentures.

Normally, dentures will be replaced by dentures, but if achieving a professionally acceptable course of treatment requires bridgework, such bridgework will be a Covered Dental Expense.

(4) Fifty percent (50%) of the reasonable and customary charge for:

- a. Initial provision of fixed bridgework
- b. Replacement of fixed bridgework or the addition of teeth to bridgework, provided the replacement or addition is due to one of the following:
 - A natural tooth is extracted and the existing appliance cannot be made serviceable
 - The existing appliance is at least 60 months old and cannot be made serviceable
 - iii. The existing appliance is temporary and within 12 months of its installation it is replaced by a permanent bridge. The total amount payable for both the temporary and permanent bridge is the amount which would have been allowed for a permanent bridge.
- (5) Fifty percent (50%) of the reasonable and customary charge for:
 - a. Correction of malocclusion of the teeth
 - b. Observation and adjustment
 - Appliances for tooth guidance or uncomplicated tooth movement
 - d. Appliances to control oral habits
 - e. Retention appliances
 - **f.** Fixed or cemented, unilateral and bilateral appliances

Section 6. Treatment Plan.

- A. If a course of treatment can reasonably be expected to involve Covered Dental Expenses of more than five hundred(\$500), a description of the procedures to be performed and an estimate of the dentist's charges should be filed with the Administrator prior to the commencement of the course of treatment. The Administrator will notify the employee of the amount, if any, that is payable.
- **B.** In determining the amount of benefits payable, professional consideration will be given to procedures, services, or courses of treatment that are customarily provided by the dental profession in conformity with

good professional practices and standards that may be performed for the dental condition concerned. The amount payable for Covered Dental Expenses will be based on the current Fee Guide for General Practitioners and Specialists approved by the Provincial Dental Association in the Province in which the covered employee resides. If the expense is not included in the Dental Fee Guide, reasonable and customary charges will be determined by the Administrator. When two or more courses of treatment covered under this Article would produce professionally adequate results for a given condition, the amount of benefits payable will be determined based on the reasonable and customary charge for the least expensive course of treatment.

- C. If a description of procedures to be performed and an estimate of the dentist's charges are not submitted in advance, the Administrator reserves the right to make a determination of benefits payable under this Article taking into account procedures, services or courses of treatment which will provide a professionally adequate result.
- D. No Treatment Plan is required to be submitted in advance of treatment under five hundred (\$500) or to emergency treatment, oral examinations, X-rays or prophylaxis. A course of treatment is one (1) or more treatments in a planned series resulting from a dental examination.

Section 7. Limitations

A. Restorative:

 Gold, baked porcelain restorations, crowns and jackets. If a tooth can be restored with a material such as amalgam, appropriate payment for that procedure will be made toward the charge for another type of restoration selected by the patient and the dentist. The balance of the treatment charge will remain the responsibility of the patient.

2. Reconstruction

Appropriate payment will be made toward the cost of procedures necessary to eliminate oral disease and to replace missing teeth. Appliances or restorations necessary to increase vertical dimension or restore the occlusion will be considered optional and their cost will remain the responsibility of the patient.

B. Orthodontics:

- If orthodontic treatment is terminated for any reason before completion, the obligation to pay benefits will cease with payment to the date of termination.
- The monthly benefit payment obligation under the orthodontic benefits provision shall cease on the termination date of this Agreement unless renewed or extended.

Section 8. Exclusions

Covered Dental Expenses do not include and no benefits are payable for:

- A. Charges for which benefits are otherwise provided under this Health Benefit Plan, through a government plan or a legally mandated program.
- **B.** Charges related to war, insurrection, the hostile action of any armed forces or participation in a riot.
- C. Charges related to the conviction of a criminal offence (excluding impaired driving or attempted assault)
- D. Charges for treatment rendered for a full mouth reconstruction, for a vertical dimension or for a correction of temporomandibular joint dysfunction.
- E. Charges for treatment by other than a dentist or denturist, except that scaling or cleaning of teeth may be performed by a licensed dental hygienist if the treatment is rendered under the supervision and guidance of the dentist.
- F. Charges for services and supplies that are solely cosmetic in nature, including charges for personalization

- or characterization of dentures, unless needed because of an accidental injury while covered under this Plan.
- G. Charges for prosthetic devices (including bridges and crowns) and the fitting thereof which were ordered while the individual was not covered for Dental Expense Benefits or which were ordered while the individual was insured for Dental Expense Benefits but are finally installed or delivered to such individual more than sixty (60) days after termination of insurance.
- **H.** Charges for the replacement of a lost, missing or stolen removable appliance.
- Charges set forth in "Exclusions" applicable to all Articles.
- J. Charges for failure to keep a scheduled visit with the dentist or denturist.
- K. Services provided by dental laboratories or expanded dental auxiliaries unless such dental laboratories or dental auxiliaries are licensed in accordance with the requirements of the jurisdiction in which it is located and the charges are reasonable and customary.
- L. Self-inflicted injuries or illnesses, whether the person is sane or insane.
- M. Charges for services or supplies when there would have been no charge at all in the absence of this benefit.
- N. Anti-snoring or sleep apnea devices.

Section 9. Definitions

- A. The term "Dentist" means a legally licensed dentist practicing within the scope of his/her license. For the purposes of this Article, the term "Dentist" also includes a legally licensed physician authorized by his/her license to perform the particular dental services he/she has rendered.
- **B.** The term "Orthodontic Treatment" means the preventive and corrective treatment of all those dental irregularities which result from the anomalous growth and development of dentition and its related anatomic structures or as a result of accidental injury and which

require repositioning of teeth to establish normal occlusion.

Section 10. Coordination With Other Dental Expense Benefits

A. The Company shall follow the same procedures with respect to the Dental Expense Benefits concerning coordination of benefits as is set forth in Article VIII, Section 3.

ARTICLE XI VISION CARE EXPENSE BENEFITS EMPLOYEES AND THEIR DEPENDENTS

Section 1. Eligibility

- A. Benefits will be paid to an employee if a covered employee or a covered employee's dependent incurs Vision Care Expense as hereinafter provided and such expenses are incurred on or after the effective date of coverage for such covered employee.
- B. Notwithstanding any other provisions of the Health Benefit Plan, an employee who has attained seniority as of 31 December 1979 will be covered for benefits provided in this Article.
- C. Any employee hired before 31 December 1979 who had not obtained seniority by 31 December 1979 will be covered for benefits provided in this Article on the first day of the month following attainment of seniority.
- D. Any employee hired on or after 1 January 1980 will be covered for benefits provided in this Article on the first day of the month following one (1) year of continuous employment.

Section 2. Effective Date

The benefits provided in this Article shall be effective 1 January 1980.

Section 3. Covered Vision Care Benefits

- A. Benefits will be paid to the employee subject to the limitations and provisions hereinafter contained for covered expenses when prescribed by an Ophthalmologist, Optometrist or Oculist as follows:
 - (1) Purchase and fitting of prescription glasses or elective contact lenses, as well as repairs, or elective laser vision correction procedures, up to the indemnity limits in Section 4 A of this Article, once during any period of 24 consecutive months, and once during any period of 12 consecutive months for covered dependents age 16 and under, and then only if new lenses are prescribed.
 - (2) Frames for prescription glasses once in any period of 24 consecutive months from the last receipt of frames and once during any period of 12 consecutive months for covered dependents age 16 and under, and then only if new lenses are prescribed.
 - (3) Cost, up to \$100 per covered person, for eye exams including refractions, once during any period of 24 consecutive months, and once during any period of 12 consecutive months for covered dependents age 16 and under. Payment for eye exams will not be in addition to provincial coverage] only in place of it in provinces where this coverage does not exist.
 - (4) Visual training.

Section 4. Indemnity Limits

A The maximum benefit payable in any period of 24 consecutive months for covered employees and covered dependents over age 16 and in any period of 12 consecutive months for dependents 16 and under when necessitated by a prescription change shall be \$135.00 for materials and professional services as provided in Section 3(1).(2) and(4).

B. The limitations provided in this section shall apply whether or not a claim for benefits are for replacement of lost, stolen or broken lenses, contact lenses or frames.

Section 5. Exclusions

- A. Covered Vision Care Expenses do not include and no benefits are payable for:
 - (1) Lenses which do not require a prescription.
 - (2) Charges for which benefits are otherwise provided under this Health Benefit Plan and the Ontario Health Insurance Plan
 - (3) Procedures determined to be special or unusual, such as, but not limited to, orthoptics, subnormal vision aids and aniseikonic lenses.
 - (4) Vision examinations or materials furnished for any condition, disease, ailment, or injury arising cut of and in the course of employment.
 - (5) Services rendered and materials ordered:
 - (a) before the employee became eligible for this benefit;
 - (b) after termination of the insured employee's employment; except for materials ordered while covered by this Plan and delivered within thirty (30) days after the date of termination;
 - (c) for which the employee is not charged or is not obligated to pay.
 - (6) Charges for failure to keep a scheduled visit with the Ophthalmologist, Optometrist or Oculist.

ARTICLE XII HEARING AID EXPENSE BENEFIT PLAN FOR EMPLOYEES AND THEIR DEPENDENTS

Section 1. Eligibility

A. Benefits will be paid to an employee if a covered employee or a covered employee's dependent incurs

- Hearing Aid Expense as hereinafter provided and such expenses are incurred on or after the effective date of coverage for such covered employees.
- B. Notwithstanding any other provisions of the Health Benefit Plan an employee hired on or before 31 December 1979 will be covered for benefits provided in this Article.
- C. Any employee hired on or after 1 January 1980 will be covered for benefits provided in this Article on the first day of the month following one (1) year of continuous employment.

Section 2. Effective Date

The benefits provided in this Article shall be effective 1 April 1980.

Section 3. Covered Hearing Aid Expense Benefits

- A. Benefits, subject to certain limitations will be paid in an amount based on the actual charge for the service or hearing aid supplies. Benefits under the plan are:
 - (1) Hearing aids designed for in-the-ear, behind-the-ear or on the body use and only if the hearing aid prescribed is based upon the audiometric examination and hearing aid evaluation test and the hearing aid provided is the make and model prescribed by the otologist, otolaryngologist or clinical audiologist.

Section 4. Indemnity Limits

- A. Benefits for services summarized in Section 3-A-(1) are limited to one (1) hearing aid in any period of thirty-six (36) calendar months after the most recent hearing aid for which benefits were payable under the Plan.
- **B.** The benefit covered under the Plan will be the actual charge, but not more than \$425.00 for the cost and installation of a hearing aid including the hearing aid, initial batteries and a minimum of six (6) months necessary follow-up care.

Section 5. Coordination of Benefits

The Company shall coordinate payment for services and supplies provided by the Hearing Aid Expense Benefit Plan with other group plans which provide hearing aid benefits as is set forth in Article VIII, Section 3.

Section 6. Services Not Covered

Covered hearing aid benefit expenses do not include and no benefits are payable for the following:

- Audiometric examinations and/or hearing aid evaluation tests.
- **B.** Medical or surgical treatment.
- C. Examinations, tests and hearing aids provided under Workplace Safety and Insurance law.
- D. Drugs or other medications.
- E. Hearing aids ordered before a person became eligible for benefits under this plan or after termination of coverage.
- F. Hearing aids ordered while covered by the plan, but delivered more than 60 days after termination of coverage.
- G. Charges for hearing aids for which no charge is made to the patient or for which no charge would be made in the absence of this plan.
- H. Charges for hearing aids which do not meet professionally accepted standards or practice, or which are not recommended or approved by the otologist, otolaryngologist or clinical audiologist.
- Charges for hearing aids received as a result of ear disease, defect or injury due to an act of war declared or undeclared.
- J. Charges for hearing aids provided by any governmental agency which are obtained without cost by compliance with any federal, provincial, municipal or other governmental body law or regulation, or any charges to the extent they are payable under any health care program supported in whole or in pari by funds of the

- federal government or any provincial or political subdivision thereof.
- **K.** Replacement of hearing aids that are lost or broken unless, at the time of such replacement, the covered person is otherwise eligible as described in Section 4-A.
- L. Charges for the completion of any insurance forms.
- **M.** Replacement parts for and repairs of hearing aids.
- N. Eyeglass-type hearing aids, to the extent the charge exceeds the covered hearing aid expense for one hearing aid.
- Charges for failure to keep a scheduled visit with an audiologist.

APPENDIX "C" THE DISABILITY BENEFIT PLAN FOR WAGE EMPLOYEES

ARTICLE I ESTABLISHMENT OF PLAN

Section 1. Preamble

John Deere Welland Works of John Deere Limited, hereinafter referred to as the Company, will provide the Disability Benefit Plan for Wage Employees (hereinafter referred to as the Plan). This Plan shall make available Weekly Indemnity coverage for Total Disability (hereinafter referred to as Weekly Indemnity), Layoff Disability Benefits coverage, and Long-Term Disability Benefits coverage, as hereinafter set forth. The Weekly Indemnity coverage has been amended effective 1 October 1998 to qualify for registration with the Employment Insurance program.

Section 2. Claims Covered Under New Plan

The benefits provided in this Plan shall be payable with respect to any claims initially incurred on or after 1 October 1998. With respect to claims initially incurred prior to 1 October 1998, benefits will be payable as per the collective agreement in effect at the date the claim was incurred.

Section 3. Effective Dates of Coverage

Employees shall have coverage effective the first day of the month following attainment of seniority, provided the employee is actively at work on that day. Otherwise, coverage will be effective on such later date that the employee first reports to work on a full-time basis.

Section 4. Physical Examinations

The Company, at its own expense, shall have the right to require an employee to submit to an examination by an independent physician designated by it for the purpose of determining continuing disability. The Company shall have the

right to examine the employee as often as it may reasonably require during an extended disability.

Section 5. Termination

Employee coverage referred to in Section 1 will terminate when the employee's employment terminates subject to the provisions of this Plan.

Section 6. Cost of Benefits

Except as otherwise specifically provided, the cost of providing benefits under this Plan will be borne by the Company and no contribution to the Plan shall be made by any employee, or retired employee.

Section 7. Amendment, Modification, and Plan Termination

Except as otherwise specifically provided, the Company may at any time amend, modify, or terminate this Plan, provided however, that no change shall reduce the amount of any benefit to which an employee or retired employee shall be entitled on account of disability incurred prior to the effective date of such change.

Section 8. Funding

Except as otherwise specifically provided, benefits shall be provided through an insurance company selected by the Company, a fund established by the Company, or from the general assets of the Company.

Section 9. Application for Benefits

In order for any employee to receive benefits pursuant to the provisions of this Plan, such employee must file a written application. The necessary form(s) for such purpose will be supplied by the Company.

Section 10. Denial of Benefits

When application for benefits provided by this Plan is denied in full or in part, the employee shall receive a written notice of the reason or reasons for such denial.

ARTICLE II CONTINUATION OF COVERAGE

Section 1. Leave of Absence

Weekly Indemnity coverage may be continued by the Company for thirty-one (31) days in the event of a leave of absence, or as agreed to between the Company and the employee during a maternity or parental leave of absence.

Section 2. Layoff

- A. Weekly Indemnity coverage will be continued for thirtyone (31) days in the event of a layoff.
- B. If an employee becomes disabled while on layoff, after thirty-one (31) days, the employee may be eligible for a Layoff Disability Benefit as set forth in Article IV, Section 3. An employee with seniority who has at least one credit unit under the SUB Plan will have Layoff Disability Insurance continued subject to the Nonduplication of Benefits provision in Article V, Section 2 for a period of time which will be determined by the number of credit units credited to the employee at the time of layoff in accordance with the provisions of the Supplemental Unemployment Benefit Plan. One month's coverage will be provided for each four (4) credit units the employee is credited with as of the date of layoff but not to exceed twelve (12) months.

ARTICLE III AMOUNT OF BENEFIT

Section 1. Determination of Earnings

A. The Earnings of each Wage employee upon which his/her Weekly Indemnity coverage for Total Disability and Long-Term Disability is based shall be determined as follows: The individual employee's hourly earning rate shall be his/her average straight-time hourly earnings as of the first day of each January, April, July and October which shall determine the benefit for each three (3) months' period beginning the following February, May, August and November. The method of computing the wage employee's average straight-time hourly earnings as of the 1st day of each January, April, July and October shall be as follows:

a. Divide the sum of all money earned during the three (3) calendar months preceding such date for hours worked, excluding overtime penalty pay, by the total of such hours worked. Newly hired wage employees or employees transferred from salary payroll to wage payroll shall have their benefit based upon the wage rate of the classification to which they are first assigned until the next regular period for benefit determination as outlined.

Your average earnings in these months	Jan Feb. Mar	Apr. May Jun.	Jul. Aug. Sep.	Oct. Nov. Dec.
As determined on these dates	1 Apr.	1 Jul.	1 Oct.	1 Jan
Determines Weekly Indemnity Benefits for claims commencing in these months	May. Jun. Jul.	Aug. Sep. Oct.	Nov. Dec. Jan.	Feb. Mar. Apr.

- B. In determining the Earnings on and after 1 October 1998 the cost-of-living allowance provided through June 1998 shall be included for employees at work on or after 1 November 1998.
- C. In determining the Earnings for Weekly Indemnity on and after 1 October 1998, any cost-of-living allowance, gain-sharing and overtime pay shall be added to the Earnings up to the extent needed such that the benefit amount calculated in section 2 of this Article equals 55% of such weekly earnings up to a maximum of \$413 per week.

Section 2. Benefit Amount

- A. The Weekly Indemnity Benefit amount shall be 55% of the employee's weekly earnings to a maximum benefit of \$503 per week. An employee's weekly earnings are computed as the Earnings determined in section 1 of this Article multiplied by forty (40).
- B. The monthly Long Term Disability Benefit amount shall be 50% of the employee's monthly earnings to a maximum benefit of \$1,975 per month. An employee's monthly earnings are computed as the Earnings determined in section 1 of this Article multiplied by twothousand eighty (2,080) and divided by twelve (12).
- C. Any Weekly Indemnity Benefits due for periods less than a full week shall be paid on the basis of one-fifth (I/5) of the weekly benefit for each day of an employee's regular five (5) day work-week.

ARTICLE IV COMMENCEMENT AND DURATION OF BENEFITS

Section 1. Weekly Indemnity

- A. If a covered employee shall be totally disabled as a result of occupational or non-occupational accidental bodily injury or sickness not hereinafter excepted, or as a result of pregnancy, childbirth or miscarriage, and provided such disability shall commence while the employee is covered under this Section, benefits will be paid to the employee for the period of such disability subject to the indemnity limits hereinafter set forth, in an amount determined as herein outlined.
- **B.** Covered employees are eligible for Weekly Indemnity Benefits for a period of time equal to their continuous employment at the time a non-occupational disability commences but in any case for not less than fifteen (15) weeks or for more than fifty-two (52) weeks. If such employee is confined in a hospital at the date of expiration of his/her Weekly Indemnity Benefits and such benefits were payable for less than fifty-two (52) weeks, benefits shall continue to be paid while he/she continues

to be confined but in no case beyond fifty-two (52) weeks. If the disability is occupational, benefits shall continue to be paid while he/she is disabled, but in no case beyond fifty-two (52) weeks.

C. Benefits cease:

- (1) at the earlier of the above noted periods, or
- (2) at the date the employee is no longer disabled, or
- (3) at age 65, or retirement if earlier, or
- (4) at layoff, but in no case less than 15 weeks if disability began prior to notice of layoff, except as otherwise specified in Article II, Section 2(A).
- D. Weekly Indemnity Benefits will be paid for not more than fifty-two (52) weeks during any one period of disability, commencing with (1) the eighth day of disability caused by accident except that if during the first seven days of disability the employee, because of such accident, becomes confined to a hospital or is treated by a duly qualified physician, there shall be no waiting period; or (2) the first day the employee is confined in a hospital; or (3) the eighth day of disability caused by sickness; or (4) the day on which the employee undergoes a surgical procedure, including an oral surgical procedure, for which the employee was not hospitalized and for which a benefit is payable by a government-sponsored health plan.
- E For the purposes of the coverage under this Section, successive periods of disability separated by less than two (2) weeks of active work on full time shall be considered one period of disability unless the subsequent disability is due to an injury or sickness entirely unrelated to the cause of the previous disability and commences after return to work.
- F. No change in classification which will provide an increase or decrease in amount of benefits shall become effective during any one period of disability.
- G. The amount of Weekly Indemnity Benefits for the first and last day of disability will be determined on the following basis:

- (1) If the employee has worked or has been paid for less than three (3) hours, the full Weekly Indemnity Benefit amount will be paid for that day.
- (2) If the employee has worked or has been paid for three (3) or more, but less than six (6) hours, onehalf the Weekly Indemnity Benefit amount will be paid for that day.
- (3) If the employee has worked or has been paid six (6) or more hours, no Weekly Indemnity Benefit will be paid for that day.
- H. Benefits under this Section will not be payable if the disability is due to a concurrent existing occupational disability for which he/she is eligible to receive benefits payable under Workplace Safety and Insurance laws.
- Benefit payments for disability caused by sickness will not commence prior to the date the employee is treated by a duly qualified physician.
- J. A disabled employee who is released to return to work by the attending physician, but is temporarily prevented from returning to his/her regular job because of medical restrictions and there are no vacant jobs the employee is capable of performing, will be continued, if eligible, on Weekly Indemnity status.
- K. When an employee has been totally disabled as the result of accidental bodily injury, returns to work and subsequently experiences any recurrence of total disability within 180 calendar days of that return, there shall be no waiting period for Weekly Indemnity Benefits, provided medical evidence indicates the total disability is due to the same injury. When such total disability commences after 180 calendar days following the first day of return to work subsequent to the initial period of disability.
 - Such subsequent periods shall be considered a disability due to sickness, and
 - (2) The waiting period shall be applied to such period or periods of disability as provided under this Section.

L. Exclusions and Limitations

- (1) This Section shall not cover disability resulting from:
 - Sickness, pregnancy, childbirth, or miscarriage for which the employee is not treated by a duly qualified physician.
 - b. An injury or sickness due to war or any act of war, whether war is declared or not, or due to any act of international armed conflict, or conflict involving the armed forces of any international authority.
 - An injury or sickness that is intentionally selfinflicted.
 - A conviction of a criminal offence (excluding impaired driving & attempted assault).
- (2) The benefits under this Section shall not be payable:
 - a. For claims incurred during a strike or lockout.
 - b. To an employee who is not receiving continuous treatment for the use of drugs or alcohol when the illness results from the use of one of these substances.
 - c. To an employee who is serving a prison sentence.
- (3) If an employee is entitled to a benefit under a Workplace Safety and Insurance law for an occupational disability, the employee's Weekly Indemnity Benefit payable under this plan will be offset by the amount of the benefit under the Workplace Safety and Insurance law.
- (4) When an employee participates in a return to work or a modified work program, the amount of benefit payable under this section will be adjusted on an hourly basis for the number of hours that the employee receives pay. The hourly rate will be the Weekly Indemnity rate divided by 40.
- M. An employee may waive irrevocably any right he/she may have to receive Weekly Indemnity Benefits with

respect to any period of disability by completing a waiver form furnished by the Company for that purpose. No Weekly Indemnity Benefits shall be payable for any period of disability covered by such waiver.

- N. Where there is a dispute as to whether or not an injury suffered by an employee grew out of his/her employment with the Company, the following procedure will be followed:
 - (1) The employee shall receive an amount of money equal to his/her current Weekly Indemnity rate, but this benefit will not be considered either Weekly Indemnity or Workplace Safety and Insurance benefits until such time as the dispute is finally resolved.
 - (2) The employee will be required to sign a reimbursement form which will provide that any Workplace Safety and Insurance judgment in favour of the employee which duplicates a payment previously made by the Company will be returned to the Company by the employee, or deducted from any final settlement the Company may be required to make.

It is understood that any of the above actions taken while the dispute is pending will in no way impair the rights of the employee or the Company nor be used to prejudice the position of either.

Section 2. Employment Insurance Program

A. The parties agree to review this Plan should any provisions of the Employment Insurance program change to disqualify the Plan from registration under the Employment Insurance program.

Section 3. Layoff Disability Benefits

A. The benefits provided in this Section shall be payable with respect to weeks of disability to any employee who is entitled to continue the coverage provided in Section 1 of this Article.

- B. The Layoff Disability Benefit shall be equal in amount to the Weekly Indemnity Benefit under Section 1 and 2 of Article III, subject to the provisions of Paragraph C below
- C. Qualifications for Benefits
 - (1) If, as a result of disease or accidental bodily injury for which the employee is not entitled to benefits under any Workplace Safety and Insurance Law, an employee becomes totally disabled while on a qualifying layoff as defined in Appendix "D", "Supplemental Unemployment Benefit Plan", a Layoff Disability Benefit shall be paid for each week of disability while on layoff provided, however, that:
 - a. The employee must have been eligible for a Regular Benefit under the SUB Plan or have been employed by another employer immediately prior to his/her having become disabled:
 - b. No Layoff Disability Benefit shall be payable for any week which the employee does not have to his/her credit at least one (1) credit unit under the SUB Plan;
 - c. No Layoff Disability Benefit shall be payable for any week during which the employee is not under the care of a physician;
 - d. No Layoff Disability Benefit shall be payable for any week during which the employee receives Weekly Indemnity Benefits or Long-Term Disability Benefits under this Article;
 - The Layoff Disability Benefit for any week shall be reduced by the amount of any Disability Benefit he/she receives for the same week under any plan financed in whole or in part by another employer;
 - f. Layoff Disability Benefits start on the first day following the last day for which a SUB Plan Regular Benefit was payable to the employee if he/she was receiving such Regular Benefits immediately prior to his/her becoming disabled,

- otherwise on the first day of the qualifying disability;
- g. No Layoff Disability Benefit shall be payable beyond the time the employee no longer satisfies the disability requirement except that if he/she remains on a qualifying layoff under the SUB Plan, benefits shall be payable for the remaining days in the same week as defined in the SUB Plan for which he/she does not receive a Regular Benefit;
- In no event will there be any duplication of Layoff Disability Benefits and SUB Benefits.
- D. An employee may waive irrevocably any right he/she may have to receive Layoff Disability Benefits with respect to any period of disability by completing a waiver form furnished by the Company for that purpose. No Layoff Disability Benefits shall be payable for any period of disability covered by such waiver.

Section 4. Long-Term Disability Benefits -- Occupational and Nonoccupational

A. Eligibility

- (1) If an employee at work shall become totally disabled as a result of bodily injury or sickness not hereinafter excepted and such disability continues beyond the period for which he/she is entitled to Weekly Indemnity under Section 1 of this Article or Layoff Disability under Section 3 of this Article, the disabled employee will be entitled during the continuation of such disability, subject to the provisions hereinafter set forth, to a monthly Long-Term Disability Benefit for the period set out below.
- (2) For the purpose of this section totally disabled means that the employee is unable to perform the duties of his/her own occupation or any occupation in the unit.

B. Commencement and Duration of Benefits

- (1) Benefits will commence the day following the last day of disability included in the maximum number of Weekly Indemnity Benefits or Layoff Disability Benefits and will be paid for not longer than the time specified below:
 - a. If the employee has less than two (2) years of continuous employment at the commencement of the disability -- one (1) year.
 - b. If the employee had two (2) or more years of continuous employment at the commencement of disability -- one (1) year plus the number of full months by which the employee's continuous employment at the commencement of the disability exceeded twenty-four (24) months.
 - c. The date of death.
 - **d.** If disability commences prior to attainment of age 60, until attainment of age 65.
 - e. If disability commences on or after attainment of age 60, until the first to occur of:
 - i attainment of age 69 or
 - ii the end of a period of 60 months less periods of Weekly Indemnity.
 - **f.** The date he/she no longer satisfies the disability requirements.
- (2) If an employes's return to work with the Company is not of sufficient duration to qualify him/her for a new period of Weekly Indemnity Benefits his/her period of disability shall not be deemed interrupted by such return. In such circumstances, the Long-Term Disability Benefit for the month shall be reduced proportionately for the period of return to work.
- (3) If an employee while receiving Long-Term Disability Benefits engages in any substantial gainful activity, his/her Long-Term Disability Benefits will be suspended for the period of such employment. After termination of such employment, Long-Term

- Disability Benefits will again be paid if the employee is then qualified as totally disabled.
- (4) Benefits accrued under this Plan during any calendar month will be payable as of the first day of the next succeeding month. Benefits payable for a period of less than a full month shall be prorated on the basis of the ratio of calendar days of eligibility to total calendar days in the month.
- (5) For the purpose of determining the maximum period for monthly Long-Term Disability Benefits any month in which such benefits are partially or wholly offset by benefit payments from sources listed below or because of return to work, i.e., not of sufficient duration to qualify him/her for a new period of Weekly Indemnity Benefits, such month shall be counted as a full month. Fractions of the first month and the last month shall be counted as fractions of a month
- (6) The cumulative total number of months during any previous periods of eligibility for Long-Term Disability Benefits for the same disability will reduce the maximum number of monthly benefit payments for which the employee is otherwise eligible when Long-Term Disability Benefits again commence.

C. Amount of Benefit

- (1) The monthly Long-Term Disability Benefit will be the amount computed in Article III Section 2 of this Plan. The monthly Long-Term Disability Benefit shall be reduced by an amount equal to the monthly equivalent of the total of the following benefits:
 - All disability retirement benefits for which the employee is eligible under any Company Pension Plan.
 - Early or normal retirement benefits the employee is receiving under any Company Pension Plan.
 - Lost-time benefits for which the employee is eligible under the Workplace Safety and

Insurance laws or other laws providing benefits for occupational injury or disease including lump sum settlements for future loss of earnings but excluding specific allowances for loss, or one hundred percent (100%) loss of use, of bodily member and any compensation for non-economic loss.

- **d.** Unreduced "Statutory Benefits" to which the person is entitled under the Old Age Security Act, the Canada Pension Plan, Federal or Provincial Law providing Old Age or Total or Permanent Disability Benefits or any future laws providing similar benefits.
- e. If the employee is not entitled to disability benefits described in (1)-d above, the amount of any temporary benefit which was payable under any Company Pension Plan prior to attainment of age 65 commencing when the employee is entitled to unreduced "Statutory Benefits" as described in (1)-d above.
- f. Benefits for which the employee is eligible under any Provincial or Federal law providing benefits for working time lost because of disability.
- (2) For employees at work on or after 1 April 1971 the amount of any increase in Paragraphs (1)-c, d or f above subsequent to the first day for which Long-Term Disability Benefits are payable (or, if later, the first day on which the amount of such governmental benefits for which reduction is made is initially determined) shall be disregarded for purposes of determining the amount of Long-Term Disability Benefit. Such increase shall not be disregarded if it represents an adjustment in the original determination.
- (3) The Company may require each employee eligible for Long-Term Disability Benefits to certify or furnish verification of the amounts of his/her deductible benefits from the sources listed above. The amount of any Long-Term Disability Benefit payments in excess

of the amount that should have been paid after reduction for such benefits will be deducted from future Long-Term Disability Benefits.

- (4) In determining the amount by which Long-Term Disability Benefits shall be reduced, the monthly equivalent of benefits paid on a weekly basis shall be computed by multiplying the weekly benefit rate by 4.33. In the case of lump sum settlements under the Workplace Safety and Insurance laws the deduction shall be the monthly equivalent of the amount of the Workplace Safety and Insurance Benefit to which the employee would have been entitled under applicable law had there been no lump sum payment, but not to exceed in total the amount of settlement.
- (5) Long-Term Disability Benefit computations shall presume eligibility for Canada Pension Plan Disability Benefits and if the employee has ten (10) years of service, total and permanent disability pension benefits. Deductions from Long-Term Disability Benefits will be made on this basis unless the person receiving benefits presents satisfactory evidence that these benefits were applied for and denied.
- (6) Increases in the Total and Permanent Disability pension benefits provided under Article VI of the John Deere Pension Plan for Wage Employees, which become effective for such benefits payable on or after 1 January 1980 shall be disregarded for purposes of determining the amount of Long-Term Disability Benefit. Such increase shall not be disregarded if it represents an adjustment in the original determination.

D. Rehabilitation

An employee will not be ineligible for Long-Term Disability Benefits because of work which is determined to be primarily for training under a recognized program of vocational rehabilitation.

When an employee participates in a return to work or a modified work program, the amount of benefit payable under this section will continue to be paid. In addition the employee will receive the difference of the rate of pay of the job and the calculated LTD hourly rate. The LTD hourly rate is the monthly benefit rate times 12 and divided by 2080.

E Proof of Disability

The Company may require an applicant as a condition of eligibility to submit to examination by a physician designated by it for the purpose of determining his/her initial or continuing disability.

F. Temporary Disability

A disabled employee who is released to return to work by the attending physician, but is temporarily prevented from returning to his/her regular job because of medical restrictions and there are no vacant jobs the employee is capable of performing, will be continued, if eligible, on Long-Term Disability status.

G. Exclusions and Limitations

- (1) The insurance under this Section shall not cover disability
 - resulting from an injury or sickness due to war
 or any act of war, whether war is declared or
 not, or due to any act of international armed
 conflict, or conflict involving the armed forces of
 any international authority;
 - resulting from services in the Armed Forces (unless he/she has been in employment with the employer at least ten (10) years after separation from such service).
 - resulting from an injury or sickness that is intentionally self-inflicted.
 - d. arising from the commission of a criminal offense for which the employee is convicted (excluding impaired driving or attempted assault).
 - resulting from cosmetic medical or surgical care.

ARTICLE V EXCLUSIONS AND LIMITATIONS

Section 1. Exclusions

No payment shall be made under this Plan in any event with respect to benefits which are payable by any employer in connection with injury or sickness and in accordance with the provisions of any Workplace Safety and Insurance law or similar law except as provided in Article IV, Section 2 and Section 4 of this Plan.

Section 2. Nonduplication of Benefits

- A. This provision shall apply to an employee whose insurance is continued during a period of layoff provided he/she is entitled to benefits as an insured employee of another employer.
- B. If any weekly disability benefits shall be provided under any other group insurance policy, or any other group plan by whatever name called, in connection with any injury, sickness or pregnancy, an amount equal to the total benefits provided through such policy or plan shall be deducted from the amount which otherwise would be payable under this Plan on account of such injury, sickness or pregnancy.
- C. The term "Other Group Plan" means any plan of another employer providing benefits or services for or by reason of medical care or treatment which benefits or services are provided on an insured or uninsured basis in connection with an employee's or an employee's dependent's employment, occupation or profession.
- D. In the event of any payment to the employee under the Disability Benefit Plan for loss of income for which the employee may have a cause of action against a third party the Company and its Carrier will have their joint interest subrogated in this regard. This will entitle the Company and its Carrier to be reimbursed for any amount that the employee recovers for loss of income from a third party which exceeds the employees actual loss of income.

The employee will execute and deliver such instruments and papers as may be required and do whatever else is necessary to secure such rights. The employee may take no action which may prejudice the subrogated rights.

The subrogated rights referred **to** in this paragraph do not apply **to** an individual plan purchased by the employee specifically for wage **loss** replacement.

APPENDIX "D" SUPPLEMENTAL UNEMPLOYMENT BENEFIT

ARTICLE I PRELIMINARY PROVISIONS

Section 1. Continuation and Amendment of the Plan

- A. The Supplemental Unemployment Benefit Plan in effect from 19 September 1983 to 18 January 1987 shall be contained as set forth in Appendix "D-1", Supplemental Unemployment Benefit Plan, as attached and effective 19 January 1987 except as otherwise specified herein and in the Plan.
- B. Provision for payment of benefits under the Supplemental Unemployment Benefit Plan referred to in A above, and in effect until 18 September 1983 shall be kept in full force and effect in accordance with the conditions, provisions, limitations, and amendments of such Supplemental Unemployment Benefit Plan, as constituted, until the effective date, as defined in the Plan.
- C. No adjustments will be made, for whatever reason, in Benefits paid prior to the effective date, as defined in the Plan.

Section 2. Term of Preliminary Provisions

A. These preliminary provisions and the Plan incorporated herein shall remain in full force and effect without change until the expiration date of the Collective Bargaining Agreement, except as there may be a termination under Section 2 above. As of that date these preliminary provisions may be terminated, modified, changed, or continued, subject to and in accordance with the provision of Article XXX of the Collective Bargaining Agreement of which these preliminary provisions are a part.

B. It is understood that termination of these preliminary provisions shall not have the effect of automatically terminating the Plan.

Section 3. Governmental Rulings

- A. The Amendments to the Plan which are provided for in Section 1 of these preliminary provisions and incorporated in Appendix "D-1" hereof shall not be effective prior to receipt by the Company from Canada Customs and Revenue Agency and Human Resources Development Canada rulings that amendments will not have any adverse effect upon the favorable rulings previously received by the Company that:
 - (1) The Plan qualifies for exemption from the Canadian Income Tax Act, Chapter 148, R.S.C. 1985 c.1 (5th supp.) as amended and now in effect.
- **B.** The Company shall apply promptly to the appropriate agencies for the rulings described in Subsection A of this Section.
- C. Notwithstanding any other provision of these preliminary provisions or of the Plan, the Company, may, during the term of these preliminary provisions make provisions in the Plan not inconsistent with the purposes, structure, and basic provisions thereof which shall be necessary to obtain or maintain any of the rulings referred to in Subsection A of this Section 3 or in Section 2 of Article VII of the Plan. Any such revisions shall adhere as closely as possible to the language and intent of the provisionsoutlined in Appendix "D-1".

APPENDIX "D1 --SUPPLEMENTAL UNEMPLOYMENT BENEFIT PLAN

ARTICLE | ELIGIBILITY FOR BENEFITS

Section 1. Eligibility for a Regular Benefit

- A An employee shall be eligible for a Regular Benefit for any week beginning on or after the approval has been received from the Government of Canada, in Ottawa, if with respect to such week he/she:
 - (1) Was on a qualifying layoff, as described in Section 2 of this Article, for all or part of the week.
 - (2) Received an Employment Insurance Benefit not currently under protest by the Company or was ineligible for an Employment Insurance Benefit only for one or more of the following reasons:
 - He did not have prior to layoff a sufficient period of employment or earnings covered by Employment Insurance;
 - Exhaustion of his/her Employment Insurance Benefit rights;
 - C. He was serving a "waiting week" or a partial "waiting week" of layoff under Employment Insurance:
 - d. He was denied an Employment Insurance Benefit and it is determined, with the concurrence of Human Resources Development Canada, that under the circumstances it would be contrary to the intent of the Plan and governmental policy to deny him/her a benefit.
 - (3) Has met any registration and reporting requirements for Employment Insurance.
 - (4) Had to his/her credit a credit unit or fraction thereof.
 - (5) Did not receive an Employment Insurance Benefit under any contract or program of another employer

or under any other "SUB" Plan of the Company (and was not eligible for such a benefit under a program of another employer with whom he/she has greater seniority than with the Company nor under any other "SUB" Plan of the Company in which he/she has credit units which were credited earlier than his/her oldest credit units under this Plan.)

- (6) Has made a benefit application in accordance with procedures established by the Company hereunder, and if he/she was ineligible for an Employment Insurance Benefit only for the reason set forth in Item b of Section I-A-(2) of this Article, is able to work, is available for work, and has not failed:
 - a. to maintain an active registration for work for Employment Insurance Benefits;
 - **b.** to do what a reasonable person would do to obtain work, and
 - c. to apply for or to accept available suitable work of which he/she has been notified by Human Resources Development Canada or by the Company.
- (7) Qualifies for a benefit of at least \$2.00.
- (8) Was not eligible for an automatic short week benefit.

Section 2. Conditions with Respect to Layoff

- A. A layoff for purposes of the Plan includes any reduction in force such as a temporary layoff or a layoff occurring or continuing because the employee was unable to do the work offered by the Company, although able to perform other work in the plant to which he/she would have been entitled if he/she had had sufficient seniority.
- **B.** An employee's layoff for all or part of any week will be deemed qualifying for Plan purposes only if:
 - (1) Such layoff was from the factory-wide unit;
 - (2) Such layoff was not for disciplinary reasons, and was not a consequence of:
 - a Any fault attributable to the employee;

- **b.** Any war or hostile act of a foreign power (but not government regulation or controls connected therewith):
- c. Sabotage or insurrection, or
- d. Any act of God; provided, however, that this Sub-paragraph"d" shall not apply to the first two (2) weeks of layoff resulting from such cause.
- (3) Has not refused to accept work when recalled pursuant to the Collective Bargaining Agreement and has not refused an offer by the Company of other available work at the plant.
- (4) With respect to such week the employee was not eligible for and was not claiming:
 - Any statutory or Company accident or sickness or any other disability benefit (except a benefit which he/she received or could have received while working full time); or
 - Any Company pension or retirement benefit; and
- (5) With respect to such week the employee was not in military service, or on a military leave.
- C. If an employee is ineligible for a benefit by reason of

Sub-paragraph B-(2) or B-(4) above, with respect to some but not all of his/her regular work days in a week, and is otherwise eligible for a benefit, he/she will be entitled to a reduced benefit payment as provided in Section I-C of Article II.

Furthermore, as required by EI regulations the following two points are understood:

- That the employees have no vested right to payments under the plan except to payments during a period of unemployment specified in the plan.
- That any payments in respect of guaranteed remuneration or in respect of deferred remuneration

or severance pay benefits are not reduced or increased by payments received under the plan.

Section 3. Disputed Claims for Employment Insurance Benefits

- A. With respect to any week for which an employee has applied for a benefit and for which he/she:
 - (1) Has been denied an Employment Insurance Benefit, and the denial is being protested by the employee through the procedure provided therefore under Employment Insurance, or
 - (2) Has received an Employment Insurance Benefit, payment of which is being protested by the Company through the procedure provided therefore under the Employment Insurance Act, and such protest has not, upon appeal, been held to be frivolous, and the employee is eligible to receive a benefit under the Plan except for such denial, or protest, the payment of such benefit shall be suspended until such dispute shall have been determined.
- B. If the dispute shall be finally determined in favor of the employee, the benefit shall be paid to him/her if he/she had not exhausted credit units subsequent to the week to which the Employment Insurance Benefit in dispute is applicable.

ARTICLE II AMOUNT OF BENEFITS

Section 1. Regular Benefits

A. During any single continuous period of layoff, the Regular Benefit payable to an eligible employee on or after the effective date shall be an amount which when added to his/her Employment Insurance Benefit and other compensation will equal 95% of his/her weekly after tax pay, minus \$12.50 to take into account workrelated expenses not incurred, but in no case more than \$130.00, and will be paid for the number of weeks

- equivalent to the employee's number of credit units at the time of layoff.
- **B.** An otherwise eligible employee entitled to a benefit reduced because of ineligibility with respect to part of the week, as provided in Section 2-C of Article I will receive one-fifth (I/5) of a Regular Benefit computed under Paragraphs A of this Section for each workday of the week for which he/she is otherwise eligible.

Section 2. Employment Insurance Benefit and Other Compensation

- A. An employee's Employment Insurance Benefit and Other Compensation for a week means:
 - (1) The amount of Employment Insurance Benefit received or receivable by the employee for the week, plus
 - (2) All pay received or receivable by the employee from the Company (including holiday payments, payments for scheduled vacations), and the amount of any pay which could have been earned, computed as if payable for hours made available by the Company but not worked, after reasonable notice has been given to the employee for such week; provided that if wages or remuneration from employers other than the Company or any military pay are received or receivable by the employee and are applicable to the same period as hours made available by the Company but not worked, only the greater of
 - (a) such wages or remunerationfrom other employers in excess of the greater of ten dollars (\$10.00) or twenty percent (20%) of such wages or remunerationor military pay in excess of ten dollars (\$10.00); or
 - (b) any amount of pay which could have been earned, computed, as if payable, for hours made available by the Company but not worked, shall be included;

- (3) All wages or remuneration, as defined under the Employment Insurance Act in excess of the greater of ten dollars (\$10.00) or twenty percent (20%) of such wages or remuneration received or receivable from other employers for such week (excluding such wages or remuneration which were considered in the calculation under A-(2) above); plus
- (4) The amount of all military pay in excess of ten dollars (\$10.00) received or receivable for such week excluding such military pay which was considered in the calculation under A-(2) of this Section
- B. If the Employment Insurance Benefit actually received by an employee for an Employment Insurance week shall be for less or more than a full Employment Insurance week (for reasons other than the employee's receipt of wages or remuneration for such Employment Insurance week),
 - (1) Because he/she has been disqualified or otherwise determined ineligible for a portion of his/her Employment Insurance Benefit for reasons other than those set forth in Section I-A-(2) of Article I, or
 - (2) Because of an underpayment or overpayment of a previous Employment Insurance Benefit, or the amount of the Employment Insurance Benefit which would otherwise have been paid to the employee for such Employment Insurance Benefit week shall be used in the calculation of "Employment Insurance Benefit and Other Compensation" for such Employment Insuranceweek.

Section 3. Insufficient Credit Units for a Full Benefit

If an employee has to his/her credit less than the full number of credit units required to be cancelled for the payment of a benefit for which he/she is otherwise eligible, he/she shall be paid the full amount of such benefit and all remaining credit units or fractions thereof to his/her credit shall be cancelled.

Section 4. Benefit Overpayments

- A If the Company determines that any benefit(s) paid under the Plan should not have been paid or should have been paid in a lesser amount, written notice thereof shall be mailed to the employee receiving the benefit(s) and he/she shall return the amount of overpayment to the Company; provided, however, that no repayment shall be required if the cumulative overpayment is three dollars (\$3.00) or less, or if notice has not been given with 120 days from the date the overpayment was established or created, except that no such time limitation shall be applicable in cases of fraud or willful misrepresentation.
- **B.** If the employee shall fail to return such amount promptly, the Company shall arrange to reimburse the Plan for the amount of overpayment by making a deduction from any future benefits otherwise payable to the employee, or by requesting the Company to make a deduction from compensation payable by the Company to the employee, or both. The Company is authorized to make such deduction from the employee's compensation.
- C. If the Company determines that an employee has received an automatic Short Week Benefit for any week for which he/she has received an Employment Insurance Benefit, and it was not deducted, the amount of such automatic Short Week Benefit or a portion of such benefit equivalent to the Employment Insurance Benefit, whichever is less, shall be treated as an overpayment and deducted in accordance with this Section from future benefits or compensation payable by the Company.

Section 5. Withholding Tax

The Company shall deduct from the amount of any benefit any amount required to be withheld by the Company by reason of any law or regulation for payment of taxes or otherwise to any Federal, Provincial, or Municipal Government. In determining the amount of any applicable tax entailing personal exemptions, the Company is required to rely on the official form filed by the employee with the

Company for purposes of Income Tax Withholding on regular wages.

ARTICLE III CREDIT UNITS AND DURATION OF BENEFITS

Section 1. General

Credit units shall have no fixed value in terms of either time or money, but shall be a means of determining eligibility for and duration of benefits.

Section 2. Accrual of Credit Units

- A. Credit units shall be credited at the rate of one-half (I/2) of a credit unit for each workweek for which an employee (1) receives any pay from the Company, (2) was on a military leave of absence, Maternity/Parental Leave or (3) was absent from work because of occupational injury or disease incurred in the course of such employee's employment with the Company and on account of such absence received Workplace Safety and Insurance benefits
 - (1) For the purpose of accruing credit units under this Section 2:
 - Pay in lieu of vacation shall be considered as pay for the workweek in which it is paid; and
 - **b.** Back pay shall be considered as pay for each workweek to which it may be allocable.
 - (2) No employee may have to his/her credit in the aggregate at any one time more than fifty-two (52) credit units under this Plan and under any other SUB Plan of the Company. However, any employee who has at any time to his/her credit in the aggregate a total of fifty-two (52) credit units under this Plan or under any other SUB Plan of the Company and who would otherwise accumulate additional credit units, may direct that such additional credit units shall be credited to him/her and a corresponding number of credit units

accumulated under this Plan or under any other SUB Plan of the Company, shall be cancelled as long as the aggregate of his/her credit units at any time does not exceed fifty-two (52).

- **B.** No employee shall be credited with any credit unit prior to the first day as of which he/she:
 - (1) Had at least one (1) year of seniority, and
 - (2) Is in active employment (or was in such active employment within thirty (30) days prior to such first day): but as of such day he/she shall be credited with credit units based upon his/her workweeks occurring while he/she is an employee.
- C. An employee who has credit units as of the last day of a week shall be deemed to have them for all of the week; provided, however, that an employee who has credit units during part of a week but forfeits them due to breaking seniority during such a week by reason of death or of retirement under the John Deere Pension Plan For Wage Employees shall be deemed to have credit units for all of such week.
- D. At such time as the amount of any benefit overpayment is repaid to the Plan, except as otherwise provided in the Plan, the number of credit units, if any, theretofore cancelled with respect to such overpayment of benefits shall be restored to the employee, except to the extent that such restoration would raise the number of his/her credit units at the time thereof above fifty-two (52), and except as otherwise provided with respect to credit unit forfeiture under Section 3 of this Article.
- E. An employee with ten (10) or more years of seniority at the time of layoff who is subsequently recalled, will thereafter be credited with one (1) credit unit for each workweek as provided for in Section 2 of this Article III.

Section 3. Forfeiture of Credit Units

A. An employee shall forfeit permanently all credit units which he/she has to his/her credit in a Unit if he/she:

- (1) Incurs a break in seniority in such Unit unless such a break in seniority results from:
 - (a) breaking seniority by retirement under the total and permanent disability provisions of the John Deere Pension Plan for Wage Employees and he/she shall subsequently have his/her seniority reinstated, the credit units previously forfeited shall again be credited to him;
- (2) Is on layoff from the Company for a continuous period of twenty-four (24) months, except that is at the expiration of such twenty-four (24) month period he/she is receiving benefits his/her credit units shall not be forfeited until he/she ceases to receive benefits.or:
- (3) Willfully misrepresents any material fact in connection with an application by him/her for benefits under the plan.

Section 4. Cancellation of Credit Units

- A. The number of credit units to be cancelled for any benefit shall be one per week, except that;
 - (1) No credit units shall be cancelled when an employee receives an Automatic Short Week Benefit.
- B. If an employee receives a Layoff Disability Benefit paid in accordance with the Health and Accident Insurance Program, with respect to any week, there shall be cancelled the number of credit units which would have been cancelled if he/she had received a Regular Benefit for such week. If an employee receives such Layoff Disability Benefit for a portion of a week and does not receive a Regular Benefit with respect to any part of such week, only one-half (1/2) the number of such credit units shall be cancelled for the Layoff Disability Benefit. If an employee receives a Layoff Disability Benefit under Article I, Section 2-C for such week, no credit units will be cancelled for the Layoff Disability Benefit.

Section 5. Armed Services

An employee who enters the Canadian Forces directly from the employ of the Company shall while in such service be deemed, for purposes of the Plan, to be on leave of absence and shall not be entitled to any benefit, and all credit units credited to the employee at the time of his/her entry into such service shall be credited to him/her upon his/her reinstatement as an employee. In addition, all credit units for which he/she is entitled to be credited with in respect to the period of his/her military service will be credited to him.

ARTICLE IV APPLICATION, DETERMINATION OF ELIGIBILITY, AND APPEAL PROCEDURES FOR BENEFITS

Section 1. Applications

- A. Filing of Applications
 - An application for a Benefit may be filed, either in person or by mail, in accordance with procedures established by the Company.
 - (2) No application for a Benefit shall be accepted unless it is submitted to the Company within thirteen (13) weeks after the end of the week with respect to which it is made; provided, however, that if the amount of the employee's Employment Insurance Benefit is adjusted retroactively with the effect of establishing a basis for eligibility for a Benefit or for a Benefit in a greater amount than that previously paid, he/she may apply within thirteen (13) weeks after the date on which such basis for eligibility is established.

B. Application Information

- Applications filed for a Benefit under the Plan will include:
 - In writing, any information deemed relevant by the Company with respect to other benefits received, earnings and the source thereof, dependents, and such other information as the

- Company may require in order to determine whether the employee is eligible to be paid a Benefit and the amount thereof.
- (2) With respect to a Regular Benefit, the exhibition of the employee's Employment Insurance Benefit check or other evidence satisfactory to the Company of either:
 - a. his receipt of or entitlement to an Employment Insurance Benefit, or
 - b. his ineligibility for an Employment Insurance Benefit only for one or more of the reasons specified in Section 1 of Article I; provided, however, that in the case of Employment Insurance Benefit ineligibility by reason the period worked in the week or pay received from the Company or from any other employer(s) or because of full-time employment with an employer other than the Company (Item c of Section 1-A-(2) of Article 1), Employment Insurance evidence for such reason of ineligibility shall not be required.

Section 2. Determination of Eligibility

- A. Application Processing by Company
 - (1) When an application is filed for a Benefit under the Plan and the Company is furnished with the evidence and information required, the Company shall determine the employee's entitlement to a Benefit.
- B. Notice of Denial of Benefits
 - (1) If the Company determines that an employee is not entitled to a Benefit, it shall notify him/her promptly, in writing, of the reason(s) for the determination.

ARTICLE V ADMINISTRATION OF THE PLAN

Section 1. Powers and Authority of the Company

- A. The Company shall have such powers and authority as are necessary and appropriate in order to carry out its duties under this Article, including, without limitation, the following:
 - (1) To obtain such information as the Company shall deem necessary in order to carry out its duties under the Plan.
 - (2) To investigate the correctness and validity of information furnished with respect to an application for a Benefit.
 - (3) To make initial determinations with respect to Benefits.
 - (4) To establish reasonable rules, regulations and procedures concerning:
 - The manner in which and the times and places at which applications shall be filed for Benefits, and
 - **b.** The form, content and substantiation of applications for Benefits.
 - (5) To establish appropriate procedures for giving notices required to be given under the Plan.
 - (6) To establish and maintain necessary records, and
 - (7) To prepare and distribute information explaining the Plan.
- B. Nothing contained in this Plan shall be deemed to qualify, limit or alter in any manner the Company's sole and complete authority and discretion to establish, regulate, determine, or modify at any time levels of employment, hours of work, the extent of hiring and layoff, production schedules, manufacturing methods, the products and parts thereof to be manufactured, where and when work shall be done, marketing of its products, or any other matter related to the conduct of its business or the manner in which its business is to be

managed or carried on, in the same manner and to the same extent as if this Plan were not in existence.

Section 2. Determination of Dependents

In determining an employee's dependents for purposes of Regular Benefit determinations, the Company shall be entitled to rely upon the official form filed by the employee with the Company for Income Tax Withholding purposes, and the employee shall have the burden of establishing that he/she is entitled to a greater number of Withholding Exemptions then he/she shall have claimed on such form.

Section 3. To Whom Benefits Are Payable in Certain Conditions

Benefits shall be payable only to the eligible employee, except that if the Company shall find that the employee is deceased or is unable to manage his/her affairs for any reason, any Benefit payable to him/her shall be paid to his/her duly appointed legal representative, if there be one, and, if not, the spouse, parents, children, or other relatives or dependents of the employee as the Company in its discretion may determine. Any Benefit so paid shall be a complete discharge of any liability with respect to such Benefit. In the case of death, no benefit shall be payable with respect to any period following the last day of layoff immediately preceding the employee's death.

Section 4. Nonalienation of Benefits

No Benefit shall be subject in any way to alienation, sale, transfer, assignment, pledge, attachment, garnishment, execution or encumbrance of any kind and any attempt to accomplish the same shall be void. In the event that the Company shall find that such an attempt has been made with respect to any Benefit due or to become due to any employee, the Company in its sole discretion may terminate the interest of the employee in such Benefit and apply the amount of the Benefit to or for the benefit of the employee, his/her spouse, parents, children, or other relatives or dependents as the Company may determine and any such

application shall be a complete discharge of all liability with respect to the Benefit.

Section 5. Applicable Law

The Plan and all rights and duties thereunder shall be governed, construed and administered in accordance with the laws of the Province of Ontario, except that the eligibility of an employee for and the amount and duration of Employment Insurance Benefits shall be determined in accordance with the Employment Insurance Act.

ARTICLE VI MISCELLANEOUS

Section 1. General

A. Purpose of Plan

It is the purpose of this Plan to supplement Employment Insurance Benefits and not to replace or duplicate them.

B. Receipt of Benefits

Any Benefit paid by the Company under the Plan shall not be considered a part of any employee's wages for any purpose. No person who receives any Benefit shall for that reason be deemed an employee of the Company during such period, and he/she shall not thereby accrue any greater right to participate in, accrue credits or receive benefits under any other Employee Benefit Plan to which the Company contributes than he/she would if he/she were not receiving such Benefit.

Section 2. Amendment and Termination of the Plan

A. So long as the Collective Bargaining Agreement of which this Supplemental Unemployment Benefit Plan as amended is a part shall remain in effect, the Plan shall not be amended, modified, suspended, or terminated, except as may be proper or permissible under the terms of the Plan or the Collective Bargaining Agreement. Upon the termination of the Collective Bargaining Agreement the Company shall have the right to continue

- the Plan in effect and to modify, amend, suspend, or terminate the Plan, except as may be otherwise provided in any subsequent Collective Bargaining Agreement.
- B. Upon any termination of the Plan, the Plan shall terminate in all respects except to pay benefits to eligible employees for a period of one (1) year following termination, if not sooner.

ARTICLE VII DEFINITIONS

As used herein:

- A. "Active Employment" -- An employee is in active employment in any pay period for which he/she draws pay, except that an employee on disciplinary layoff, vacation, personal leave of absence, or sick leave will, be considered as in active employment and if subject to layoff during such period his/her layoff will be considered as becoming effective at the end of such period.
- **B.** "Base Hourly Rate" (excluding all premiums and bonuses of any kind) means:
 - (1) With respect to a Regular Benefit, the employee's straight-time hourly rate on his/her last day of work; except, that
 - a. if he/she had a higher straight-time hourly rate of record within one hundred eighty (180) calendar days immediately preceding his/her last day worked, base hourly rate shall be such higher rate; or
 - b. if greater, his/her average earned hourly rate for the last four (4) pay periods, or lesser number if that is all that he/she worked, in which he/she worked, or, if higher, the employee's average earned hourly rate for the first four (4) pay periods worked subsequent to the one hundred eightieth (180th) calendar day immediately preceding his/her last day worked.
 - c. If an employee claims and it is established that he/she worked in at least four (4) consecutive pay periods within one hundred eighty (180)

calendar days immediately preceding his/her last day worked, base hourly rate shall be the employee's average earned hourly rate for the last four (4) pay periods in which he/she worked and for which he/she had any earnings or, if higher, the employee's average earned hourly rate for the first four (4) consecutive pay periods worked and for which he/she had any earnings subsequent to the one hundred eightieth (180th) calendar day immediately preceding his/her last day worked.

- I. In no case will his/her base hourly rate, as defined in b, c or d exceed one hundred fifty percent (150%) of his/her average occupational rate(s), as defined and calculated in Article XIII, Section 13.03, Computation, of the Collective Bargaining Agreement.
- (2) The base hourly rate determined under (1) above shall be adjusted to include:
 - The amount of any applicable cost-of-living allowance in effect with respect to the week for which the benefit is paid, and
 - b. With respect to benefits, the amount of any general wage increase which became effective (pursuant to the Collective Agreement) after the day or period used to establish his/her base hourly rate. In such event the amount of general wage increase shall be the amount applicable to the job classification in which the employee worked either on the day, or the last day of the period for which his/her base hourly rate was determined under (1) or (2) above. The base hourly rate adjustment due to the general wage increase shall be effective with respect to benefits which may be payable for and subsequent to the week in which such general wage increase became or becomes effective.

- **C.** "Collective Bargaining Agreement" means the currently effective Collective Bargaining Agreement of the Company which incorporates this Plan by reference.
- D. "Compensated or Available Hours" for a week shall be the sum of
 - (1) All hours for which an employee receives pay from the Company (excluding pay in lieu of vacation) with each hour paid at premium rates to be counted as one (1) hours; plus
 - (2) All hours scheduled for or made available to the employee by the Company but not worked by him/her after reasonable notice has been given to the employee (including any pay period on leave of absence); plus
 - (3) All hours not worked by the employee because of any of the reasons disqualifying a period of layoff under Section 2-B(2) and 2-B-(4) of Article 1; plus
 - (4) Provided, however, that the hours referred to in the foregoing shall include only hours in an employee's shift on his/her regularly scheduled workday(s).
- E. "Continuous Period of Layoff" means successive periods of layoff separated by less than two (2) weeks of continuous active work.
- F. "Credit Units" means the units determining duration of an employee's benefits which are credited to him/her generally by reason of his/her weeks of active service and cancelled at specified rates for the payment of certain benefits.
- G. "Dependent" means a spouse or a person recognized as a dependent under the Canadian Income Tax Act for establishing the employee's Withholding Tax Exemptions.
- **H.** "Effective Date" means the date approval of the Plan is received from the Government of Canada in Ottawa.
- "Employee" means an employee in the Bargaining Unit covered by the Plan.

- **J.** "Plan" means the Supplemental Unemployment Benefit Plan as set forth in this Appendix "D-1".
- **K.** "Factory" means a location or locations in, or out of, which an employee works.
- "Seniority" means seniority status as defined in the Collective Bargaining Agreement;
 - (1) "Break in Seniority" means any break in or loss of seniority pursuant to the Collective Bargaining Agreement;
 - (2) "Years of Seniority" solely for the purpose of Section 4-B of Article III, the term "Years of Seniority" means continuous employment as defined in Article XII, of the Collective Bargaining Agreement.
- М. "Employment Insurance" means any system or program established pursuant to Canadian law for paying benefits to persons on account of their unemployment under which a person's eligibility for benefit payments is not determined by application of a "means" or "disability" test: excluding, however, any such system or program established for the primary purpose of education or vocational training, even though such program may provide for subsistence allowances or benefits to individuals not employed while undergoing such training. "Employment Insurance Benefit" means a benefit payable under Employment Insurance, except that it does not include benefits in excess of the normal period under Employment Insurance which are payable only because the recipient is undergoing approved training.
- N. "Supplementation" means recognition of the right of a person to receive both an Employment Insurance Benefit and a Weekly Supplemental Benefit under the Plan for the same week of layoff at approximately the same time and without reduction of the Employment Insurance Benefit because of the payment of the Weekly Supplemental Benefit under the Plan.
- O. 'Week" when used in connection with eligibility for and computation of benefits with respect to an employee means a period of layoff equivalent to a workweek or a short workweek, or a workweek other than a short

workweek during which an employee is on layoff part of the week. "Short workweek" means a workweek during which an employee has less than forty (40) compensated and available hours and meets the eligibility requirement for an Automatic Short Week Benefit.

- P. "Weekly After-Tax Straight-Time Pay" means the amount of an employee's weekly straight-time pay reduced by the sum of all Federal taxes and contributions which would be required to be collected, deducted, or withheld by the Company from a regular weekly wage of such amount if paid to him/her for the last pay period he/she worked.
- Q. "Weekly Straight-Time Pay" means an amount equal to an employee's base hourly rate (as determined for a Weekly Supplemental Benefit) multiplied by forty (40) (or, in the case of part-time employee, by the number of hours he/she is regularly scheduled to work during a workweek).
- R. "Workweek" or "Pay Period" means seven (7) consecutive days beginning on Monday at the regular starting time of the shift to which the employee is assigned immediately prior to being laid off.

EXHIBIT B AUTOMATIC SHORT WORK WEEK

ARTICI FI

Section 1. Automatic Short Week Benefits

- A. The Automatic Short Week Benefit payable to an eligible employee for any week beginning on or after the effective date shall be an amount will equal the number by which forty (40) exceeds his/her compensated or available hours computed to the nearest tenth of an hour multiplied by eighty percent (80%) of his/her base hourly rate.
- **B.** An employee who breaks seniority during a week by reason of death or Retirement under the provisions of the John Deere Death, Disability and Pension Plan and is eligible for an Automatic Short Week Benefit prior to the date his/her seniority is broken, will receive an amount computed as provided in Paragraph A of this Section, based on the number by which the hours for which the employee would regularly have been compensated exceeds his/her compensated or available hours with respect to that part of the week prior to the date his/her seniority is broken.
- C. "Base Hourly Rate" with respect to an Automatic Short Week Benefit means the average straight-time hourly earnings as computed in accordance with Article XV of the Collective Bargaining Agreement.

Section 2. Eligibility for An Automatic Short Week Benefit

- A. An employee shall be eligible for an Automatic Short Week Benefit for any workweek beginning on or after the effective date if:
 - (1) During such workweek he/she had less than forty (40) compensated or available hours, and
 - a. He/She performed some work for the Company, or

- b. For such workweek he/she received some jury duty pay, bereavement pay, excused personal absence pay, vacation or military pay from the Company, or
- c. For such workweek he/she received only holiday pay from the Company and for the immediately preceding week he/she either received an Automatic Short Week Benefit or had forty (40) or more compensated or available hours, or
- d. For such workweek he/she was on a disciplinary suspension for some but not all of the workweek.
- (2) He/She had at least one (1) year of seniority as of the last day of such workweek or during some part of such workweek he/she had at least one (1) year of seniority and broke seniority by reason of death or of retirement under the provisions of the John Deere Death, Disability and Pension Plan for Wage Employees;
- (3) He/She was on a qualifying layoff, as described in Section 2 of Article I of the SUB Plan for some part of such workweek or he/she was ineligible, as defined in the Collective Bargaining Agreement, for pay from the Company for all or part of a period of jury duty, bereavement or short-term active duty of thirty (30) days or less because he/she was called to active military service or similar unit by provincial or federal authorities in case of public emergency during the workweek and during all or part of such period he/she would otherwise have been on a qualifying layoff under the Plan; and
- (4) He/She received an Employment Insurance Benefit not currently under protest by the Company or was ineligible to receive an Employment Insurance Benefit because;
 - a. His/Her pay and any pay which he/she could have secured for hours scheduled for or made available to him/her but not worked was

- sufficient to disqualify him/her for an Employment Insurance Benefit or waiting week credit:
- **b.** He/She was serving a waiting week or partial waiting week of layoff under Employment Insurance; or
- c. He/She had exhausted his/her Employment Insurance Benefit rights, and did not have prior to layoff a sufficient period of employment or earnings under Employment Insurance.
- B. No application for an Automatic Short Week Benefit shall be required of an employee. However, if an employee believes that, he/she is entitled to (1) an Automatic Short Week Benefit for a week which he/she does not receive on the date when such benefits for such week are paid, or (2)an Automatic Short Week Benefit in an amount greater than he/she received, he/she may file written application therefore within sixty (60) calendar days after such date in accordance with procedures established by the Company.
- C. An Automatic Short Week Benefit payable for a week shall be in lieu of any other benefit under the SUB Plan for that week. Furthermore, Automatic Short Work Week Benefits are not considered an approved SUB benefit under EI regulations and will therefore be considered as earnings for purposes of determining entitlement to Employment Insurance benefits.

APPENDIX "F" TUITION REFUND PLAN

Section 1. Purpose

The purpose of this Plan is to provide financial assistance to employees who take educational courses outside regular working hours on a voluntary basis for self-improvement.

Section 2. Eligibility

- A. Tuition aid will be granted to employees who have completed one (1) or more years of continuous employment. Employees who are laid off will also be considered eligible for tuition aid unless eligible for tuition aid elsewhere. An employee who is otherwise eligible but who does not have one (1) year of continuous employment will be permitted to participate in the Plan, but tuition aid will be withheld until one (1) year of continuous employment has been completed. If an employee's employment is terminated prior to completion of a course, he/she will not be eligible for tuition aid
- **B.** To be eligible, the employee may enroll only in such courses which meet the following requirements:
 - (1) They are job related; that is, they will tend to improve the employee's performance on work covered by his/her seniority classification, or;
 - (2) They will help to prepare him/her for future assignments with the Company for which he/she might reasonably be expected to qualify, or;
 - (3) They are part of a curriculum leading to a degree in a field which is job related.

Section 3. Approved Schools and Courses

A Employees will be permitted to enroll in either credit or non-credit regular extension, or correspondence courses offered by accredited and approved colleges and universities. Enrollment in high schools, technical

- schools and area community colleges will also be accepted.
- **B.** Since only off-work-hour courses are eligible for approval under the Plan, a schedule will be approved only when it is reasonably certain that the course work will not affect the employee's health or job performance, and not to exceed nine (9) credit hours per semester.

Section 4. Application for Approval

- A. An application form may be obtained from the Human Resources Department or the employee's supervisor. The form must be completed and submitted to the employee's supervisor at least one (1) month prior to the first meeting of the class.
- B. To be approved, the course must be recommended by the employee's supervisor, department head, and local administrator of the Plan.

Section 5. Amount of Tuition Aid

- A. The Company will provide tuition aid in an amount equal to sixty-six and two thirds percent (66-2/3%) of the total tuition and laboratory, registration, and activity fees expense, if any, normally required for an approved course in accordance with the following procedure:
 - (1) The Company will advance one-third (1/3) of the total tuition and laboratory, registration and activity fees to the employee upon approval of the course, and:
 - (2) Upon satisfactorily completing the course, the Company will reimburse the employee for the second one-third (1/3) of the total tuition, laboratory, registration and activity fees.
- B. Under no circumstances will tuition aid be granted for covering the costs of textbooks, materials, examination fees, or transportation. No tuition aid will be granted for courses for which any part of the expenses are met by scholarships or eligibility for Veterans.

APPENDIX "G" JOHN DEERE GROUP LIFE AND DISABILITY INSURANCE PLAN FOR WAGE EMPLOYEES

ARTICLE I ESTABLISHMENT OF PLAN

Section 1. Preamble

John Deere Welland Works of John Deere Limited, hereinafter referred to as the Company, will provide Group Life, Accidental Death, Dismemberment and Disability insurance, hereinafter referred to as the Plan, for its wage employees as hereinafter set forth.

Section 2. Cost of Benefits

The **cost** of providing benefits under this Plan will be borne by the Company and no contribution to the Pian shall be made by an employee or retired employee except **as** hereinafter provided.

Section 3. Nonencumbrance of Benefits

Assignment, pledge or encumbrance of any benefits under this Plan will not be permitted or recognized. Benefits shall not be subject to attachment, execution, garnishment, or other legal process.

Section 4. Amendment, Modification and Plan Termination

The Company may at any time amend, modify or terminate the Plan, provided, however, that no change shall reduce the amount **of** benefit **to** which an employee, retired employee or beneficiary shall be entitled in respect to claims incurred prior to the effective date of such change.

Section 5. Funding

Except as otherwise specifically provided, benefits shall be provided through an insurance company selected by the

Company, a fund established by the Company, or from the general assets of the Company.

Section 6. Application for Benefits

In order for any employee or beneficiary to receive benefits pursuant to any provisions of this Plan, such employee or beneficiary must file written application supplying information required to verify such application. The necessary form(s) for such purpose will be supplied by the Company.

Section 7. Denial of Benefits

- A. When application for benefits provided by this Plan is denied in full or in part, the employee or beneficiary shall receive written notice of the reason or reasons for such denial
- B. A request for review of the denial of benefits must be submitted within sixty (60) days of such denial, in writing, to the Company. The Company will reply to such request within sixty (60) days.

ARTICLE II ELIGIBILITY AND AMOUNT OF BENEFIT

Section 1. Eligibility of Employees and Commencement of Insurance

- A. Except as hereinafter provided, an employee
 - (1) hired prior to 1 October 1995 shall be eligible and insurance shall commence on such date, except that if an eligible employee is not actively at work on 1 October 1995, the employee shall not be insured under this Plan until the date such employee returns to work:
 - (2) who is eligible, but not actively at work on 1 October 1995 shall be covered under the terms and conditions of the Group Life and Disability Insurance provided by the 1987 Plan if the employee is otherwise eligible for such insurance;

- (3) hired on and after 1 October 1995 shall be eligible and insurance shall commence on the date the employee first reports for work.
- (4) hired on and after 1 October 1998 shall be eligible on the first of the month following the month seniority is attained, provided the employee is actively at work that day. Otherwise, on such later date that the employee first reports for work on a full-time basis.

Section 2. Group Life Insurance

- A. Prior to retirement, an employee will be covered by the following life insurance:
 - (1) Less than one month of continuous employment

a.	Basic Life insurance	\$ 5,000

- **b.** Accidental Death insurance \$4,000
- c. Accidental Dismemberment insurance loss of one member \$2,000 loss of two members \$4.000
- (2) At least one month but less than one year of continuous employment

a.	Basic Life insurance	\$10,000

b. Accidental Death insurance \$ 9,000

c. Accidental Dismemberment insurance

loss of one member \$ 4,500 loss of two members \$ 9,000

- (3) One or more years of continuous employment:
 - Basic Life insurance: one times Earnings rounded up to the next \$100, but not greater than \$50.000.
 - b. Accidental Death insurance: one times Earnings rounded up to the next \$100, but not greater than \$49,000.

c. Accidental Dismemberment insurance:

loss of one member: one-half (1/2) times Earnings rounded up to the next \$50, but not greater than \$24,500.

loss of two members: one times Earnings rounded up to the next \$100, but not greater than \$49,000.

B. For the purposes of determining the amount of benefit in this Section, the Earnings for each covered employee shall be the Earnings as determined in Appendix C, Article III, Section 1, multiplied by two-thousand one hundred & ninety (2,190).

Section 3. Accidental Death and Dismemberment Insurance

- A. If death results within one year from a nonoccupational injury, there will be paid to the beneficiary of the insured employee an additional amount equal to the Accidental Death Insurance as set forth in Section 2 of this Plan.
- B. If death results within one year from an occupational injury, there will be paid to the beneficiary of the insured employee an additional amount equal to one-half of the Accidental Death Insurance as set forth in Section 2 of this Plan.
- C. Accidental dismemberment benefits will be paid for dismemberments occurring within one year of the date of the accident as follows:
 - (1) The Accidental Dismemberment Insurance-Two Members as set forth in Section 2 of this Plan is payable for the loss of both hands, both feet, one hand and one foot, sight of both eyes, either hand or foot and the sight of one eye.
 - (2) The Accidental Dismemberment Insurance-One Member as set forth in Section 2 of this Plan is payable for the loss of one hand, one foot, or the sight of one eye.
- D. Loss shall mean, with regard to hands and feet, dismemberment by severance through or above wrist or

- ankle joints; with regard to eyes, entire and irrecoverable loss of sight.
- E. Only one of the amounts so specified above (A, B, C-(1) or C-(2)), the largest, will be paid for all injuries resulting from any one accident.

F. Exclusions

The insurance under this Section shall not cover any loss

- (1) caused or contributed to by bodily or mental infirmity, disease or medical or surgical treatment therefore, or infection (except pus-forming infection which shall occur through an accidental cut or wound), even though the proximate and precipitating cause of the loss is accidental bodily injury;
- (2) caused or contributed to by war or any act of war, whether war is declared or not, or by any act of international armed conflict, or conflict involving armed forces of any international authority:
- (3) resulting from suicide or any attempt thereat (sane or insane).

Section 4. Total and Permanent Disability

A. An employee who becomes totally and permanently disabled prior to attainment of age 69 (an employee shall be deemed to be permanently and totally disabled when, on the basis of medical evidence satisfactory to the Company, he/she is found to be totally and permanently prevented from performing any and all work for the Company as a result of any medically demonstrable and determinable physical or mental condition resulting from bodily injury or mental or physical disease, either occupational or non-occupational in cause) and who does not qualify for normal, early or disability retirement benefits under the John Deere Pension Plan for Wage Employees, will receive the Basic Life Insurance in 50 monthly installments at the rate of \$20 per each \$1,000 of life insurance in lieu of the death benefit. The first of such installments shall be payable as of (1) the first day

of the month after the submission of the required proof of such disability and monthly thereafter, or (2) the first of the month coinciding with or immediately following the date the employee has ceased to be eligible to receive weekly disability benefits and monthly long-term disability benefits, if any, as the case may be, whichever is later.

- **B.** If an employee should die while the monthly installments are being paid, the remaining installments will be paid to the beneficiary in a lump sum. The lump sum shall not be less than \$1,500. If an employee should die after all the installments have been paid, the beneficiary will receive \$1,500.
- C. If after payment of installments to which an employee is entitled under A above, such employee shall return to active work with the Company, the amount of insurance shall be the amount to which the employee is entitled under the provisions of the Planthen in effect.

Section 5. Transition Survivor Income Benefit and Bridge Survivor Income Benefit

Upon the death of an active employee who was at work A. on or after 1 January 1971, and who had at least one month of continuous employment, who was covered for life insurance, or upon the death prior to age 55 of an employee who was retired under the provisions of Article III. Section 2, of the John Deere Pension Plan for Wage Employees, a Transition Survivor Income Benefit is payable to the survivor or survivors as defined herein provided there are such survivors living to receive it. commencing on the first day of the month following the death of the employee and continuing for not more than 24 months. The Transition Survivor Income Benefit is also payable to the survivor or survivors of an employee retired prior to 1 January 1992 on total and permanent disability pension under Section 3 of Article III, whose disability pension has not been redetermined in accordance with Section 3-D of Article III of the John Deere Pension Plan for Wage Employees, or of an employee who retired after 1 January 1992 on total and

permanent disability pension who has not attained age 65 and who dies while insured by Group Life Insurance under this Program. The Transition Survivor Income Benefit shall be \$350 for any month the eligible survivor or survivors are not eligible for any survivor's, disability or retirement benefit under the Canada Pension Plan or any Old Age Security benefit, as now in effect or as hereafter amended and otherwise \$200. The Transition Benefit shall be reduced by the survivor's pension benefit entitlement as provided under Article IV of the John Deere Pension Plan for Wage Employees. For months in which two or more survivors share a benefit, each survivor's share is computed as a fraction of the benefit that would be paid to a sole survivor according to such survivor's own eligibility for benefit under the Canada Pension Plan or any Old Age Security benefit as now in effect or as hereafter amended.

- (1) For purposes of this paragraph
 - a A "Class A Survivor" means:
 - I. the surviving spouse of a deceased employee, but only if the surviving spouse was legally married to the deceased employee for at least one year immediately prior to the employee's death; or
 - **ii.** the surviving common law spouse of the deceased employee who had been living with the employee for at least three years immediately prior to the employee's death; or
 - **III.** the surviving common law spouse of the deceased employee who had been living with the employee in a relationship of some permanence and the employee and the surviving common law spouse had been the natural or adoptive parents of a child.
 - b. A "Class B Survivor" means the employee's child who at the employee's death and at the time a survivor income benefit first becomes payable to such child is both unmarried and either (1) under 21 years of age or (2) at least

age 21 but under age 25 or (3) totally and permanently disabled at any age over 21; provided, however, that a child under clause (2) or (3)must have been legally residing with and dependent upon the employee at the time of his/her death. A child ceases to be a Class B Survivor upon marrying or if not totally and permanently disabled, upon reaching his/her 25th birthday. To qualify as the employee's child, the child must be one of the following:

- The employee's own child born prior to the first of the month following the employee's death
- ii. The employee's legally adopted child or a child with respect to whom the employee had initiated legal adoption proceedings which were terminated by death.
- iii. The employee's stepchild who resided with the employee at the time of death.
- c. A "Class C Survivor" means a parent of the deceased employee for whom the employee had, during the calendar year preceding the employee's death, provided at least 50 percent of the parent's support.
- (2) This Transition Survivor Income Benefit shall be paid as follows:
 - a. If the employee is survived by a Class A Survivor, the monthly income shall be payable to such survivor. If the employee is not survived by a Class A Survivor, the monthly income shall be payable in equal shares to the employee's Class B Survivors, but if the employee is not survived by a Class B Survivor, in equal shares to the employee's Class C Survivors;
 - b. If a Class A Survivor dies while monthly income payments are still payable, any remaining payments will be made, in equal shares, to the employee's then surviving Class B Survivors,

- but if none are then surviving, in equal shares to the employee's then surviving Class C Survivors but if none are then surviving, no further monthly income payments shall be made:
- c. If a Class B Survivor dies while monthly income payments are still payable, and if another Class B Survivor(s) is still alive, the living Class B Survivor(s) Benefit(s) shall be redetermined according to each Class B Survivor(s) eligibility for Social Insurance Benefits.
- d. If a Class B Survivor dies while monthly income payments are still payable, and is not survived by another Class B Survivor, any remaining payments will be made, in equal shares, to any Class C Survivors then surviving, but if no Class C Survivor is then surviving, no further monthly income payments shall be made;
- e. If a Class C Survivor dies while monthly income payments are still payable, and is survived by another Class C Survivor, the monthly amount which the living Class C Survivor had been receiving shall be redetermined according to the Class C Survivor's eligibility for Social Insurance Benefits. and
- f. If a Class C Survivor dies while monthly income payments are still payable, and is not survived by another Class C Survivor, no further monthly income payments shall be made.
- **B.** A Bridge Survivor Income Benefit of \$350 per month reduced by a survivor's pension benefit entitlement as provided under Article IV of the John Deere Pension Plan for Wage Employees and by a survivor's or disability benefit under the Canada Pension Plan as now in effect or as hereafter amended, is payable to the Class A Survivor as defined in Section 5-A above who was 45 years of age or more on the date of the employee's or retired employee's death or whose age (determined to the nearest month as of the date of the employee's or retired employee's death) when combined

with the employee's or retired employee's service credit, including partial year's service credit, as of the date of death, totals 55 or more and who has received 24 monthly payments of the Transition Survivor Income Benefit provided in Section 5-A.

- (1) The Bridge Survivor Income Benefit will become payable commencing with the first month following the month for which the 24th monthly payment of the Transition Survivor Income Benefit is paid.
- (2) The Bridge Survivor Income Benefit will not be paid beyond the earlier to occur of the following:
 - (a) the death of the Class A Survivor or
 - (b) attainment by the Class A Survivor of such age at which Old Age Security benefits become payable other than on a "needs" basis under any Federal or Provincial legislation as now in effect or bereafter amended
- (3) The Bridge Survivor Income Benefit will also be payable as outlined above to the Class A Survivor as defined in Section 5-A above who is not eligible above, who has received 24 monthly payments of the Transition Survivor Income Benefit provided in Section 5-A, and who is severely disabled. A Class A Survivor shall be deemed severely disabled if, at the date of the employee's death and on the date any Bridge Benefit becomes payable, the survivor is suffering from a chronic disease, impairment, or deformity which wholly prevents the survivor from performing normal home responsibilities or any gainful work.

Section 6. Continuance of Insurance

- A. An employee who retires under Section 1 or Section 4 of Article III of the John Deere Pension Plan for Wage Employees shall have the Basic Life Insurance reduced as provided in Section 7.
- B. An employee who retires under Section 2 of Article III of the John Deere Pension Plan for Wage Employees shall have the Basic Life Insurance continued until attainment

- of age 65 in the amount that was in force at the time of the retirement and thereafter the insurance will be reduced as provided for in Section 7.
- C. An employee who retires under Section 3 of Article III of the John Deere Pension Plan for Wage Employees shall have the Basic Life Insurance continued at the value it bore on the last day worked until the employee's disability pension is redetermined as provided in Section 3-D of Article III of the John Deere Pension Plan for Wage Employees, and thereafter the insurance will be reduced as provided for in Section 7.

Section 7. Reduction of Insurance

- A. 1. On the first day of the calendar month following the month in which an employee reaches age 65 or retires, whichever is later, but in no event later than attainment of age 69, Accidental Death and Dismemberment coverage shall terminate. The remaining Group Life Insurance in force shall be reduced by 2 percent and further reduced by the same amount the first day of each succeeding month to a minimum of 1-1/2 percent of the insurance in force at age 69 or the date of retirement (if earlier), multiplied by the years of service credit at age 69 or the date of retirement (if earlier), but in no case will the amount be reduced to less than \$5.000.
 - 2. Employees may make a one-time election to opt for a non-taxable death benefit equal to the lesser of the benefit entitlement determined in accordance with the formula outlined in Section 7.A(1) of this Article and \$10,000, in lieu of the taxable group life insurance benefit provided in accordance with Section 7.A(1).

Section 8. Termination of Insurance

- A Except as provided in Sections 6 and 7 above, insurance provided for an employee in this Article shall terminate as follows:
 - (1) Upon discharge or quitting, insurance will terminate 31 days from date employment terminates.

- (2) Upon leave of absence except for Maternity/Parental Leave, insurance will continue for the remainder of the month in which the leave became effective plus 31 days. During the balance of such leave the employee may continue the insurance in force by payment of 60¢ per month per \$1,000 of Group Life Insurance in force, but for no more than twelve months.
- (3) In case of layoff, insurance will continue for the remainder of the month in which the layoff commenced plus 31 days plus:
 - a. For employees under SUB plans, a period of time which will be determined by the number of SUB credit units credited to the employee at the date of the layoff. One month's coverage will be provided for each four credit units the employee is credited with as of the date of layoff but not to exceed twelve months.
 - For employees not under SUB plans, one b. month's additional coverage for each year of service credit, but not to exceed twelve months. Following the expiration of the maximum number of months for which the insurance is continued without cost to the employee, the employee may continue the insurance for up to an additional twelve months by paying 60¢ per month for \$1,000 of Group Life Insurance in force provided that such employee does not qualify for replacement coverage under another employer-sponsoredor group benefit plan. Ifa laid-off employee, whose insurance had not terminated is employed by another unit of the Company, the employee shall be covered by the amount of Group Life and Disability Insurance in effect with the prior employing unit of the Company.
- (4) In case of sickness or injury which has not resulted in total and permanent disability, insurance coverage will continue for the period of sickness or injury or for a period equal to the amount of the

- employee's continuous employment, whichever is the lesser, but in any event not less than 52 weeks.
- (5) Upon entering Military Service, insurance will continue for the remainder of the month in which the leave becomes effective plus 31 days.
- (6) Upon absence from work for any other reason, insurance will terminate after 31 days from date employment terminates.
- (7) An employee whose insurance is terminated under Paragraph (2) above and who retires under Sections 1, 2, 3 or 4 of Article III of the John Deere Pension Plan for Wage Employees will have \$1,500 insurance upon such retirement.
- (8) An employee whose insurance is terminated under Paragraph (2) or (3) above and is unable to return to work when recalled because of sickness or injury will have insurance coverage reinstated tor a period of time as provided in (4) above.
- (9) An employee whose insurance is terminated under Paragraph (3) or (4) above and who retired under Sections 1, 2, 3 or 4 of Article III of the John Deere Pension Plan for Wage Employees will have insurance coverage reinstated upon such retirement

Section 9. Exclusions & Limitations

- A. Temporary employees, namely (1) employees hired for a specific job with the understanding that the work is temporary and that they will not be continued in the employment of the Company after the job is completed or (2) employees hired for a specific period of time of less than three months, and
- B. Part-time employees shall not be entitled to the insurance provided for in this Plan.
- C. In the event of any payment to the employee under the Group Life and Disability Plan for loss of income for which the employee may have a cause of action against a third party the Company & its Carrier will have their

joint interest subrogated in this regard. This will entitle the Company & its Carrier to be reimbursed for any amount that the employee recovers for loss of income from any third party which exceeds the employee's actual loss of income.

D. The employee will execute and deliver such instruments and papers as may be required and do whatever else is necessary to secure such rights. The employee may take no action which may prejudice the subrogated rights.

The subrogated rights referred to in this paragraph do not apply to an individual plan purchased by the employee specifically for wage loss replacement.

Section 10. Designation of Beneficiary

- **A.** The beneficiary shall be designated by the employee in writing and filed with the Company.
- **B.** The right to change of beneficiary is reserved to the insured employee.
- C. An employee, insured under the provisions of this Plan, may at any time designate a new beneficiary for the indemnity for loss of life by filling with, and on forms provided by, the Company a written request for such change. Upon receipt of such request by the Company, the change shall take effect as of the date the employee signed such request.
- D. Upon receipt of such request by the Company the change shall relate back to and take effect as of the date the employee signed such request whether or not the employee is living at the time of the receipt of such request but without prejudice to the Company on account of any payment made by it before such request shall have been received.

Section 11. Privilege of Obtaining Individual Insurance

A. In case of the termination of any insurance under this Plan due to termination of an employee's employment, such employee shall be entitled to have issued without further evidence of insurability upon application within thirty-one days after such termination of employment, and upon the payment of the premium applicable, an individual policy of life insurance. Such policy will become effective following 31 days after the date employment terminates. Such individual policy shall be without disability or other supplementary benefits in one of the forms, except Term Insurance, customarily issued and in an amount equal to or, at the option of the employee, less than, the amount of such terminated insurance.

- B. Any employee, eligible for benefits specified in Section 5 of this Plan on the day immediately preceding the first date such employee is entitled to make application for an individual policy as outlined in A above, may increase such amount of insurance by an amount equal to the total survivor income benefit that would have become payable with respect to such employee under the provisions of Section 5 of this Article if such employee's death had occurred on such day.
- C. Information as to the coverage available and premium rates can be obtained from the Company when insurance terminates

Section 12. Physical Examinations

The Company, at its own expense, shall have the right to require an employee to submit to an examination by an independent physician designated by it for the purpose of determining continuing disability. The Company shall have the right to examine the employee as often as it may reasonably require.

Section 13. Common Law Spouse

The Term "common law spouse" as used in the Plan with respect to an employee means a person who is residing and cohabiting with the employee, provided that:

the employee is not legally married to any other person;

- (2) the person:
 - (i) has been living with the employee for at least three years, or
 - (ii) has been living with the employee in a relationship of some permanence if the employee and this person are the natural or adoptive parents of a child;

APPENDIX "H" JOHN DEERE PROFIT SHARING PLAN

Section 1. Type of Plan and Purpose

- A. This Plan is a profit sharing plan. The purpose of the Plan is to provide contingent benefits to employees to reflect their efforts in contributing to the profitability of the Company and to serve as an incentive for the employees further to contribute to the continued and further financial success of the Company and to its ability to provide continued employment opportunities to its employees.
- **B.** It is understood and agreed by all parties hereto that the duty of the Company, its Board of Directors, and the management they select is to provide the Company's shareholders protection of, and a maximum return on, their investment, consistent with retention in the business of such profits as the Board of Directors of the Company deems prudent, and with fair and competitive prices, wages, benefits and other terms of employment; no provision of this Plan shall be construed as altering that objective or in any way limiting management or such Board of Directors in the performance of their duties.

Section 2. Definitions

- A. "Company" means Deere & Company and its various U.S. and Canadian subsidiaries and affiliates that adopt or have adopted the Plan.
- B. "Effective date" of the Plan means 29 October 2007.
- C. "Employee" means any person who on or after the effective date is in the regular full-time employ of the Company and is employed for work on the prevailing schedules of the department to which he/she is assigned, and who is included in a group to whom the Plan has been made available by the Collective Bargaining Agreement or by extension by the Company and includes any such person while absent from work under circumstances which do not break continuity of service.

- D. "Company Service" means the total period elapsed subsequent to an Employee's first date of hiring as an employee by the Company, exclusive of any period during which an employee is not, or was not, in active service as an employee of the Company (whether resulting from discharge, suspension, resignation or quit, or any other cause) except that there shall be included in such total period of service all periods of absence pursuant to leave of absence granted by the Company, all periods of layoff after the employee's last date of hiring as an employee by the Company and for up to twelve (12) months of each prior period of layoff which commenced after 1 December 1976. Upon reemploymentfollowing any break in service,
 - (a) prior service shall be reinstated regardless of duration of such break in service in accordance with the preceding sentence; and
 - (b) if the employee is reemployed by the Company after 1 December 1976 within one year from the day he/she last performed an hour of service for the Company, he/she will receive additional service for his/her period of absence from active service with the Company equal to the lesser of
 - (i) his/her period of absence, or
 - (ii) one year less any period of absence otherwise credited.

The term "first date of hiring" means the first day an employee performs one hour of service with the Company. The term "break in service" means the period which begins on the date he/she severs his/her service with the Company and ends if an employee is reemployed by the Company on the first day the employee performs one hour of service following such reemployment. An employee "severs his/her service" on the date he/she quits, retires, is discharged, dies, or otherwise terminates his/her employment with the Company. The records of the Company with respect to an employee's service will be conclusive unless shown to the Plan Administrator's satisfaction to be incorrect.

- E. "Participant" means any employee who becomes eligible to be covered by the Plan pursuantto Section 3-A.
- **F.** The "plan year" is the Company's fiscal accounting year.
- G. The "fiscal accounting year" is the 12-month period used by the Company to conduct its official business for internal and external reporting purposes. The dates for the fiscal years covered by this agreement are as follows:

Fiscal 2008: 29 October 2007 through 2 November 2008

Fiscal 2009: 3 November 2008 through 1 November 2009

Fiscal 2010: 2 November 2009 through 31 October 2010

Section 3. Eligibility and Participation

- A Each employee of the Company who is a member of a group of employees to whom the Plan has been extended by the Company or by any applicable Collective Bargaining Agreement then in effect shall be eligible to be covered by the Plan and become a participant as of the later of the effective date or the first day of any payroll period on or after the date he/she has completed one or more years of Company service.
- **B.** Any participant shall be eligible for a profit sharing benefit under the Plan for any plan year which commences after 29 October 2007 provided that he/she is an active employee of the Company on the last day of that plan year or is on leave of absence or layoff from the Company on the last day of that plan year; except that any otherwise eligible employee who died, retired, or was employed at a facility of the Company which was sold during such year shall also be covered as if he/she were an active employee on the last day of that plan year.

Section 4. Amount of Benefit

A. The amount of the benefit which shall accrue for a participant for any Plan Year shall be computed by multiplying the following three elements:

- (a) the number of hours worked in that Plan Year by the participant;
- (b) the straight-time hourly rate of pay plus any cost of living allowances as of the last day of the Plan Year (or as d the last day of active work d the employee if earlier); and
- (c) the Profit Sharing Benefit Percent(s) determined in Paragraph B below times 50%.

The total benefit will equal the addition of the 50% benefit determined from the combined Worldwide Agricultural and Worldwide Commercial & Consumer Equipment rate schedule plus the 50% benefit determined from the Worldwide Consolidated rate schedule.

A participant's number of hours worked shall include only the actual number of hours that the participant was actively at work in such fiscal accounting year and shall not include any hours for which the participant was paid but during which he/she performed no work for the Company; except that

- (i) in the case of a Union representative who was exercising the privileges and/or performing the legitimate duties of his/her office, there shall also be counted the hours for which he/she was paid wages pursuant to the collective agreement; and
- (ii) there shall also be counted any hours for which a participant is awarded back pay as a result of a final resolution of a grievance pursuant to the Collective Bargaining Agreement. The applicable "profit percentage" for any plan year shall be determined as follows:
 - Divide (a) Combined Worldwide Agricultural and Worldwide Commercial & Consumer Equipment profit by (b) the sum of the Worldwide Agricultural and Worldwide Lawn & Grounds Care Business Segment month end asset balances from October to October, excluding November divided by twelve (12). Assets are defined in accordance with Financial

Accounting Standards Board Statement 14 and represent the identifiable assets of the Worldwide Agricultural and Worldwide Commercial & Consumer Equipment operations.

2. Divide (a) Worldwide Consolidated profit by (b) the sum of the Worldwide Consolidated month end asset balances from October to October excluding November divided by twelve (12). Assets are defined as the Equipment Operations assets with the Financial Services subsidiaries included on an equity basis.

Profits in calculations 1 and 2 are the total net income after taxes before deducting:

- bonus payments to salaried employees (net of tax),
- 2. profit sharing benefits under this plantor benefits under any similar profit sharing planton (net of tax); and
- extraordinary gains or losses, as defined under generally accepted accounting principles (net of tax).

For purposes of 1, 2, and 3 above, taxes shall be calculated using the effective tax rate determined from the respective statements of income. The profit and assets values in this section shall be prepared using generally accepted accounting principles. The profit percentages and underlying profits and assets shall be audited by the Company's independent certified public accountants, whose opinion shall be final and conclusive. The amount of benefit shall not be a part of the participant's hourly rate of pay and shall not be included in the calculation of any amount of other pay, allowance or benefit under any other agreement or plan covering such participant.

B. Payment Schedule

The following schedule shall be used to determine the applicable Profit Sharing Benefit Percent.

World Consoli Return on	idated	Profit Sharing Benefit	World Agricultu World Comme Consu Equipo Return o	Profit Sharing Benefit	
At Least	But Less Than	Percent	At Least	But Less Than	Percent
0.00%	0.00% 2.92		0.00%	2.09	0.00%
2.92	3.10	0.12	2.09	2.30	0.12
3.10	3.29	0.25	2.30	2.52	0.25
3.29	3.47	0.37	2.52	2.73	0.37
3.47	3.65	0.50	2.73	2.95	0.50
3.65	3.84	0.62	2.95	3.16	0.62
3.84	4.02	0.75	3.16	3.38	0.75
4.02	4.20	0.87	3.38	3.59	0.87
4.20	4.39	1.00	3.59	3.81	1.00
4.39	4.57	1.12	3.81	4.02	1.12
4.57	4.75	1.25	4.02	4.24	1.25
4.75	4.94	1.37	4.24	4.45	1.37
4.94	5.12	1.50	4.45	4.67	1.50
5.12	5.30	1.62	4.67	4.88	1.62
5.30	5.48	1.75	4.88	5.10	1.75
5.48	5.67	1.87	5.10	5.31	1.87
5.67	5.85	2.00	5.31	5.53	2.00
5.85	6.03	2.12	5.53	5.74	2.12
6.03	6.22	2.25	5.74	5.96	2.25
6.22	6.40	2.37	5.96	6.17	2.37
6.40	6.58	2.50	6.17	6.39	2.50

6.58	6.77	2.62	6.39	6.60	2.62
6.77	6.95	2.75	6.60	6.82	2.75
6.95	7.13	2.87	6.82	7.03	2.87
7.13	7.32	3.00	7.03	7.25	3.00
7.32	7.50	3.12	7.25	7.46	3.12
7.50	7.68	3.25	7.46	7.68	3.25
7.68	7.86	3.37	7.68	7.89	3.37
7.86	8.05	3.50	7.89	8.11	3.50
8.05	8.23	3.62	8.11	8.32	3.62
8.23	8.41	3.75	8.32	8.54	3.75
8.41	8.60	3.87	8.54	8.75	3.87
8.60	8.78	4.00	8.75	8.97	4.00
8.78	8.96	4.12	8.97	9.18	4.12
8.96	9.15	4.25	9.18	9.40	4.25
9.15	9.33	4.37	9.40	9.61	4.37
9.33	9.51	4.50	9.61	9.83	4.50
9.51	9.70	4.62	9.83	10.04	4.62
9.70	9.75	4.75	10.04	10.25	4.75
9.75	and above	4.87	10.25	and above	4.87

Section 5. Payment of Benefit

A Any benefit amount which is payable for any year shall be paid to an eligible participant not later than 15 January of the year following the fiscal year for which the benefit amount is computed. The amount of such benefit shall be paid by a separate payment less union dues and required withholding for federal and provincial taxes. In the event a benefit amount is calculated to be less than \$10, the amount of such benefit amount may be included with, and if included shall be identified on, the employee's Earnings Statement which is paid during the payroll period in which such benefit amount is due and payable.

B. If a participant is deceased at the time any benefit is payable to him, the amount of such benefit shall be payable to the same person or persons and in the same proportionate mount as shall be payable to the beneficiary or beneficiaries designated under the John Deere Group Life and Disability Insurance Plan.

Section 6. Miscellaneous

- A. Administration of the Plan. Except as otherwise expressly provided in an applicable Collective Bargaining Agreement, the Pian shall be administered by the Company who shall be the Plan Administrator and shall be authorized to
 - (a) determine all questions arising in the administration of the Plan,
 - **(b)** establish rules and proceduresto carry out its duties and responsibilities,
 - (c) delegate such duties and responsibilities to employees of the Company, and
 - (d) do all other acts which in its judgment are necessary for the proper administration of this Plan.
- B. Facility of Payment. If the Plan Administrator shall receive satisfactory evidence that any participant or other person entitled to receive a benefit under this Plan is physically or mentally incompetent to receive such benefit and to give a valid release therefore, the Plan Administrator shall have the discretion to make payment in one or more of the following ways:
 - (a) directly to such participant or person,
 - (b) to his/her legal guardian or conservator, or
 - (c) to his/her spouse or to any other person to be expended for his/her benefit.

The decision of the Pian Administrator shall be in each case final and binding on all persons in interest.

C. <u>Employment Rights</u>. Participation in the Plan will not give any employee of the Company any right to be retained in the service of the Company nor any right to

- claim any benefit under the Plan unless such right or claim has specifically accrued under the terms of the Plan.
- D. Nonassignability. The interests of participants and their beneficiaries under the Plan are not in any way subject to their debts or other obligations and may not be voluntarily or involuntarily sold, transferred or assigned by them except with respect to indebtedness owing to the Company. Any amount alienated or assigned to the Company shall not exceed 10% of the amount payable under the Plan to the participant or beneficiary and such alienation or assignment shall be revocable at any time by the person making such alienation or assignment.
- E. <u>Denial of Benefits</u>. In the event that a conflict of opinion arises between any employee and the Company with respect to the payment of benefits under this Plan, such conflict shall be resolved according to the provisions of the Grievance Procedure as set out in the Collective Bargaining Agreement.
- F. <u>Action by the Company</u>. Any action required or permitted to be taken by the Company hereunder may, except as otherwise expressly provided, be taken by persons designated by the Company.
- G. Gender and Number. Where the context admits, words in the masculine gender shall include the feminine gender, the plural shall include the singular, and the singular shall include the plural.
- H. Waiver of Notice. Any notice required under the Plan may be waived by the person entitled thereto.
- Limitation of Liability. To the extent permitted by law, neither the Plan Administrator, nor the Company nor any director, officer, or employee of the Company, shall have any personal liability of any nature for any act done α omitted to be done in good faith, under or in connection with the Plan.
- J. <u>Amendment and termination of Plan.</u> Deere & Company shall have the power at any time and from time to time, by action of its board of directors, to amend or terminate this Plan; provided, however, that no amendment or

termination, under any circumstances, with respect to any employees that are represented by a collective bargaining unit may be adopted without the consent of the appropriate collective bargaining representative during the term of the applicable collective bargaining agreement which extends this Plan to such employees.

APPENDIX "I" JOHN DEERE SAVINGS PLAN FOR WAGE EMPLOYEES

Section 1. Purpose

The purpose of the John Deere Savings Plan for Wage Employees (the 'Plan') is to make available to eligible employees of John Deere Limited (the 'Company') a means of savings through regular payroll deductions.

There is no guarantee under the Plan against loss because of fluctuations in the value of savings in any Registered Retirement Savings Plan as defined in Section 146 of the Income Tax Act (Canada) (the 'RRSP'). In seeking the benefits of participating in any RRSP, each employee must also accept the risks.

Section 2. Eligibility

The Plan is available to all full-time wage employees of the Company who:

- (a) have attained the age of majority as determined by the law of their place of residence;
- **(b)** have attained seniority with the Company;
- (c) are compensated in whole or in part on a Canadian payroll of the company

Section 3. Enrollment

An eligible employee may, at his/her election, enroll as a participant in the Plan by obtaining from Deere Direct Canada and completing and delivering to that office (1) a Payroll Deduction Authorization Form, (2) an Opening and RRSP Account Form, and (3) any other forms as are necessary to authorize the transfer of payroll deductions to an RRSP selected by the participant. Upon receipt by Deere Direct Canada of such forms, payroll deduction will be effective on the first Monday of the following month and continue until death or other termination of employment with the Company or other termination of eligibility, or until the

participant's written notice **of** termination is received by Deere Direct Canada.

Section 4. Payroll Deductions

Each participant specifies in the payroll deduction authorization the amount to be withheld from his/her Earnings, which may be 1% to 25% of Earnings, in one-percentage point increments. Payroll deduction authorizations remain effective until changed or terminated as provided below.

"Earnings" mean straight-time earnings, overtime pay, shift premium pay, jury duty pay, military pay, ISB (income security benefits), call-in pay, productivity bonus, CIA (continuous improvement award) and CIPP reserve payments received by a participant.

A participant may increase or decrease his/her payroll deduction in one percentage point increments of Earnings, specified above, by telephone call to Deere Direct Canada or written request to Deere Direct Canada but not more frequently than once in any one month period. A change of payroll deduction will be effective on the first Monday of the following month upon receipt of participant's notification by Deere Direct Canada.

Section 5. Registered Retirement Savings Plans

The RRSP selected by the participant for the deposit of payroll deductions must be one of the RRSP's offered by the Toronto Dominion Bank. Upon receipt by Deere Direct Canada of properly completed forms, funds will be forwarded promptly after each pay period for deposit under the terms of such RRSP. The Company reserves the right to designate, at any time, a different bank or trust company to which the Company will deliver RRSP deposits. It is the employee's responsibility to monitor their RRSP contributions and available RRSP contribution limits. The Company is not responsible for any costs or penalties which the employee

incurs as a result of the employee making contributions in excess of such limits.

Section 6. Voluntary Termination of Participation in Plan

A participant may terminate his/her participation in the Plan at any time by delivering to Deere Direct Canada, a Notice to Terminate Payroll Deduction Form, as provided by Deere Direct Canada. Such termination shall become effective as soon as practicable after receipt of participant's notification by Deere Direct Canada.

Section 7. Automatic Termination of Participation in Plan

Participation in the Plan and payroll deduction authorizations terminate automatically without notice upon death or other termination of employment with the Company or other termination of eligibility.

Section 8. Amendment or Termination of Plan

The Company reserves the right to amend or terminate the Plan at any time that such action is deemed advisable by the Company. No such amendment or termination of the Plan will result in the forfeiture of any amounts deducted from the earnings of any participant.

74b

_																				
JA s	NUA M	ARY T	w	т	F	s		BRI	JAR T	Y	_	_			RC M		w	.	F	s
-				1 .	2	3	S 1	2	3	4	T 5	F 6	S 7	S 1	2	3	4	T	6 .	7
4 11	5 12	6 13	7 14	8 15	9 16	10 17	8 15	9 16	10 17	11 18	12 19	13 20	14 21	8 15	9 16	10 17	11 18	12 19	13 20	14 21
18	19	20	21	22	23	24	22	23	24	25	26	27	28	22	23	24	25	26	27	28
25	26	27	28	29	30	31								29	30	31				
APRIL						MA	·Υ						JU	NE						
s	М	.T	W 1	T	F 3	S	s	.М	Т	W	Т	F 1	S 2 9	S	M 1	T	W	T .	F 5 12	S 6
5	6	7	8	9	10	11	3	4	5	6	7	8	ē	7	8	9	3	11	12	13
12 19	13 20	14 21	15 22	16 23	17 24	18 25	10 17	11 18	12 19	13 20	14 21	15 22	16 23	14 21	15 22	16 23	17 24	18 25	19 26	20 27
26	27	28	29	30			24	25	26	27	28	29	30	28	29	30	-		20	
_																				
111	ıv								ст					65	от	- 140	ЕВ			
JU s	LY	т	w	1.	F	s	AU S	IGU M	ST T	w	т	F	s	SE		EMB T	W	т	F	s
S	M		1.	2	3	4	s	М	Т				1	S	М	1	W 2	3	4	S 5
S 5	M 6 13	7 14	1 8 15	2 9 16	3 10 17	4 11 18	S 2 9	M 3 10	T 4 11	5 12	6 13	7 14	1 8 15	S 6 13	M 7 14	1 8 15	W 2 9 16	3 10 17	4 11 18	12 19
S	M 6	7	1 .	9	3 10	4 11	s	M 3	T 4	5	6	7	1 8	s · ·	M 7	T 1. 8	W 2 9	3 10	4 11	12
5 12 19 26	6 13 20 27	7 14 21 28	1 8 15 22 29	2 9 16 23	10 17 24	4 11 18	9 16 23	3 10 17 24	4 11 18 25	5 12 19 26	6 13 20	7 14 21	1 8 15 22	6 13 20 27	7 14 21 28	1 8 15 22 29	9 16 23 30	3 10 17	4 11 18	12 19
5 12 19 26	6 13 20 27	7 14 21 28 BER	1 8 15 22 29	2 9 16 23 30	3 10 17 24 31	4 11 18 25	\$ 2 9 16 23	3 10 17 24	4 11 18 25 WBE	5 12 19 26	6 13 20 27	7 14 21 28	1 8 15 22 29	6 13 20 27	7 14 21 28	1 8 15 22 29	9 16 23 30	3 10 17 24	4 11 18 25	12 19 26
5 12 19 26	6 13 20 27	7 14 21 28 BEF	1 8 15 22 29	2 9 16 23 30 T	3 10 17 24 31	4 11 18 25 S	\$ 2 9 16 23 NC S 1	3 10 17 24	4 11 18 25 MBE T	5 12 19 26 R W	6 13 20 27	7 14 21 28 F	1 8 15 22 29 S 7	6 13 20 27 DE S	7 14 21 28 CE	1 8 15 22 29 MBE T	2 9 16 23 30 ER W	3 10 17 24 T 3	4 11 18 25 F 4	12 19 26 S 5
5 12 19 26	6 13 20 27	7 14 21 28 BER	1 8 15 22 29	2 9 16 23 30	3 10 17 24 31	4 11 18 25	2 9 16 23 NC S	3 10 17 24	4 11 18 25 WBE	5 12 19 26 R	6 13 20 27	7 14 21 28	1 8 15 22 29	6 13 20 27	7 14 21 28	1 8 15 22 29 MBE T	9 16 23 30 ER W	3 10 17 24	4 11 18 25	12 19 26
5 12 19 26 OC S	6 13 20 27 TO M	7 14 21 28 BEF T 6 13 20	1 8 15 22 29 W 7 14 21	2 9 16 23 30 T 1 8 15 22	3 10 17 24 31 F 2 9 16 23	4 11 18 25 S 3 10 17 24	\$ 2 9 16 23 NC S 1 8 15 22	3 10 17 24 VE M 2 9 16 23	4 11 18 25 MBE T 3 10	5 12 19 26 R W	6 13 20 27 T 5 12	7 14 21 28 F 6 13	1 8 15 22 29 S 7	6 13 20 27 DE S 6 13 20	7 14 21 28 CE M 7 14 21	1 8 15 22 29 MBE T 1 8 15 22	2 9 16 23 30 R W 2 9 16 23	3 10 17 24 T 3 10 17 24	4 11 18 25 F 4	12 19 26 s 5
5 12 19 26 OC S	6 13 20 27 TO M 5	7 14 21 28 BEF T 6 13	1 8 15 22 29 W 7 14	2 9 16 23 30 T 1 8 15	3 10 17 24 31 F 2 9 16	4 11 18 25 S 3 10 17	\$ 2 9 16 23 NC S 1 8 15	3 10 17 24 VE M 2 9	4 11 18 25 WBE T 3 10 17	5 12 19 26 R W 4 11 18	6 13 20 27 7 5 12 19	7 14 21 28 F 6 13 20	1 8 15 22 29 S 7 14 21	6 13 20 27 DE S	7 14 21 28 CE M	1. 8 15 22 29 MBE 1 1 8	2 9 16 23 30 R W 2 9 16	3 10 17 24 T 3 10 17	4 11 18 25 F 4 11 18	12 19 26 S 5 12 19

_	U		U																	
JA S 3 10 17 24 31	M 4 11 18 25	5 12 19 26	6 13 20 27	T 7 14 21 28	F 1 8 15 22 29	\$ 2 9 16 23 30	7 14 21 28	BRI M 1 8 15 22	JAR T 2 9 16 23	W 3 10 17 24	T 4 11 18 25	F 5 12 19 26	\$ 6 13 20 27	7 14 21 28	RC M 1 8 15 22 29	H T 2 9 16 23 30	W 3 10 17 24 31	T 4 11 18 25	F 5 12 19 26	\$ 6 13 20 27
AP	FIL 5 12 19 26	T 6 13 20 27	7 - 14 21 28	T 1 8 15 22 29	F 2 9 16 23 30	\$ 3 10 17 24	MA S 2 9 16 23 30	3 10 17 24 31	T 4 11 18 25	W 5 12 19 26	6 13 20 27	7 14 21 28	\$ 1 8 15 22 29	JU \$ 6 13 20 27	7 14 21 28	8 15 22 29	W 2 9 16 23 30	T 3 10 17 24	F 4 11 18 25	\$ 5 12 19 26
JU S 4 11 18 25	5 12 19 26	6 13 20 27	7 14 21 28	T 1 8 15 22 29	F 2 9 16 23 30	\$ 3 10 17 24 31	AU S 1 8 15 22 29	JGU 2 9 16 23 30	ST T 3 10 17 24 31	W 4 11 18 25	T 5 12 19 26	F 6 13 20 27	\$ 7 14 21 28	5 12 19 26	6 13 20 27	7 14 21 28	8 1 8 15 22 29	T 2 9 16 23 30	F 3 10 17 24	\$ 4 11 18 25
3 10 17 24	# 4 11 18 25	5 12 19 26	6 13 20 27	7 14 21 28	F 1 8 15 22 29	\$ 2 9 16 23 30	7 14 21 28	DVE M 1 8 15 22 29	MBI 2 9 16 23 30	3 10 17 24	T 4 11 18 25	F 5 12 19 26	\$ 6 13 20 27	5 12 19 26	6 13 20 27	7 14 21 28	ER W 1 8 15 22 29	T 2 9 16 23 30	F 3 10 17 24 31	\$ 4 11 18 25